VOLUME • CİLT: 7 / ISSUE • SAYI: 1 MARCH • MART 2023 ISSN: 2636-8307

JOURNAL OF RESEARCH IN ECONOMICS

İKTİSAT ARAŞTIRMALARI DERGİSİ



Journal of Research in Economics • İktisat Araştırmaları Dergisi

Volume • Cilt: 7 / Issue • Sayı: 1 / March • Mart 2023 6 Aylık Uluslararası Hakemli Akademik Dergi • Biannual International Peer-Reviewed Academic Journal ISSN: 2636-8307

Marmara Üniversitesi Rektörlüğü Adına İmtiyaz Sahibi • Owner

Mustafa KURT (Rektör • Rector)

Derginin Sahibi • Owner of the Journal: Marmara Üniversitesi İktisat Fakültesi Adına • On behalf of Marmara University Faculty of Economics Sadullah ÇELİK (Dekan • Dean)

Editör • Editor in Chief: Pınar DENİZ

Editör Yardımcıları • Assistant Editors: Altan BOZDOĞAN, Cemre KÜÇÜK

Yayın Kurulu • Editorial Board

Shireen ALAZZAWI, Santa Clara University, ABD Erhan ASLANOĞLU, Piri Reis Üniversitesi, Türkiye José Ignacio GALAN ZAZO, University of Salamanca, İspanya Nathalie HILMI, Centre Scientifique de Monaco, Monaco Rafal MATERA, University of Lodz, Polonya Mehmet PINAR, Edge Hill University, İngiltere Rahel SCHOMAKER, German Research Institute for Public Administration, Almanya Mahmut TEKCE, Marmara Üniversitesi, Türkiye

Danışma Kurulu • Advisory Board

Terence Huw EDWARDS, Loughborough University, İngiltere Jeffrey B. NUGENT, University of Southern California, ABD Gökhan ÖZERTAN, Boğaziçi Üniversitesi, Türkiye Young-Joon PARK, Ajou University, Güney Kore Thanasis STENGOS, University of Guelph, Kanada Erinç YELDAN, Kadir Has Üniversitesi, Türkiye

Dizgi Burcu DİKER

İletişim Bilgileri • Contact Details

Adres • Address: Marmara Üniversitesi İktisat Fakültesi Göztepe Kampüsü 34722 Kadıköy, İstanbul Tel • Phone: +90 (216) 777 30 00 Faks • Fax : +90 (216) 777 30 01 E-posta • E-mail: jore@marmara.edu.tr URL: http://dergipark.gov.tr/jore https://ikf.marmara.edu.tr/yayinlar/iktisat-arastirmalari-dergisi https://ikf.marmara.edu.tr/en/research/jore

Marmara Üniversitesi Yayınevi • Marmara University Press Adres • Address: Göztepe Kampüsü 34722 Kadıköy, İstanbul Tel • Phone: +90 (216) 777 14 00 Faks • Fax: +90 (216) 777 14 01 E-posta: yayinevi@marmara.edu.tr

İktisat Araştırmaları Dergisi sadece İngilizce makale kabul eden ve yılda iki kez (Mart ve Ekim aylarında) yayınlanan uluslararası hakemli bir dergidir. JORE, konusunda uzman yerli ve yabancı araştırmacıların iktisadın tüm alanlarına ilişkin teorik ve uygulamalı araştırma makalelerini yayınlamayı hedeflemektedir. Dergimiz, ECONLIT ve EBSCO'da indekslenmektedir ve diğer endeksler için izlenme sürecindedir. JORE, makale başvurusu aşamasından basım ve erişime kadar ücretsiz bir dergidir.

Journal of Research in Economics accepts manuscripts in English only and is a semi-annually (March and October) published international peer-reviewed journal. The aim of the JORE is to publish high quality theoretical (modeling) and applied research papers in all fields of economics. JORE is indexed in ECONLIT and EBSCO and is in the monitoring process of several indexes. JORE is a free journal from submission process to publication and access.

İçindekiler • Contents

ARAŞTIRMA MAKALELERİ / RESEARCH ARTICLES

An Empirical Analysis of Speculative Behavior and the Spillover Effect in Cryptocurrency Markets
Emrah DOĞAN, Selin YALÇINTAŞ1
The Extent of Household Poverty in Afghanistan: A Case Study of Mazar-I-Sharif City, Balkh Province (2019/20 and 2020/21)
Baqir KHAWARI
A Cross Sectional Analysis on the Impacts of ICT on Turkish Manufacturing Industry Yasemin ÖZBAL
The Evolution of the Digital Divide in Turkey <i>Deniz Ece DALGIÇ-TETIKOL, Emin KÖKSAL, Bülent GÜLOĞLU</i>
What a Hard Decision: Capital Structure of Reits in Istanbul Stock Exchange <i>Vijdan UĞURLUAY, Aslı AYBARS</i> 84

ARAŞTIRMA MAKALESİ / RESEARCH ARTICLE

AN EMPIRICAL ANALYSIS OF SPECULATIVE BEHAVIOR AND THE SPILLOVER EFFECT IN CRYPTOCURRENCY MARKETS

Emrah DOĞAN^{*}[©] Selin YALÇINTAŞ^{**}[©]

Abstract

For risk management and stable pricing in the cryptocurrency market, it is necessary to determine the interdependence of speculative behaviour and crypto assets. The correlation and high volatility caused by the interdependence of financial assets in the cryptocurrency market can lead to spreading risks. The study aims to measure the speculative behaviour and spillover effect in the prices of financial assets in the cryptocurrency market. The study used the SADF test, the generalized Dickey-Fuller test (GSADF), and the frequency domain causality test of Breitung and Candelon (2006) to determine the speculative behaviour and spillover effect in the prices of financial assets in the cryptocurrency market. Empirical evidence of speculative bubble formation between January 1, 2018, and December 2021 for the cryptocurrency assets covered in the study (ADA, BNB, BTC, DOGE, ETH, XLM, and XRP) is presented. Moreover, the frequency domain causality results obtained in the study show a contagion and spillover effect between crypto assets. The results provide essential information on the development of speculative behaviour and spread risk in the formation of financial asset prices in the cryptocurrency market.

Keywords: Cryptocurrency Markets; Bubbles; Spread Risks; Right-tailed Unit Root Tests, Frequency Domain Causality

JEL Classification: D53, F38, G00

I. Introduction

Developed financial markets have positive effects on economic growth and development. One of the most studied factors among the determinants of a well-developed financial system is the interdependence among financial assets. The main reason is the integration of financial markets and assets as part of globalization. Globalization causes a shock in one country's financial system to spread rapidly to the rest of the world (Polat and Eş-Polat, 2022). A similar situation applies to the cryptocurrency market. The lack of a regulatory and supervisory mechanism in

^{*} Istanbul Gelisim University, Department of International Trade and Finance, E-mail: emdogan@gelisim.edu.tr, ORCID: 0000-0001-9870-5719

^{**} Istanbul Gelisim University, Department of International Trade and Finance, E-mail: syalcintas@gelisim.edu.tr, ORCID: 0000-0003-2431-4875

the cryptocurrency market and the still developing and immature technology of blockchain technology increases the volatility in the relevant market, increasing the correlation and cooperation relationship between the assets in the cryptocurrency market.

The high correlation relationship between assets in the cryptocurrency market is one factor that is also effective in financial decision-making. This is because, as Huynh (2019) states, determining the degree of interdependence between financial instruments is essential for developing portfolio management and hedging strategies. Therefore, in assessing the degree of cooperation in the cryptocurrency market, the management of financial assets is critical to the forecasting and pricing process. Another important topic is the interdependence between financial markets, the development of the movement and volatility of financial instruments, and asset prices. In particular, unstable pricing in financial markets can have substantial effects that can lead to a global financial crisis, as was the case in the 2008 global crisis. In this context, it can be assumed that one factor that triggers these strong effects on modern financial markets and instruments is the increasing correlation and volatility in the cryptocurrency market.

Research on the causes of high correlation and volatility in the cryptocurrency market is gaining momentum in two different areas (Moratis, 2021). One is that fundamental external factors such as economic, financial, and geopolitical uncertainty cause high correlation and volatility in the cryptocurrency market (Giudici and Abu-Hashish, 2019; Smales, 2019; Panagiotidis et al., 2018; Moratis, 2021). The other is intrinsic fundamentals, such as increased volatility in the cryptocurrency market and the high correlation between crypto assets (Francés et al., 2018; Ji et al., 2019). As these internal and external factors prevent investors from reducing risk, they inhibit market dynamics in cryptocurrency and all financial markets.

The primary motivation of this study is to determine the speculative behaviours and the spillover effect in the prices of seven crypto assets (ADA, BNB, BTC, DOGE, ETH, XLM, and XRP) are dominant in terms of market value in the crypto money market. Compared to other related studies examining speculative price behaviour and spillover effects in cryptocurrencies, it differs from similar studies regarding subject and method. First, the study differs from other studies in analyzing the interconnectedness and persistence of seven significant cryptocurrencies. Secondly, the study provides an essential guide, especially in shaping the markets and investor decisions, by revealing the spread of speculative price behaviours and causality effects among cryptocurrencies. Third, the fact that the selected period of the study covers the period between January 1, 2018, and December 2021 contributes to the observation of the possible effects of investors' tendency to invest in different cryptocurrencies following the rapid increase in Bitcoin prices in the last quarter of 2017. Fourth, SADF and GSADF tests, which detect price bubbles, allow better inferences than the methods in the literature, thanks to their dynamic structure, unlike indirect methods. The frequency domain causality test, another technique used, makes a significant difference in determining whether there is a connection between cryptocurrencies in the short, medium, and long term, as it allows investigation of the causality dynamics at different

frequencies. Therefore, the study may provide more compelling evidence than similar studies in the literature.

This paper is organized as follows. After this introductory section, Section 2 presents an overview of previous research on the issue. Section 3 presents this study's model, dataset, and method. Section 4 introduces the empirical results of the analysis. Finally, Section 5 concludes the research undertaken in this study.

2. Literature

According to the scope of the study, the literature, bubble formation, spillover effects, and causality are examined. If the bubble concept is evaluated from an economic perspective, it is characterized as a deviation from the fundamental value of the current asset. However, it isn't straightforward to determine this fundamental value, especially in the cryptocurrency market. For this reason, bubbles in cryptocurrencies are defined as price breakouts and provide an opportunity to do more reliable valuations (Enoksen et al., 2020).

The various dynamics behind the price increase in the cryptocurrency market can be grouped under two headings in general; i) the price increase experienced as a result of the introduction of various macroeconomic dynamics that will affect the returns of traditional investment instruments, as market participants turn to digital investment instruments to compensate for their potential losses ii) price increase through speculative effects. In the literature, it is seen that the studies on the values of crypto assets primarily focus on speculative effects. It is widely believed that difficulties in determining the fundamental importance of digital currencies set the stage for speculative behaviour. Market price formation is shaped around these relationships (Kristoufek, 2013; Shahzad et al., 2022). This view is supported by the assumption that the factors that play a role in price formation in cryptocurrency markets are not based on the same dynamics as the determinants of traditional asset markets. The difference between crypto money markets from traditional financial markets is that their supply is fixed, and the investor expectations on the demand side have a critical role. Therefore, the active part of the participants in the price formations in the crypto market makes the market dynamics open to speculative behaviours. Evlimoğlu and Güder's (2021) studies support this view. The main points, how and where the determinants of potential bubbles that may occur in crypto markets and the economic bubbles experienced in the past differ, were stated in their studies. These factors are listed as the fact that the fundamental value has not been determined in the crypto markets, the supply is limited, and blockchain technology is still developing. Therefore, it is argued that the decisionmaking processes of market actors are determined not on a rational basis, that is, on complete information, but on asymmetric information and irrational expectations (Yanık and Aytürk, 2011). The fact that the value of crypto assets is shaped in line with the perception of market actors triggers unstable price formation, preparing the ground for speculative bubbles. Due to the price movements in cryptocurrency markets in recent years, studies focusing on bubbles in

this area have come to the forefront Yermarck (2015) argues that Bitcoin, which has the most significant value in the cryptocurrency market, is a speculative asset, while Cheah and Fry (2015) argue that Bitcoin has speculative bubbles. In another study that comes to similar conclusions, cryptocurrency markets are found to be highly volatile and subject to speculative effects (Fry and Cheah, 2016). In this context, the supremum-augmented Dickey-Fuller tests (SADF) of Phillips et al. (2011) and the generalized supremum-augmented Dickey-Fuller tests (GSADF) of Philips et al. (2015) are widely used. Several studies using the method have found evidence of cryptocurrency bubble formation (Cheung et al., 2015; Su et al., 2018; Bouri et al., 2019; Waters and Bui, 2021). The empirical studies by Souza et al. (2017) using RADF, SADF, and GSADF tests prove that speculative bubbles are common in cryptocurrency. On the other hand, the study by Buğan (2021), which investigated the formation of bubbles in cryptocurrencies, found that the bubbles detected in Litecoin and Cardano were not statistically significant as a result of the GSADF test, while the existence of bubbles was accepted for Bitcoin, Ethereum, Ripple, and Chainlink. In Şahin (2020) study, the bubbles in cryptocurrencies Bitcoin, IOTA, and Ripple were tested by the GSADF test, and the bubble formation in cryptocurrencies was confirmed again. In addition, the study drew attention to the impact of news manipulation on explaining the periods when bubbles were formed.

The literature also contains studies that examine the formation of bubbles in different types of markets. Maouchi et al. (2022), using the real-time bubble detection method proposed by Phillips and Shi (2020), investigated the existence of digital financial bubbles and detected bubble formation in 3 NFT, 9 DeFi tokens, Bitcoin and Ethereum. The study's findings covering the Covid-19 period are that the bubbles in DeFi and NFTs are more giant than those in Bitcoin and Ethereum but occur less frequently. Using the PSY test (GSADF), Gharib et al. (2021) point to boom periods in the crude oil and gold markets between 2010 and 2020. In particular, the Covid-19 period has shown the contagion effect in the bubbles in the two markets. When crypto asset prices are volatile, markets give signals of uncertainty and instability. The seizure of these factors in the markets raises financial concerns for crypto assets. Therefore, it is essential to measure the interdependence and volatility spreads of cryptocurrencies in shaping the risk management mechanism within the scope of the decision processes of investors. For this reason, in addition to detecting bubbles, evaluating the contagion effect of bubbles is essential in deepening the discussion of cryptocurrencies. Uncovering the spillover and causality effects between cryptocurrencies is crucial, especially in shaping markets and investors' decisions.

Various studies have been conducted in the literature on whether there are causality and volatility spillovers between cryptocurrencies. The logistic regression results in the study by Bouri et al. (2019) show that the probability of an explosion period in cryptocurrencies is shaped by the presence of explosions in other cryptocurrencies. Huynh (2019) investigates the spillover effects between five cryptocurrencies (bitcoin, Ethereum, XRP, Litecoin, Stellar) through VAR – SVAR Granger causality and the Copulas method. The research results show that Ethereum is independently compared to other cryptocurrencies, while the validity of the spillover effect between the different cryptocurrencies is questioned. On the other hand, the Student's t-Copulas

test suggests a contamination risk when cryptocurrencies contain extreme values. When examining the competition between cryptocurrencies, one study's empirical results indicate a spread from Ripple to Bitcoin (Fry and Cheah, 2016). In another study, they pointed out the presence of structural breaks in the cryptocurrency market. They concluded that systematic price fluctuations spread from currencies with low market values to those with high market values (Canh et al., 2019). Yi et al. (2018), according to the results of their studies, the existence of a spillover effect is assumed in cryptocurrencies. Global finance, uncertainty effects, and trading volume are the variables that trigger the spillover effect. Ji et al. (2019) studied the return and volatility spreads of six cryptocurrencies and found that Bitcoin and Litecoin are at the centre of returns. In addition, positive returns were shown to be weaker than negative returns.

In their study, Enoksen et al. (2020) investigated the dynamics associated with the presence of bubbles. They used the PSY (GSADF) method to detect bubbles in cryptocurrency markets, and it was found that the variables that predict bubble formation are trading volume, transactions, and volatility. Cryptocurrency bubbles show a positive relationship with EPU (economic policy uncertainty index) and a negative relationship with VIX (fear index).

Canh et al. (2019) used data from seven cryptocurrencies (Bitcoin, Litecoin, Ripple, Stellar, Monero, Dash, and Bytecoin); the Granger causality test, the LM test for ARCH, and the DCC-MGARCH method were preferred. The results of the study show that there are structural breaks and volatility spillovers in the cryptocurrency market. It is found that the spillover effect is from more minor market cap currencies to more significant coins. Empirical evidence shows that cryptocurrencies exhibit strong and positively correlated volatility spillovers. Kirikkaleli et al. (2020) present empirical evidence of bubbles in Bitcoin and Ethereum, Litecoin, and Ripple between 2016 and 2019 and accept a positive relationship between Bitcoin and three other cryptocurrencies in the short run. In their studies using the quantile Granger non-causality test, Kim et al. (2021) conclude that coins with a high market value do not exhibit a strong bidirectional relationship with other currencies. While XRP has bidirectional causality with other coins, EOS has the weakest causal relationship with all coins. On the other hand, BNC has bidirectional causality with all coins except EOS. Katsiampa et al. (2019) studied the relationship between Bitcoin-Ethereum, bitcoin-litecoin, and etherium-litecoin between August 7, 2015, and July 10, 2018, using the BEKK model. The results show that cryptocurrency price volatility relates to prior volatility and currency shocks. While there is a bidirectional spread between Bitcoin and the other two cryptocurrencies, the spread between Ethereum and Litecoin is one-way. In addition, studies examining the relationship and spillover effect between cryptocurrencies and other financial assets are also prominent. Using the VAR GARCH model, Bouri et al. (2018) found that bitcoin returns are associated with traditional assets such as stocks, commodities, currencies, and bonds. The study also found that Bitcoin is a receiver rather than a transmitter of volatility. The volatility spillover index was created using the TVP-VAR model of Cao and Xie (2022). It was found that there is an asymmetric and time-varying volatility spillover effect between cryptocurrency and the Chinese financial market. At the same time, it has been determined that the risk spread of the financial market has a feeble impact on cryptocurrency. In contrast, the risk spread of cryptocurrency on the financial market is substantial. In the study by Elsayed et al. (2020), which investigated the spillover effects between three cryptocurrencies and nine foreign currencies using the Diebold-Yilmaz method, the return spillover effect for Bitcoin and Litecoin in the first three quarters of 2017 was determined. As a result of the Bayesian chart structure model VAR (BGSVAR), it was found that the level of bitcoin to the Chinese yuan, the bitcoin and litecoin values of Ripple, and the level of litecoin are dependent on Ripple and the Chinese yuan. The result of the study is causality between cryptocurrencies; among foreign currencies, only the Chinese yuan influences cryptocurrencies.

When considered as a whole, external dynamics, such as the fact that the cryptocurrency market is an unregulated market and the technological infrastructure development process, have not yet been completed. The increase in economic and geopolitical uncertainty leads to a rise in the vulnerability of cryptocurrencies to speculative behaviours in the market and triggers the formation of a bubble. By encouraging the spread of interdependence and volatility among cryptocurrencies, these developments pave the way for market efficiency deterioration.

3. Data Set and Method

3.1. Dataset

The study empirically investigates the existence of asset price bubbles in cryptocurrency markets, asset interdependence, and the spillover effect. In this regard, the variables used in the study were ADA, BNB, BTC, DOGE, ETH, XLM, and XRP, depending on the availability of data and the volume of transactions in the cryptocurrency market. The descriptive test statistics for the above variables are shown in Table 1. Accordingly, daily data was used for the selected variables between January 1, 2018, and December 2021, obtained from the Yahoo Finance database. On December 31, 2021, the cryptocurrency market cap was approximately 92 billion USD. On the same date, the share of cryptocurrencies selected as the study's sample in the market volume was approximately 65% (https://www.coinecko.com/en/global-charts, Access Date: 15.01.2023). Another factor affecting the period selection in the study is that, following the rapid increase in Bitcoin prices in the last quarter of 2017, investors tended to invest in different cryptocurrencies.

According to the results of the descriptive statistics given in Table 1, it is seen that the cryptocurrencies with the highest standard deviation are BTC and BNB. The lowest standard deviation is seen in DOGE. On the other hand, all variables used in the study are skewed to the right. Jarque-Bera test results, which indicate whether the variables show a normal distribution or not, suggest that the variables do not comply with the normal distribution.

	Mean	Median	Max.	Min.	Std. Dev.	Skewness	Kurtosis	J-B	Obs.
ADA	0.46	0.10	2.96	0.02	0.68	1.689	4.737	878.75(0.00)***	1461
BNB	107.67	19.69	675.68	4.52	177.68	1.723	4.491	858.83(0.00)***	1461
BTC	18375	9475	67566	3236	17760	1.303	3.153	415.30(0.00)***	1461
DOGE	0.053	0.003	0.684	0.001	0.107	2.256	7.934	2722(0.00)***	1461
ETH	937.36	346.52	4812	84.308	1196	1.646	4.490	795(0.00)***	1461
XLM	0.20	0.144	0.896	0.033	0.148	1.022	3.598	276(0.00)***	1461
XRP	0.52	0.363	3.377	0.139	0.389	2.272	11.377	5529(0.00)***	1461

Table 1: Descriptive Statistics

Note: Values in parentheses are probability values. In addition, * indicates the significance levels of 0.10, **0.05, and *** 0.01.

3.2. Research Methodology

In the study, first, whether there are speculative bubbles in the cryptocurrency market, Phillips et al. (2011) ekus ADF (SADF) and Phillips et al. (2015) generalized Dickey-Fuller (GSADF) test. The methods in question are recursive and right-justified unit root tests that have been widely used recently due to their excellent performance in detecting speculative bubbles and their occurrence.

The Exus-ADF test (SADF), one of the most commonly used right-tailed unit root tests among these methods, was developed by Phillips et al. (2011), and the extended standard Dickey-Fuller test (ADF) was developed to detect speculative bubbles and when they occur. As Homm and Breitung (2012) found, this test performs as well as other tests using similar procedures. The SADF test is essentially based on an iterative estimation of the standard ADF test. The SADF test is obtained as the lower value corresponding to the statistical ADF sequence and is obtained by estimating the values given in Equation 1 using least squares (Philips et al., 2015).

$$x_{t} = \mu_{x} + \delta x_{t-1} + \sum_{j=1}^{J} \phi_{jj} \Delta x_{t-j} + \varepsilon_{x,j} \varepsilon_{x,t} \sim NID(0, \sigma_{x}^{2})$$

$$\tag{1}$$

For some values of J given in equation 1, the null hypothesis H0: δ =1 and the alternative hypothesis H1: δ > 1 are formed in the right-tailed SADF unit root test so that the NID is independent and normally distributed. Iterative regressions on the sample data then increase one observation at each run. The result is repeatedly estimated using subsets.

$$sup_{r\in[r_0,1]} ADF_r \to sup_{r\in[r_0,1]} \frac{\int_0^r \widetilde{W} dW}{\left(\int_0^r \widetilde{W}^2\right)^{1/2}}$$
(2)

Equation 2 shows standard Brownian motion W and reduced Brownian motion $\overline{W}(r) = W(r) - \frac{1}{r} \int_0^1 W$ (Philips et al., 2011: 206-207). Considering the criticism in the literature that the statistical power of the SADF test decreases in the case of multiple bubbles,

Phillips et al. (2015) developed the generalized GSADF unit root test to address the shortcomings of the SADF test in this direction. Although the GSADF test has similar features to the SADF test, it differs because the standard ADF test uses an iterative soft estimate of the regression obtained from the standard ADF test in computing the test, allowing for long-term nonlinear structures and structural breaks. In this regard, the GSADF test outperforms the SADF and standard ADF unit root tests by providing more consistent and accurate results in the case of multiple bubbles (Phillips et al., 2015). Although the GSADF test is based on the recursive operation of the ADF test in subsamples, similar to the SADF test, it is referred to as the most significant ADF test because it is much broader than the SADF test.

To calculate the GSADF test statistic, we first estimate the iterative regression equation 3. Here, k is the lag length, and r1 and r2 are included in the equation to represent the start and end points of the subsample so that iterative regression estimates can be performed (Çağlı and Mandacı, 2017: 66).

$$\Delta y_{t} = \hat{\alpha}_{r_{1}, r_{2}} + \hat{\beta}_{r_{1}, r_{2}} y_{t-1} + \sum_{i=1} \hat{\psi}_{r_{1}, r_{2}}^{t} \Delta y_{t-i} + \hat{\varepsilon}_{t}$$
(3)

The GSADF test Equation 3 is repeatedly estimated for multiple subsamples using subsets with a future date. Unlike the SADF test, subsamples are created where the initial points of the subsamples in r1 change dynamically instead of the final moments in r2 and deviate from zero (Çağlı and Mandacı, 2008). 2017:66). From this point of view, the GADF test is calculated using the formula given in equation 4 (Philips, Shi, and Yu, 2015: 1049)

$$GSADF(r_0) = sup_{r_2 \in r_1 \in [0, r_2 - r_1]}[r_0, 1] \{ADF_{r_1}^{r_2}\}$$

$$\tag{4}$$

The frequency domain causality test, another method used in the study, allows for investigating the causality relationship of the variables under study at multiple time points. Since traditional causality tests generate test statistics for a single t-period, they ignore the possibility that the causality relationship changes at different frequencies and periods (Bozoklu & Yılancı, 2013). On the other hand, traditional causality methods perform a linear causality analysis between the variables included in the study. Geweke (1982) and Hosoya (1991) proposed a causality analysis method based on spectral density decomposition at a specific frequency to address this shortcoming of traditional causality analysis. Subsequently, Breitung and Candelon (2006) developed a computational method that simplifies the complex structure of frequency-based causality analysis. This calculation method has created a procedure based on the autoregressive parameters based on the VAR model (Başarır, 2018). Due to its structure, the method also has the advantage of performing a nonlinear causality analysis between the variables included in the study. In this context, the causality analysis can be performed for different frequencies as follows:

$$M_{y \to x}(\omega) = \log \left[\frac{2\pi f_x(\omega)}{|\Psi_{11}(e^{-i\omega})|^2} \right] = \log \left[1 + \frac{|\Psi_{12}(e^{-i\omega})|^2}{|\Psi_{11}(e^{-i\omega})|^2} \right]$$
(5)

8

According to equation 5, in the case of $|\Psi 12(e^{-i\omega})|=0$ above a certain w frequency, there is no causality relationship from the y variable to the x variable (Ciner, 2011:500). Breitung and Candelon (2006) change the hypothesis to equation #5, according to which if $My \rightarrow x(\omega)=0$. $|\Psi 12(e^{-i\omega})|=0$ then ,

$$\Psi(L) = \Theta(L)^{-1} G^{-1} \text{ ve } \Psi_{12}(L) = -\frac{g^{22} \Theta_{12}(L)}{|\Theta(L)|}$$
(6)

In equation 6, g^{22} represents the common diagonal elements of the G^{-1} matrix, $|\Theta(L)|$ represents the determinant of $\Theta(L)$. In this case, causality in the frequency domain can be tested with the following equation. (Bodart and Candelon, 2009: 143).

$$\left|\Theta_{12}(e^{-i\omega})\right| = \left|\sum_{k=1}^{p} \Theta_{12,k} \cos(k\omega) - \sum_{k=1}^{p} \Theta_{12,k} \sin(k\omega) i\right| = 0$$
⁽⁷⁾

Since θ_{12} indicates the element of θ_k and θ_k in equation 7, the expression $|\theta_{12}(e-i\omega)|=0$ can be expressed such that "*y*" is not the cause of "*x*" at "*w*" (Tarı et al., 2012: 10). Breitung and Candelon (2006) model the method as a function of linear constraints, as shown in equation 8. In this case, the equation VAR can be formed with 9 for *x*.

$$\sum_{\substack{k=1\\p}}^{p} \Theta_{12,k} \cos(k\omega) = 0$$
$$\sum_{\substack{k=1\\k=1}}^{p} \Theta_{12,k} \sin(k\omega) = 0$$
(8)

$$x_{t} = \alpha_{1}x_{t-1} + \dots + \alpha_{p}x_{t-p} + \beta_{1}y_{t-1} + \dots + \beta_{p}y_{t-p} + \varepsilon_{1t}$$
(9)

Since the hypothesis $My \rightarrow x(\omega)=0$ is equivalent using equations 8 and 9 with linear constraints, the H0 hypothesis can be stated in equation 10.

$$R(\omega) = \begin{bmatrix} \cos(\omega) & \cos(2\omega) & \cdots & \cos(p\omega) \\ \sin(\omega) & \sin(2\omega) & \cdots & \sin(p\omega) \end{bmatrix}$$
(10)

Thus, H0: $(\omega)\beta=0$ ($\beta=[\beta_1,...,\beta_p]'$) $R(\omega)$ is calculated using the following equation. On the other hand, it is possible to separate the causal dynamics between the variables studied in the frequency domain causality analysis temporarily and permanently. Accordingly, a short-term (temporary) causality analysis is performed when the ω -frequency value is calculated for a high frequency ($\omega = 2.5$). When the value of $\omega=1.5$, a medium-term causality analysis is performed, while when the value of $\omega=0.5$ (low frequency), a permanent causality analysis is possible. Therefore, causality analysis in the frequency domain allows the decomposition of causality into more than one time period.

4. Empirical Results

In this part of the study, the hypothesis formulated as H1 is first tested using the prices of 7 financial assets in the cryptocurrency market. The hypothesis states that increasing financial interconnectedness with globalization will cause a shock in the financial system to spread quickly to the rest of the world. In the case of a spillover effect, bubbles can occur when investors continue to hold assets because they believe they can sell them at a higher price than other investors, even though the financial asset's price exceeds its fundamental value. This situation, which also applies to the cryptocurrency market, leads to the unstable pricing of cryptocurrency market assets. In other words, bubbles can occur in the prices of crypto assets.

H1: External factors affecting the cryptocurrency market make for unstable pricing.

The SADF and GSADF tests were used to determine the presence of bubbles by testing the hypothesis expressed as H1 and to determine when bubbles occur. In applying the above tests, 2000 replicate Monte Carlo simulations were used for each observation. The results of the estimations are reported in Table 1.

	SADF Test	GSADF Test
	Statistic	Statistic
ADA	3.00***	12.52***
BNB	19.72***	19.81***
BTC	5.86***	8.04***
DOGE	15.99***	16.01***
ETH	5.79***	6.68***
XLM	-1.00	6.139***
XRP	-1.72	6.128***

Table 2: The SADF and GSADF Test Statistics

Note: Critical values for SADF statistics are 0.43, 0.69, and 1.15 for 10%, 5%, and 1% significance levels, respectively. Critical values for GSADF statistics are 1.28, 1.46, and 1.91 for 10%, 5%, and 1% significance levels, respectively. In addition, the significance levels * 0.10, ** 0.05, and *** 0.01 are given. These critical values were obtained by Monte Carlo simulation with 2,000 replicates.

Examination of the SADF and GSADF test statistics in Table 2 shows that the estimated test statistics for the cryptocurrencies ADA, BNB, BTC, DOGE, and ETH are more significant than the critical values. Therefore, a speculative bubble in these currencies was established for the analyzed periods. On the other hand, when examining the SADF and GSADF test statistics obtained for the XLM and XRP currencies from the selected assets in the cryptocurrency market, it can be seen that the estimated SADF test statistics are smaller than the critical values. In other words, the H0 hypothesis is accepted. However, the estimated GSADF test statistics are shown to be larger than the critical values, so the H0 hypothesis is rejected. Phillips et al. (2015) found that the GSADF test is more consistent and gives better results than the SADF and standard ADF

tests. Based on this view, it can be said that a speculative bubble occurred for the XLM and XRP currencies during the analyzed periods.

In summary, the test results show that although the prices of all currencies exceed the fundamental value of the prices of the analyzed period, they continue to hold assets because they believe they can sell them at a higher price than other investors. In other words, it can be said that bubbles were created in the cryptocurrency market during the studied period. Thus, the obtained results confirm the hypothesis that external factors affecting the cryptocurrency market make the pricing unstable and lead to the formation of bubbles.

Having established the presence of bubbles in the selected cryptocurrencies, the second phase began to identify the periods in which bubbles occurred. In this way, it is possible to determine which factors cause instability in price formation and lead to the formation of bubbles.



Figure 1: ADA Cryptocurrency Test Results Charts

From the SADF and GSADF test charts shown in Figure 1, it can be seen that a bubble formed during the period from late January 2021 to early June 2021. During the period in question, the technological upgrade of the cryptocurrency ADA led to excessive demand for the cryptocurrency ADA by many investors, creating a bubble.





Based on the SADF and GSADF test charts of the cryptocurrency BNB shown in Figure 2, it was determined that a bubble formed from the beginning of 2021 to the end of May 2021. In the studied period, it can be said that the interventions of the cryptocurrency exchange Binance to reduce the total supply of BNB cryptocurrency and the excessive demand for BNB due to the increase in transaction costs in Ethereum drive up prices and cause the formation of a bubble.



Figure 3: BTC Cryptocurrency Test Results Charts

According to the SADF and GSADF charts for bitcoin in Figure 3, a bubble in the bitcoin price was observed in the last quarter of 2018, the middle of 2019, and between the last quarter of 2020 and the second quarter of 2021. During the earlier periods, the improvements in the system's functioning with the blockchain system's updates have increased the demand for bitcoin and pushed the prices. This has led to a bubble in BTC prices. On the other hand, it can be said that the big rally in BTC price was effective in the bubble formation observed between the last quarter of 2020 and the second quarter of 2021.





The SADF and GSADF test charts of the cryptocurrency DOGE, shown in Figure 4, indicate that there have been several bubble formations between the last quarter of 2020 and mid-2021. In the mentioned period, it can be observed that external factors are particularly effective in

bubble formation in DOGE cryptocurrency prices. In particular, social media posts for the cryptocurrency DOGE created excessive demand by directing investors to this cryptocurrency during the period in question. The high demand for the stocks in question led to a large price rally. As a result, the sharp rise in prices led to a bubble.





According to the SADF and GSADF charts for Ethereum in Figure 5, a price bubble can be observed from early 2021 to mid-2021. The reason for the bubble formation in the mentioned period is the announcement by the financial institutions that the Ethereum Trust will be reopened for public trading in the mentioned period. Also, in the mentioned period, the tendency of retail investors to engage in decentralized trading of virtual currencies increased the demand for Ethereum, one of the currencies with the largest market volume in the cryptocurrency markets. It contributed to the formation of speculative price bubbles.





The SADF chart for Stellar (XLM) in Figure 6 shows no speculative price bubble during the period in question. However, the graphs of the GSADF test, which gives more accurate results than the SADF test, indicate the existence of several different bubbles during the period in

question. The main reason for this difference is that while the SADF test is a powerful method for detecting bubbles, it can be weak, especially in more than one price bubble. As confirmed by the GSADF graphs, price bubbles occurred in three different periods during the relevant period: the last quarter of 2020, the first quarter of 2021, and the second quarter. It can be said that regulatory decisions made in developed countries regarding the blockchain system and cryptocurrencies were effective in forming these bubbles. In the same period, developments such as the partnership of major banks with Stellar in Europe led to an increase in demand. They became one of the factors contributing to the inflation of the Stellar price.





The SADF chart of Ripple (XRP) in Figure 7 shows no speculative price bubble during the period. However, the charts from the GSADF test, which provides more accurate results than the SADF test, provide empirical evidence of the existence of several different bubbles during the relevant period. As shown in the GSADF charts, price bubbles are observed in the third quarter of 2018, the first and fourth quarters of 2020, and the first and third quarters of 2021. In the formation of price bubbles, banks in Japan and South Korea announced their intention to test Ripple's blockchain technology in 2018. In late 2019, Japan and South Korea will begin testing blockchain technology to reduce the time and costs of international money transfers between the two countries. In 2021, price increases in other cryptocurrencies drove up Ripple's prices and contributed to the formation of a bubble.

In this part of the study, the hypothesis formulated as H2 is tested using the prices of 7 financial assets in the cryptocurrency market.

H2: Assets in the cryptocurrency market have the power to affect each other directly

The said hypothesis, Frequency Domain Causality Test, was used to determine whether the assets in the cryptocurrency market have the power to influence each other.

The Frequency Domain Causality Test used to test the hypothesis formulated as H2, can distinguish between temporary or permanent causal dynamics between crypto assets. For this purpose, test

statistics with high (ω =2.5) frequency were used when examining short-term causality, while test statistics with medium frequency (ω =1.5) were utilized for medium-term causality. Test statistics with low (ω =0.5) frequency were used to study long-term permanent causality. The test results are presented in Table 2, Table 3, and Table 4.

Causality Direction	ADA	BNB	BTC	DOGE	ETH	XLM	XRP
ADA 🖚	-	6.00**	2.83	6.53**	9.81***	10.81***	6.15**
BNB 🖚	17.68***	-	31.82***	51.25***	6.92**	23.20***	18.65***
BTC ***	5.02*	10.00***	-	19.50***	17.23***	3.48	4.97*
DOGE 🖚	24.88***	18.13***	2.33	-	3.64	21.18***	10.99***
ETH 🖚	11.83***	1.51	20.48***	27.79***	-	16.40***	15.73***
XLM 🖚	1.27	1.56	11.59***	7.21**	0.008	-	1.29
XRP 🖚	0.54	1.33	7.39**	13.86***	2.02	1.68	-

Table 2: Short-term (ω =2.5) Frequency Domain Causality Test Results

Note: the significance levels * 0.10, ** 0.05, and *** 0.01 are given.

According to the results of the short-term frequency domain causality test in Table 2, a bidirectional causality relationship was found between ADA cryptocurrency and BNB, DOGE, and ETH. A bidirectional causality relationship was found between BNB cryptocurrency and ADA, BTC, and DOGE cryptocurrencies. A short-term and bidirectional causality relationship was found between the cryptocurrency BTC and the cryptocurrencies BNB, ETH, and XRP. A statistically significant and bidirectional causality relationship was found between the cryptocurrency DOGE and the cryptocurrencies ADA, BNB, XLM, and XRP. It is found that there is a transitory and bidirectional causality relationship between the cryptocurrency ETH and the cryptocurrency values ADA and BTC. Finally, a bidirectional causality relationship existed between XRP and BTC, DOGE.

On the other hand, a one-way causality relationship was found from cryptocurrency ADA to cryptocurrencies XLM and XRP. A one-way causality relationship was found between BNB and ETH. Similarly, a one-way causality relationship was found to exist from BTC to ADA. A unidirectional and statistically significant causality relationship exists between ETH to DOGE, XLM, and XRP. A unidirectional causality relationship was found to exist between XLM cryptocurrency and BTC.

			× ·				
Causality Direction	ADA	BNB	BTC	DOGE	ETH	XLM	XRP
ADA 🖚	-	5.80*	2.63	7.19**	9.53***	10.99***	7.02**
BNB 🖚	18.74***	-	28.93***	56.03***	7.59**	23.79***	20.22***
BTC →	5.22*	8.62**	-	20.34***	15.99***	3.63	5.59*
DOGE 🖚	26.57***	20.33***	1.65	-	3.60	21.55***	12.18***
ETH 🖚	12.47***	1.36	19.46***	29.18***	-	16.91***	16.95***
XLM 🖚	1.24	1.53	10.40***	7.98**	0.01	-	0.79
XRP 🖚	0.40	1.29	6.37**	15.33***	2.04	1.11	-

Table 3: Mid-term (ω =1.5) Frequency Domain Causality Test Results

Note: the significance levels * 0.10, ** 0.05, and *** 0.01 are given.

According to the medium-term frequency domain causality test results listed in Table 3, a bidirectional causality relationship was found between ADA cryptocurrency and BNB, DOGE, and ETH. A bidirectional causality relationship was found between BNB cryptocurrency and ADA, BTC, and DOGE cryptocurrencies. A medium-term and bidirectional causality relationship was found between the cryptocurrency BTC and the cryptocurrencies BNB, ETH, and XRP. A statistically significant and bidirectional causality relationship was found between the cryptocurrencies ADA, BNB, XLM, and XRP. A bidirectional causality relationship existed between the cryptocurrency ETH and the cryptocurrency assets ADA and BTC. Finally, a bidirectional causality relationship existed between XRP and BTC, DOGE.

It was found that there is a one-way causality relationship between the cryptocurrency XLM and BTC. On the other hand, a one-way causality relationship existed between the cryptocurrency ADA and the cryptocurrencies XLM and XRP. It was found that there is a one-way causality from BNB to ETH, XLM and XRP. Similarly, it was found that there is a one-way causality relationship between BTC to ADA and DOGE. A unidirectional and statistically significant causality relationship exists between ETH to DOGE, XLM, and XRP.

			-		•		
Causality Direction	ADA	BNB	BTC	DOGE	ETH	XLM	XRP
ADA 🖦	-	3.22	1.30	12.27***	5.71*	12.96***	21.42***
BNB 🖚	22.03***	-	2.33	88.73***	10.14***	29.38***	41.89***
BTC ***	10.21***	1.15	-	31.54***	3.19	7.40**	16.05***
DOGE 🖚	26.43***	28.70***	4.91*	-	1.29	22.12***	25.77***
ETH 🍽	14.36***	0.28	8.25**	36.93***	-	22.12***	34.05***
XLM **	0.93	1.21	0.32	12.98***	0.13	-	4.59
XRP 🖚	1.62	0.94	0.40	22.74***	1.66	3.14	-

Table 4: Long-term (ω =0.5) Frequency Domain Causality Test Results

Note: the significance levels * 0.10, ** 0.05, and *** 0.01 are given.

According to the long-term frequency domain causality test results in Table 4, a bidirectional causality relationship was found between ADA cryptocurrency and DOGE and ETH. A bidirectional causality relationship was found between BNB cryptocurrencies and DOGE cryptocurrencies. A bidirectional causality relationship was found between BTC and DOGE. A statistically significant and bidirectional causality relationship was found between DOGE cryptocurrency and ADA, BNB, BTC, XLM, and XRP cryptocurrencies. It was found that there is an ongoing and bidirectional causality relationship between the cryptocurrency ETH, the cryptocurrency assets ADA, and BTC. It was found that there is a bidirectional causality relationship between XLM and DOGE cryptocurrencies. Finally, a bidirectional causality relationship existed between XRP and DOGE.

On the other hand, a one-way causality relationship existed between ADA cryptocurrency and XLM and XRP cryptocurrencies. It was found that there is a one-way causality from BNB

cryptocurrencies to the cryptocurrencies ADA, ETH, XLM, and XRP. Similarly, it was found that there is a one-way causality relationship between BTC to ADA, XLM, and XRP. A unilateral and persistent causality relationship exists between ETH to BTC, DOGE, XLM, and XRP.

When the results of the frequency domain causality test in Table 2, Table 3, and Table 4 are evaluated together, it can be concluded that there are spillover and contagion effects between cryptocurrency markets. It can be observed that the cryptocurrency with the strongest contagion and spillover effect in the short and medium term is Binance Coin (BNB). Also, a contagion and spreading effect can be seen in Binance Coin and other cryptocurrency assets in the long term. Moreover, another conclusion is that the said effect is permanent. On the other hand, although Stellar (XLM) and Ripple (XRP) cryptocurrencies have a contagion and spread impact of these cryptocurrencies to other cryptocurrencies is weak. Therefore, it can be observed that the risk of Stellar (XLM) and Ripple (XRP) spreading to other cryptocurrencies is low. Another result of the frequency domain causality test is that the cryptocurrency DOGE has the highest contagion and propagation effects among other cryptocurrencies. In other words, the cryptocurrency DOGE has a very high degree of dependence on other cryptocurrencies and has the highest risk of propagation. Finally, Bitcoin (BTC) and Ethereum (ETH) have a contagion and spillover effect that causes the prices of other cryptocurrencies to change.

In contrast, the degree of influence of other cryptocurrencies is low. Bitcoin (BTC) and Ethereum (ETH) are independent cryptocurrencies with spillover effects but low impact. In conclusion, the obtained results confirm the correctness of the H2 hypothesis, which states that assets in the cryptocurrency market can directly influence each other.

5. Conclusion

The globalization process that has taken place in the financial markets in recent years has put on the agenda the need for alternative currency systems and new financial instruments. This situation has led to the emergence of cryptocurrencies, especially following the 2008 crisis. Cryptocurrencies have started to attract attention in the financial system with their advantages, such as the alternative monetary system they offer and the potential to generate high returns. Moreover, the existing regulations in the cryptocurrency market are still in their infancy, which makes the financial assets in the cryptocurrency market vulnerable to high volatility and speculative developments. In this context, the speculative behaviours observed in the cryptocurrency market may lead to price bubbles. Moreover, the correlation and high volatility caused by the interdependence of financial assets in the cryptocurrency market can lead to spreading risks. Therefore, determining the interdependence of speculative behaviour and crypto assets is necessary for risk management and stable pricing in the cryptocurrency market.

In the study, Phillips et al. (2011) (SADF) and Phillips et al. (2015) generalized the Dickey-Fuller test (GSADF) used to determine whether the external factors affecting the cryptocurrency

market cause instability in price formation. In other words, it aims to determine whether the speculative behaviour of the assets in the cryptocurrency market creates a price bubble and to measure the interdependence and spillover effect of the assets in the cryptocurrency market using the frequency domain causality test. Therefore, the study estimates the speculative behaviour and spread risk in the cryptocurrency market in two dimensions. The study results show that it is statistically significant for cryptocurrencies ADA, BNB, BTC, DOGE, ETH, XLM, and XRP. Therefore, there is empirical evidence of the formation of speculative bubbles between January 1, 2018, and December 2021, which is discussed in the study. On the other hand, when examining the SADF and GSADF test statistics obtained for the XLM and XRP currencies from the selected assets in the cryptocurrency market, it was found that the SADF test is not, while the GSADF test is statistically significant. Based on the view that the GSADF test is more consistent and provides better results than the SADF test, it can be said that empirically a speculative bubble occurred within the XLM and XRP currencies for the analyzed periods.

On the other hand, the results of the frequency domain causality test in the study provide empirical evidence that there is a spillover and contagion effect between financial assets in the cryptocurrency market. In other words, price changes between selected currencies in the cryptocurrency market cause increased correlation and volatility. In particular, the degree of pegging the cryptocurrency DOGE to other cryptocurrencies was relatively high. Stellar (XLM) and Ripple (XRP) cryptocurrencies also have a high degree of pegging to other cryptocurrencies in the short, medium, and long term. However, the price changes observed in Stellar (XLM) and Ripple (XRP) cryptocurrencies do not affect other crypto assets. In other words, when a market event occurs in the Stellar (XLM) and Ripple (XRP) cryptocurrencies, it has little potential to cause an upward or downward trend in the prices of other cryptocurrencies. Therefore, sudden changes in other cryptocurrencies can be expected to simultaneously affect DOGE, Stellar (XLM), and Ripple (XRP) and increase the risk of a spread. Another important finding of the study is that Bitcoin (BTC) and Ethereum (ETH) have a contagion and spillover effect that causes the prices of other cryptocurrencies to change. In contrast, the degree of influence by other cryptocurrencies is low. Bitcoin (BTC) and Ethereum (ETH) are cryptocurrencies in their own right that pose spillover risks but are only affected by spillover and volatility risks to a small extent.

In the context of the results obtained in the study, assets in the cryptocurrency market have a spillover effect in the form of overvaluation with the impact of internal and external factors. In other words, the high interdependence of crypto assets in the crypto money market is a significant obstacle to stable price formation when supported by speculative pricing behaviour. Therefore, the study results provide a better understanding of the interconnectedness of assets in the cryptocurrency market and the transmission of contagion effects. In addition, the study's findings indicate that investors should pay attention to the moving signals in the markets. This means that any current and past change in one cryptocurrency could have a negative impact on the movement of other cryptocurrencies. Therefore, the study's findings to establish a dynamic early warning mechanism for risk management and stable pricing in the cryptocurrency market will be an essential guide.

Future studies may expand the scope of work with other currencies in the cryptocurrency market. In addition, the studies on this topic can use quantitative methods to investigate the macroeconomic and socioeconomic factors determining the cryptocurrency market's spread risk.

References

- Başarır, Ç. (2018). Korku Endeksi (VİX) ile Bist 100 arasındaki ilişki: Frekans alanı nedensellik analizi. Dokuz Eylül Üniversitesi İşletme Fakültesi Dergisi, 19(2), 177-191.
- Bodart, V. ve Candelon, B. (2009). Evidence of interdependence and contagion using a frequency domain framework. Emerging Markets Review, 10(2), 140–150.
- Bouri, E., Das, M., Gupta, R. and Roubaud, D. (2018). Spillovers between Bitcoin and other assets during bear and bull markets. Applied Economics, 50(55), 5935-49.
- Bouri, E., Shahzad, S.J.H. and Roubau, David. (2019). Co-explosivity in the cryptocurrency market. Finance Research Letters, 29, 178-183.
- Bozoklu, S., & Yilanci, V. (2013). Energy consumption and economic growth for selected OECD countries: Further evidence from the Granger causality test in the frequency domain. Energy Policy, 63, 877-881.
- Breitung, J. ve Candelon, B. (2006). Testing for short and long-run causality: A frequency domain approach. Journal of Econometrics, 132(2), 363–378.
- Buğan, M.F. (2021). Bitcoin ve altcoin kripto para piyasalarında finansal balonlar. Akademik Araştırmalar ve Çalışmalar Dergisi, 13(24), 165-180.
- Çağlı, E. Ç., & Evrim, P. (2017). Borsa İstanbul'da Rasyonel BalonVarlığı: Sektör Endeksleri Üzerine Bir Analiz. Finans Politik ve Ekonomik Yorumlar, (629), 63-76.
- Canh, N.P., Wongchoti, U., Thanh, S.D. & Thonga, N.T. (2019). Systematic risk in cryptocurrency market: Evidence from DCC-MGARCH model. Finance Research Letters, 29, 90-100.
- Cao, G. & Xie, W. (2022). Asymmetric dynamic spillover effect between cryptocurrency and China's financial market: Evidence from TVP-VAR based connectedness approach. Finance Research Letters, 49(103070), 1-10.
- Cheah, E.T. and Fry, J. (2015). Speculative bubbles in Bitcoin markets? An empirical investigation into the fundamental value of Bitcoin. Economics Letters, 130, 32-36.
- Cheung, A., Roca, E. & Su, J.J. (2015). Crypto-currency bubbles: An application of the Phillips–Shi–Yu (2013) methodology on Mt. Gox bitcoin prices. Applied Economics, 47(23), 2348–58.
- Ciner, Ç. (2011). Eurocurrency interest rate linkages: A frequency domain analysis. International Review of Economics and Finance, 20(4), 498–505.
- Elsayed, A.H., Gozgor, G. & Lau, C. K. M. (2020). Causality and dynamic spillovers among cryptocurrencies and currency markets. International Journal of Fİnance and Economics, 27(2), 2026-2040.
- Enoksen, F.A., Landsnes, C. J., Lučivjanská, K. & Molnár, P. (2020). Understanding risk of bubbles in cryptocurrencies. Journal of Economic Behavior & Organization, 176, 129-144.
- Evlimoğlu, U. and Güder, M. (2021). Tarihteki ekonomi balonlar ışığında kripto paralara genel bakış. Abant Sosyal Bilimler Dergisi, 21(3), 469-496.
- Francés, C. J., Carles, P. G., & Arellano, D. J. (2018). The cryptocurrency market: A network analysis. Esic Market Economics and Business Journal, 49(3), 569-583.
- Fry, J. & Cheah, E.T. (2016). Negative bubbles and shocks in cryptocurrency markets. International Review of Financial Analysis, 47, 343-352.

- Geweke, J. (1982). Measurement of linear dependence and feedback between multiple time series. Journal of the American statistical association, 77(378), 304-313.
- Gharib, C., Mefteh-Wali, S. & Jabeur, S.B. (2021). The bubble contagion effect of COVID-19 outbreak: Evidence from crude oil and gold markets. Finance Research Letters, 38, 1-10.
- Giudici, P., & Abu-Hashish, I. (2019). What determines bitcoin exchange prices? A network VAR approach. Finance Research Letters, 28, 309-318.
- Homm, U., & Breitung, J. (2012). Testing for speculative bubbles in stock markets: a comparison of alternative methods. Journal of Financial Econometrics, 10(1), 198-231.
- Hosoya, Y. (1991). The decomposition and measurement of the interdependency between second-order stationary processes. Probability theory and related fields, 88(4), 429-444.
- Huynh, T.L.D. (2019). Spillover risks on cryptocurrency markets: A look from VAR-SVAR granger causality and student's-t copulas. J. Risk Financial Management, 12(2), 1-19.
- Ji, Q., Bouri, E., Lau, C.K.M. & Roubaud, D. (2019) Dynamic connectedness and integration among large cryptocurrencies. International Review of Financial Analysis, 63, 257-272.
- Katsiampa, P., Corbet, S. & Lucey, B. (2019). Volatility spillover effects in leading cryptocurrencies: A BEKK-MGARCH analysis. Finance Research Letters, 29, 68-74.
- Kim, M.J., Canh, N.P. & Park, S.Y. (2021). Causal relationship among cryptocurrencies: A conditional quantile approach. Finance Research Letters, 42, 1-8.
- Kırıkkaleli, D., Çağlar, E. & Onyibor, K. (2020). Crypto-currency: Empirical evidence from GSADF and wavelet coherence techniques. Accounting, 6, 199-208.
- Kristoufek, L. (2013). Bitcoin meets Google Trends and Wikipedia: Quantifying the relationship between phenomena of the internet era. Nature Scientific Reports, 3: 3415, 1-7.
- Maouchi, Y., Charfeddine, L., and Montasser G. (2022). Understanding digital bubbles amidst the COVID-19 pandemic: Evidence from DeFi and NFTs. Finance Research Letters, 47(102584), 1-8.
- Moratis, G. (2021). Quantifying the spillover effect in the cryptocurrency market. Finance Research Letters, 38, 101534.
- Panagiotidis, T., Stengos, T., & Vravosinos, O. (2018). On the determinants of bitcoin returns: A LASSO approach. Finance Research Letters, 27, 235-240.
- Phillips, P. C. B., Wu, Y. & Yu, J. (2011). Explosive behavior in the 1990s Nasdaq: when did exuberance escalate asset values? International Economic Review, vol. 52, no. 1, pp. 201–26.
- Phillips, P.C.B., Shi, S. & Yu, J. (2015). Testing for multiple bubbles: historical episodes of exuberance and collapse in the S&P 500. International Economic Review, 56(4), 1043-1077.
- Polat, O. & G. Eş-Polat (2022), "Kriptopara Bağlantılılığı ve COVID-19: DieboldYılmaz ve Frekans Bağlantılılığı Yöntemleri", Sosyoekonomi, 30(51), 283-300.
- Şahin, E.E. (2020). Kripto para fiyatlarında balon varlığının tespiti: Bitcoin, IOTA ve Ripple örneği. Selçuk Üniversitesi Sosyal Bilimler Enstitüsü Dergisi, 43, 62-69.
- Shahzad, S. J. H., Anas, M., and Bouri, E. (2022). Price explosiveness in cryptocurrencies and Elon Musk's tweets. Finance Research Letters. 47(102695), 1-11.
- Smales, L. A. (2019). Bitcoin as a safe haven: Is it even worth considering?. Finance Research Letters, 30, 385-393.
- Souza, M.C., Souza, E.T.C. & Pereira, H.C.I. (2017). Cryptocurrencies bubbles: New evidences. The Empirical Economics Letters, 16(7), 739-746.
- Su, C.W., Li, Z.Z., Tao, R. & Si, D.K. (2018). Testing for multiple bubbles in bitcoin markets: A generalized sup ADF test. Japan and the World Economy, 46, 56-63.

- Tarı,R., Abasız, T. & Pehlivanoğlu, F. (2012). TEFE (ÜFE) TÜFE fiyat endeksleri arasındaki nedensellik ilişkisi: Frekans alanı yaklaşımı" Akdeniz İ.İ.B.F. Dergisi, (24), 1 15.
- Waters, G.A. & Bui, T. (2021). An empirical test for bubbles in cryptocurrency markets. Journal of Economics and Finance, 1-14.
- Yanık, S. and Aytürk, Y. (2011). Rational speculative bubbles in Istanbul Stock Exchange. Muhasebe ve Finansman Dergisi, 51, 175-190.
- Yermack, D. (2015). Is Bitcoin a real currency? An economic appraisal. D.L.K. Chuen (Ed.), In: Handbook of Digital Currency, (pp 31-43), Cambridge: Elsevier.
- Yi, S., Xu, Z. & Wang, G.J. (2018). Volatility connectedness in the cryptocurrency market: Is Bitcoin a dominant cryptocurrency? International Review of Financial Analysis, 60, 98–114.

ARAŞTIRMA MAKALESİ / RESEARCH ARTICLE

THE EXTENT OF HOUSEHOLD POVERTY IN AFGHANISTAN: A CASE STUDY OF MAZAR-I-SHARIF CITY, BALKH PROVINCE (2019/20 AND 2020/21)

Baqir KHAWARI®*

Abstract

There is limited literature review and analysis of poverty in Afghanistan, particularly in the analysis of an urban area. Therefore, due to the limited information on the extent of poverty in Mazar-i-Sharif city especially at the micro/household level, this paper will provide such information and a more current one. To conduct the study, an actual data of 1060 households in Mazar-i-Sharif, obtained from a strictly random process, is used and applied the "Foster-Greer-Thorbecke (FGT)" measures of poverty to analyse poverty based on income and expenditure approach in two waves, before "COVID-19" (March 21, 2019-March 20, 2020) and during "COVID-19" (March 21, 2020-March 21, 2021). Also, the "Independent t-test" is applied to compare the mean of poverty indices in wave 1 compared to wave 2. It is found that, overall, the poverty rate is high in Mazar-i-Sharif, and more than two-thirds of the population severely suffers from the phenomenon, and it increased during the pandemic compared to pre-pandemic time. Also, the depth and severity of poverty are also serious issues and the indices increased in wave 2 compared to wave 1. Further, the study suggests that the government and international organizations should do urgent actions to save million lives and to overcome of this phenomenon.

Keywords: Headcount Ratio, Poverty Gap, Pandemic, Afghanistan **JEL Classification:** O150, O120, I32, N15

I. Introduction

Economic development is highly desired by many developing countries, including Afghanistan. The first president of Afghanistan, Hamid Karzai, has developed the "Interim Afghanistan National Development Strategy (I-ANDS)" for 15 years to achieve Afghanistan's Millennium Development Goals (MDGs) in 2020 (ANDS, 2005). Therefore, based on the strategy, many projects have been implemented by the Afghan government and foreign direct investment (FDI) to achieve the goal, unfortunately, the country is still so far from its MDGs and severely involved with a serious and dark phenomenon, poverty. According to Mohsen et al. (2021), Afghanistan is one of the poorest countries in the world.

^{*} Samangan University, Aybak, Afghanistan, E-mail: baqirkhawari1991@gmail.com, ORCID: 0000-0001-5142-4784

The poverty rate in Afghanistan has increased over time. According to "National Risk and Vulnerability Assessment" (NRVA), in 2007/08, the poverty headcount rate was 34%, raised to 38% in 2011/12 (NRVA, 2009, 2012), and 54.5% in 2017 (CSO, 2018). However, based on "Income, Expenditure, and Labor Force Survey" (IE&LFS 2020), the poverty rate decreased from 54.5% to 47% in 2019 (NSIA, 2021), in 2020, due to the "COVID-19 pandemic", the poverty rate again dramatically increased. The World Bank estimated that approximately 15 million persons were susceptible to the COVID-19 lockdown (April-June) in Afghanistan. The analysis showed that, in urban areas, the poorest percentile of the households experienced about a 35% decrease in their consumption while the richest percentile experienced about a 19% reduction. In contrast, in rural areas, the consumption of the poorest percentile of the households was reduced by about 21%, while it was estimated to be about 24% for the richest percentile (Cancho & Pradhan, 2020). As a result, the pandemic caused the poverty rate to increase from 55% in 2017 to 72% in 2020 (United Nations, 2021).

Besides, in July 2020, the ex-president of Afghanistan, Dr. Mohd. Ashraf Ghani, announced that 90% of Afghan people are below the poverty line, \$2 per person/day (AFN154), (Omid, 2020). Notably, the rate is expected to increase because of the recent political changes that have pushed the country into a predicament situation (UNDP, 2021a). On 15 August 2021, the Taliban took the power in Afghanistan, where its economy was already facing more developmental challenges such as insecurity, severe drought conditions which negatively affected agriculture production, and COVID-19, of which the third wave started in April 2021. Hence, the United Nations Development Program (UNDP) reported that about 97% of Afghans' citizens would dive below poverty line by mid-2022 (UNDP, 2021b). Also, World Food Program (WFP) announced that 95% of Afghans do not have enough food to survive (WFP, 2021). Furthermore, based on the UNDP (2021) report, the poverty gap was estimated to be 13.5% in 2019, increasing to 21% in 2020 because of the pandemic. It is predicted that the poverty gap will rise to 30% by mid-July 2022 if the poverty gap will be more than double in 2022 compared to 2019, and it needs urgent actions to save lives.

Besides, the poverty rate differs among the regions. For instance, according to the World Bank estimation, the headcount ratio for the southwest and central provinces was 0%-30%, for the north, west, and south provinces were 30%-40%, and for the western central, north-eastern, and eastern provinces of Afghanistan was 40%-50%. Also, it was mentioned that poverty concentration is highest in more urbanized and densely populated provinces (World Bank, 2015). So, this study focuses on one of the biggest and most populated cities, Mazar-i-Sharif. It is the provincial capital of Balkh province that was ranked as the fourth most populated province in Afghanistan in 2021 (NSIA, 2021). It may seem significant to analyze poverty in the capital of the province because poverty was reported to be relatively high in the province and its capital (NRVA, 2009). For instance, the poverty rate and poverty gap were estimated to be between 61%-76% and 14.8%-18.5%, respectively, in Balkh province in 2008 (NRVA, 2009).

Hall (2011) considered the per capita monthly total consumption of AFN1,289 and estimated that the poverty rate in Balkh province was 60.3%. Later, Hall (2014) conducted a research on poverty in different cities of Afghanistan and released that poverty in Mazar-i-Sharif city is serious and almost 81% of its inhabitants are under the poverty line (AFN1,710 per person/ month) based on expenditure approach and 83.5% based on the income approach. To go further, "Afghanistan Research and Evaluation Unit" (AREU) reported that the poverty rate, poverty gap, and Gini coefficient for the province's consumption were estimated to be 60.3%, 17.4%, and 27%, respectively (World Bank, 2013). Temory (2017) found that the Gini coefficient for Balkh province was 0.25 or 25%. Also, the income distribution inequality in the province was found to be 20.33% between the bottom 20% and the top 20%, which shows a huge gap between the first bottom quintile and the top quintile. Generally, in rural areas poverty looks more serious than urban areas; however, Kandahar, Kabul, Balkh, Herat, and Kunduz are the provinces where urban poverty is increasing because of trends in internally displaced people and returnees from abroad. It is estimated that 80% of urban poverty is distributed in these provinces (EASO, 2020).

As shown, poverty has been a big and challenging phenomenon in Afghanistan for a long time, particularly in Balkh province and its capital. Therefore, the article will provide a more current insight regarding the issue among households in the provincial capital of Balkh province of Afghanistan in two waves, 2019/20 and 2020/21, as well as examine the influence of the pandemic on the household poverty level. Besides, the article will address the subsequent research questions:

What is the nature and extent of poverty in Mazar-i-Sharif city? How does the pandemic influence the poverty level? For the study, poverty is measured based on two approaches, income and expenditure, by employing the data which are collected from 1060 households in Mazar-i-Sharif city from May-July 2021. Further, the FGT measures of poverty and "Independent t-test" are applied to analyse poverty and test the significance of poverty indicators in the two waves.

2. Basic Concept of Poverty

The social sciences have faced difficulties in agreeing to a single definition of poverty due to its complex and multi-faceted nature (Chamhuri et al., 2012). Poverty is understood as the inability to meet a least level of living (World Bank, 1990). Gass and Adetunmbi (2000) and Raji et al. (2006) define poverty as a lack of resources that prevent individuals from achieving a basic level of social rights, such as ingress to food, water, shelter and clothing. Additionally, Tirkaso and Hess (2015) assessment poverty as an absence of sufficient income to afford the purchase of essential goods and services.

Furthermore, according to the traditional perspective, persons who do not have sufficient earnings or spending to raise them beyond a sufficient minimal level are measured as poor. Poverty line is often used to refer to this threshold. According to this perspective, poverty is primarily understood in financial terms. Another way that poverty may indeed be defined is as a lack of a particular good or service, such as housing, food, or health. These aspects of poverty are often directly measurable, for instance, by assessing education or food. The capacity of the person to operate in society is the emphasis of the widest method to well-being (and poverty). Poor individuals often lack essential skills; they may not have enough money or schooling, be in poor health, feel helpless, or lack political liberties (World Bank, 2005). For instance, tracking achievement toward the Millennium Development Goals is often done using this simple monetary approach for measuring poverty (Sanchez-Martinez & Davis, 2014). The concepts of poverty given above include a variety of conditions, including absolute poverty, relative poverty, and the idea of the poverty line, which is succinctly described as follows.

2.1. Absolute Poverty

Absolute poverty is considered to be an absence of incomes essential to meet one's basic needs, including food, water, shelter, healthcare, education, and other necessities. This type of poverty is gauged by a universal baseline that does not take into account others' incomes or access to commodities, and failure to meet the baseline indicates poverty (Eskelinen, 2011). According to the United Nations (1995), absolute poverty relies on both income and access to services and is defined by extreme deprivation of essential human necessities. This type of poverty is more concerning in circumstances where there is a risk of starvation, rather than in areas where everyone has the means to provide for themselves (Ruggeri Laderchi et al., 2003).

2.2. Relative Poverty

The level of poverty experienced by a person is dependent upon how it is evaluated in comparison to the social norms of the country and culture they live in, and this can change over time (Sanchez-Martinez & Davis, 2014). The relatively poor are individuals whose earnings are lesser than those of the rest of the population, even if they can obtain an appropriate subsistence level. In other words, relative poverty refers to those who are poorer than the rest of the community. Hence, the term "relative poverty" refers to the delivery of income and, consequently, the disparity of living circumstances within a population (Demeke et al., 2003). Measuring this kind of poverty is feasible only for developed countries (Ravallion, 1992). Thus, for least developed countries (LDCs), including Afghanistan, where the largest share of its population are living in absolute poverty (UNDP, 2021a), the emphasis on relative poverty is not of primary relevance.

2.3. Poverty Line

When measuring poverty in a certain nation and determining the most effective means of poverty reduction, one is naturally drawn to a poverty line that is deemed acceptable for that country (Hagenaars & de Vos, 1988). The beginning points for examining poverty is poverty line (PL), and it is often the most disputed. Methods to calculate the PL significantly impact poverty profiles, which are used to formulate poverty alleviation initiatives. PLs provide a variety

of functions. According to Ravallion (1992), "the poverty line is the minimum level of income deemed adequate in a particular country".

Moreover, since poverty lines vary greatly across countries, the World Bank sets the international poverty line by considering the cost of living for essential food and non-food goods and services such as cloth, shelter, education, and health. Therefore, as a result, the United Nations and World Bank have chosen per individual poverty line of \$1 and \$2 per day/person for worldwide analyses, however for comparison of poverty inside a country, national poverty line will be more appropriate (United Nations, 2005). In July 2020, the Ministry of Economy of Afghanistan announced the national poverty line, \$2 per person/day (AFN154), which contains food, cloth, shelter, healthcare, and education which follows the international poverty line. It means that if a person earns less than \$2 per day, they identify as poor (Omid, 2020).

3. Measuring Poverty: Income or Expenditure Approach

The extent of poverty is largely ground on income or expenditure, which specify a person's access to goods and services. This has been a focus of a great deal of research, particularly around the United Nations' 2005 report, as it is often used to measure social and economic progress or failure. Lekobane and Seleka (2014) have argued that income and consumption are good indicators of well-being since they demonstrate a person's capability to gain the necessities of life.

According to the studies such as Beverly (1999); Mayer (1997); Mayer & Jencks (1989) and Rector et al. (1999), the income approach has been acknowledged as a viable tool for capturing the financial situation of families. It has also been seen to be advantageous when it comes to examining administration and societal well-being policies, such as food stamps, medical aid, subsidies, job assistance, and other monetary transfers (Ringen, 1988; Melkamu & Mesfin, 2016). The income method could be a good proxy for showing the ability of households to purchase basic goods and services because it measures households' resources, including individual tastes and preferences (Ali, 2019). Atkinson (1991) also stated that income is a well proxy for measuring living standards, generally difficult to quantify. Income is largely used to measure economic deprivation, and it is simpler to account accessible for much bigger samples (Meyer & Sullivan, 2003).

On the other hand, expenditure is typically a superior predictor of living standards compared to income, especially in developing nations (Boskin et al., 1998; Cutler et al., 1991; Fisher et al., 2013; Mayer & Jencks, 1993; Slesnick, 2002, 1994). Therefore, the consumption approach to measuring poverty is becoming more common (Fox et al., 2015). For instance, data from 88 developing countries showed that 36 countries used household income surveys, and 52 of them employed household expenditure surveys for measuring poverty or welfare (Ravallion, 2001). Consumption is thought to be more stable than income, especially in developing countries where income can often be subject to seasonal variation. To maintain a consistent level of utility,

households will use savings or debt to balance out their spending during years of high and low income (Atkinson et al.,1994).

This was supported by McKay & Lawson (2003) and Milanovic (1999). Milanovic (1999) stated that collecting income data is more complicated than data on consumption or expenditure of households, so the output of expenditure measurement is more accurate than income in transition countries. Duclos and Araar (2006) argued that, compared to income, consumption is a much better indicator of one's accomplishments and the ability to meet fundamental requirements. Moreover, consumption can be observed, remembered, and measured in a much more accurate way than income, and there is less of an issue with underreporting. Furthermore, it is also important in understanding the necessity of consumption when it comes to determining poverty (Grosh & Glewwe, 2000).

It is shown that each approach has its advantages, and we cannot ignore them so in this article poverty indices are measured based on the two approaches to have a better analysis of poverty in Mazar-i-Sharif city, Balkh province of Afghanistan.

4. Foster-Greer-Thorbecke (FGT) Measures of Poverty

The most well-known indicators of poverty such as headcount ratio (HCR), poverty gap (PG), and poverty gap squared (PGS) indices initially defined by Foster et al. in 1984. All of the indices are often used in research (Duniya & Rekwot, 2015) to assess the incidence, depth, and severity of poverty respectively in a society (Bellù & Liberati, 2005; United Nations, 2017). The indices can be computed based on the income or expenditure approach. So, many scholars such as Dharmadasa et al. (2018); Imran et al. (2020); Le et al. (2019); Nahar et al. (2017); Olowa et al. (2013); Shroff (2009); and Adams et al. (2008) used per capita household income while Afera (2015); de Silva (2008); Etuk et al. (2015); Duniya & Rekwot (2015); Mussa (2014); Ogwumike & Akinnibosun (2013); and Oyekale et al. (2012) used per capita household expenditure to capture the FGT indicators and measure poverty. The FGT poverty index (Pa) can be broken down to show the amount of poverty experienced by various population sub-groups and how much of the overall poverty level is due to each sub-group (Borko, 2016). The formula for the FGT measure of poverty is as follows:

$$P_{\alpha} = \frac{1}{N} \sum_{i=1}^{M} \left(\left(\frac{z - y_i}{z} \right)^{\alpha} * I(y_i < z) \right), (\alpha \ge 0)$$

The P α measure of poverty is determined by the values of indices P α , where N is the total population (or sample), M is the number of people living under PL, z is the PL, y_i is the per capita income or expenditure of the ith household, and α is a measure of the sensitivity of the index to poverty. With values greater than 0, the measure is decreased when living standards are lower.

If α is greater than 1, the greater the poverty, the more the measure is impacted by a decrease in living standards. This is considered to be "strictly convex" in incomes, while "weakly convex" is applicable to $\alpha = 1$. The indicator function I has the value of 1 if y_i is less than z and 0 if y_i is equal or greater than z. The P_α class model is described as follows:

$$P_0 = HCR$$

 $P_1 = PG$
 $P_2 = PGS$

The equations for the poverty indices are as follows respectively.

$$P_{0} = \frac{1}{N} \sum_{i=1}^{M} \left(\left(\frac{z - y_{i}}{z} \right)^{0} * I(y_{i} < z) \right) = \frac{1}{N} \sum_{i=1}^{M} I(y_{i} < z) = \frac{M}{N}$$

The P_0 is the headcount ratio (poverty incidence) that measures poverty rate. If the per capita income of the household is less than \$2¹ per day/person (AFN154), then the household is identified as poor otherwise non-poor.

$$P_1 = \frac{1}{N} \sum_{i=1}^{M} \left(\left(\frac{z - y_i}{z} \right)^1 * I(y_i < z) \right)$$

The P₁ measures the depth of poverty. It shows that how far the poor is from the poverty line.

$$P_2 = \frac{1}{N} \sum_{i=1}^{M} \left(\left(\frac{z - y_i}{z} \right)^2 * I(y_i < z) \right)$$

The P_2 measures the severity of poverty. The measure puts more weight the further a poor person's observed income falls below the poverty line.

In short, the correlation among the values of the above poverty indices is shown in following figure.

1 \$1=AFN77





Source: United Nations (2017)

5. Sample Size

Initially, the study designed a questionnaire to collect data about household income and expenditure for the two periods, before the "COVID-19 pandemic (March 21, 2019-March 20, 2020)" and during the "COVID-19 pandemic (March 21, 2020-March 21, 2021)". The respondents were asked to give information about their income and expenditure in the two periods at the time of the survey. The study intended to survey 1100 households to raise the reliability of the result but from the original sample size, 1100 households, 40 households were not surveyed because of some problem such as unwillingness to cooperate, absence of the head of a family or a man at the house, having moved out, or being unavailable at home when the interview was conducted, so it ended up having only 1060 households.

6. Sampling Technique and Procedure

The research used a multi-stage simple random sampling approach to pick 1100 houses in the study region, which may be stated as follows:

1. Mazar-i-Sharif has 12 regions (Nahiyah). So, based on the lottery method, 4 regions, 4,7,11, and 12, were selected in the first step.

2. In the second step, since each region contains some areas (Guzars), again based on the lottery method, from each region 2 Guzars were selected randomly, which made up a total number of 8 Guzars.

3. In the third stage, since each area consists of some streets, another simple random selection was made, and 7 streets were selected. This made up a total number of 56 streets.

4. The last stage involved a systematic random sampling of 19 or 20 households from each street,

making a total of 275 households for each region (Nahiyah).

The stages of the sampling procedure are shown in figure 2.



Figure 2: Stages of Sampling Procedure

According to a meeting that was held by the mayor of Mazar-i-Sharif, Abdul Haq Khurami, in May 2021, the four selected regions (Nahiyah) allocates more that 25% of the total population of the case study (484,492 people) to themselves. So, taking sample from these four regions could be a good representative of the total targeted population.

7. Descriptive Analysis

Table 1 displays the descriptive statistics of the demographic characteristics' of 1060 households head. The table shows that 88 per cent of the households' heads were male while around 12 per cent were female. Around (47%) of the households' heads fall above 50-year-old while the rest of the heads fall under the productive age group with 10% between the range of 18-28 years old and 44% between the range of 29-50 years old. In terms of marital status, a large percentage of the households' head (90 per cent) were married people, followed by singles (6%) and divorced (0.5%) and widows (3.4%).

Regarding the education background, most households' heads have primary and secondary education, lower and upper, with (30%) and (27.5%) respectively. 19.5 per cent with Islamic education and 15.5 per cent with university and above, while those who have zero level of education is 7.5%. Approximately 38 per cent of the households' head have elementary occupations, professional (14%), manager (1%), plant and machine operators and assemblers (around 9%). The rest of the households' head (38%) has some other occupations. Moreover, it indicates that 37 per cent and 33 per cent of the households' head were employed and self-employed, while out of the remaining 30 per cent, 6 per cent were unemployed, around 23 per cent were PAF², and only 1 per cent were retired heads. The majority of employed heads work in the private sector (about 78%), followed by government sectors (20%) and foreign institutions (2%).

² The sum of the two groups persons seeking work but not immediately available and persons available to work but not seeking is called the potential additional labour force (PAF)."

Variablas	No Rem	ittance	Int	ernal	Intern	ational	Both		Total	
Variables	Obs	0/0	Obs	<u>%</u>	Obs	<u>%</u>	Obs	0/0	Obs	0/0
Gender	003.	70	003.	70	003.	70	003.	70	000.	70
Male	451	97.6	234	75 7	245	87 5	7	77.8	937	88.4
Female	11	2.4	75	24.3	35	12.5	2	22.2	123	11.6
Total	462	13.6	300	24.5	280	26.4	0	0.8	1060	100.0
Age	402	45.0	507	27.2	200	20.4	,	0.0	1000	100.0
18-28	29	63	31	10.0	34	12.1	0	0.0	94	89
29-50	238	51.5	130	42.1	98	35.0	6	66 7	472	44 5
above 50	195	42.2	148	47.9	148	52.9	3	33.3	494	46.6
Total	462	43.6	309	29.2	280	26.4	9	0.8	1060	100.00
Marital Status	102	45.0	507	27.2	200	20.4		0.0	1000	100.00
Single	25	5.4	21	6.8	18	6.4	0	0.0	64	6.0
Married	425	92.0	271	87 7	252	90.0	7	77.8	955	90.1
Divorced	425	92.0	2/1	1.0	232	0.7	0	0.0	5	0.5
Separated	0	0.0	0	0.0		0.7	0	0.0	0	0.0
Widowed	12	2.6	14	4.5	8	2.0	2	22.2	36	3.4
Total	12	43.6	300	20.2	280	2.9	0	0.8	1060	100.0
Education	402	43.0	309	29.2	200	20.4	9	0.8	1000	100.0
No Education at all	25	76	20	0.7	14	5.0	0	0.0	70	75
Islamic School	00	21.4	60	9.7 10.4	14	J.0 15.7		44.4	207	10.5
Drimary School	112	21.4	00	20.1	112	10.0	2	44.4 22.2	207	20.2
Lower and Upper	115	24.5	95	50.1	112	40.0	2	22.2	520	30.2
Secondary School	140	30.3	79	25.6	69	24.6	3	33.3	291	27.5
University and above	75	16.2	47	15.2	41	14.6	0	0.0	163	154
Total	462	43.6	309	29.2	280	26.4	9	0.8	1060	100
Occupation	102	45.0	507	27.2	200	20.4	,	0.0	1000	100
Elementary										
occupations	187	40.5	103	33.3	111	39.6	1	11.1	402	37.9
Manager	7	1.5	4	1.3	1	0.4	0	0.0	12	1.1
Professional	69	14.9	38	12.3	40	14.3	1	11.1	148	14.0
Plant and machine										
operators, and	50	10.8	20	6.5	16	5.7	6	66.7	92	8.7
assemblers										
Others	149	32.3	144	46.6	112	40.0	1	11.1	406	38.3
Total	462	43.6	309	29.2	280	26.4	9	0.8	1060	100.0
Status of Employment										
Employed	211	45.7	96	31.1	87	31.1	1	11.1	395	37.3
Self-Employed	156	33.8	97	31.4	98	35.0	1	11.1	352	33.2
Unemployed	23	5.0	16	5.2	26	9.3	0	0.0	65	6.1
Retired	4	0.9	2	0.6	1	0.4	0	0.0	7	0.7
PAF	68	14.7	98	31.7	68	24.3	7	77.8	241	22.7
Total	462	43.6	309	29.2	280	26.4	9	0.8	1060	100.0

Table 1: Characteristics of the Households' Head

Sector or Institution of Employment										
Government Sector	41	19.4	22	22.9	16	18.4	1	100.0	80	20.3
Private Sector	166	78.7	71	74.0	70	80.5	0	0.0	307	77.7
Foreign Institution(s)	4	1.9	3	3.1	1	1.1	0	0.0	8	2.0
Total	211	53.4	96	24.3	87	22	1	0.3	395	100.0

Table 2 indicates other related factors, which are also important as far as the characteristics of the households' heads are concerned. As can be seen from the table, 96.5 per cent of the households do not have any disabled person in their families, while 3.5 per cent represent having at least one disabled person in their family. In terms of households' structure, it is demonstrated that almost half (49%) of the sample size has more females compared to males in their families, while 25 per cent of the households have a male majority. In the rest of the households (26%), the number of males and females are equal. Interestingly, it is shown that in all categories of the households, the number of households with a female majority is greater than the other two groups. Besides, half (50%) of the sample size have more than six members in their families, around 41 per cent have a family size between the range of 4-6 people, and a low percentage of the sample size (9%) have a family size between range of 1-3 people. There is 19 per cent of the sample size have received assistance from the ex-government and NGOs since 21 March 2020, while 81 per cent receive nothing. Households who received the assistance reported that most of them (81%) received non-cash assistance than cash assistance (5%); around 14 per cent of them received both types of assistance due to COVID-19.

In addition, non-cash assistances include food and non-food goods such as clothes, coal, and wood. So, 70.7 per cent of the assistance's recipients received food while only 0.5% received non-food, and the rest (28.9%) received both types of the assistance. In terms of Zakat, 2 out of 1060 households received the Islamic assistance; however, our country is an Islamic country. Thus, the government should have a special look at these Islamic elements, which significantly affects poverty reduction in a country. Finally, it represents that around 11 per cent of the households take a loan to provide the basic needs while 81 per cent of them do not take a loan for daily needs.

Variables	No Remittance		Internal Remittance		International Remittance		Both		Total	
	Obs.	%	Obs.	%	Obs.	%	Obs.	%	Obs.	%
Disability										
Yes	13	2.8	13	4.2	11	3.9	0	0.0	37	3.5
No	449	97.2	296	95.8	269	96.1	9	100.0	1023	96.5
Total	462	43.6	309	29.2	280	26.4	9	0.8	1060	100.0
HH Formation										
Male Majority	97	21.0	101	32.7	68	24.3	3	33.3	269	25.4
Female Majority	227	49.1	155	50.2	131	46.8	2	22.2	515	48.6

Table 2: Other Important Characteristics of the Households

Female = Male	138	29.9	53	17.2	81	28.9	4	44.4	276	26.0
Total	462	43.6	309	29.2	280	26.4	9	0.8	1060	100.0
Household size										
1-3 people	22	4.8	34	11.0	37	13.2	0	0.0	93	8.8
4-6 people	178	38.5	130	42.1	120	42.9	3	33.3	431	40.7
above 6	262	56.7	145	46.9	123	43.9	6	66.7	536	50.6
Total	462	43.6	309	29.2	280	26.4	9	0.8	1060	100.0
Received Assist.										
Because of Covid-19										
Yes	44	9.5	70	22.7	84	30.0	6	66.7	204	19.2
No	418	90.5	239	77.3	196	70.0	3	33.3	856	80.8
Total	462	43.6	309	29.2	280	26.4	9	0.8	1060	100.0
Type of Assistances										
Cash	0	0.0	8	11.4	3	3.6	0	0.0	11	5.4
Non-Cash	41	93.2	49	70.0	69	82.1	6	100.0	165	80.9
Both	3	6.8	13	18.6	12	14.3	0	0.0	28	13.7
Total	44	21.6	70	34.3	84	41.2	6	2.9	204	100.0
Type of non-cash										
assistance(s)										
Food	35	79.5	50	71.4	58	69.0	1	16.7	144	70.6
Non-food	1	2.3	0	0.0	0	0.0	0	0.0	1	0.5
Both	8	18.2	18	25.7	28	33.3	5	83.3	59	28.9
Total	44	21.6	70	34.3	84	41.2	6	2.9	204	100
Received Zakat										
Yes	1	0.2	1	0.3	0	0.0	0	0.0	2	0.2
No	461	99.8	308	99.7	280	100.0	9	100.0	1058	99.8
Total	462	43.6	309	29.2	280	26.4	9	0.8	1060	100.0
Taking loan for										
basic needs										
Yes	38	8.2	41	13.3	35	12.5	0	0.0	114	10.8
No	424	91.8	268	86.7	245	87.5	9	100.0	946	89.2
Total	462	43.6	309	29.2	280	26.4	9	0.8	1060	100.0

8. Result and Discussion

Table 3 shows the means of poverty indices based on income and expenditure approaches for the two periods, 2019/20 and 2020/21. First, based on income approach, the result indicates that about 70% of the household are below the poverty line of \$60 per month/person (AFN 4620) before the pandemic time. In contrast, the rate has increased to 77% during the pandemic time which shows a 7% increase in the headcount ratio. In terms of poverty gap, the finding demonstrates that, before the pandemic, the PG estimated to be 24% while during the pandemic time the PG increased by 4.5%. The severity of poverty is calculated 11% in the period of 2019/20 while in the next period, 2020/21, the severity of poverty increased to 13%.
On the other hand, based on expenditure approach, the table represents that the headcount ratio was about 76% in wave 1 while the rate increased to around 88% in wave 2. The poverty gap is estimated to be around 25% before the pandemic and 34% during the pandemic, which shows a 9% increase. In addition, the squared of poverty gap estimated about 11% in the first wave and 16% in the second wave which shows the inequality among the poor themselves are high as well that should be considered in policy making.w

	I	ncome Based App	proach	Expenditure Based Approach				
Mean of Poverty Indices	2019/20 (Pre-C)	2020/21 (D-C)	t-test (Pre-C vs D-C)	2019/20 (Pre-C)	2020/21 (D- C)	t-test (Pre-C vs D-C)		
HCR	69.6%	77.3%	(-3.99)***	75.7%	87.8%	(-7.34)***		
PG	24.2%	28.7%	(-4.42)***	24.7%	34.4%	(-10.57)***		
PGS	11.2%	13.4%	(-3.41)***	10.7%	16.2%	(-9.25)***		

 Table 3: Poverty Profile of Mazar-i-Sharif

(***), (**), and (*) represent the level of significance at 1%, 5%, and 10% respectively based on the result of Independent t-test.

Note: Pre-C = Pre-COVID-19; and D-C = During-COVID-19.





In short, the findings of the study present that, either use income or expenditure approach, poverty indices increased during the pandemic time compared to before the pandemic which is clearly shown in figure 3. It means that during the pandemic more households slip into poverty

compared to pre-pandemic time; the PG and PGS indices also significantly increased in wave 2 compared to wave 1. Moreover, according to the independent t-test, all changes between wave 1 and wave 2 poverty indicators are statistically significant.

Figure 4 illustrates the differences between income and expenditure approach. It shows that in both periods, 2019/20 and 2020/21, poverty indices are estimated to be greater based on expenditure approach compared to income approach. It is because some portion of the household income might shift to saving and payment of loan that decrease the household expenditure (consumption). In addition, the figure represents that the differences between poverty indices based on expenditure approach compared to income approach is more in wave 2 than wave 1 and it is because during the pandemic households may more interested to shift higher portion of their income into saving to use it later in urgent time. Therefore, we can conclude that since households may save more during the pandemic, it is better to measures poverty indices based on income approach rather than expenditure.





9. Conclusion

Poverty has been a significant obstacle for the Afghan administration for quite some time. As a result, most of the population severely suffer from this phenomenon. In this study, we measured poverty indices based on two approaches, income and expenditure, in two different periods, before COVID-19 and During COVID-19, for provincial capital of Balkh province, Mazar-i-Sharif city. Overall, the findings indicate that poverty is a serious issue in Mazar-i-Sharif city and almost two third of its population are below the poverty line in each period. In addition, it is found that either use income or expenditure approach, the poverty indices are estimated to be

high during the pandemic compared to pre-pandemic time. During the pandemic not only, more households fell into poverty but also the depth and severity of poverty among poor households also relatively increased. Furthermore, the result shows that poverty indices are estimated to be greater based on expenditure approach than income, and especially during the pandemic. Hence, it is matter whether use income or expenditure approach in measuring poverty indicators, particularly during the pandemic or other economic shock.

Besides, the main reason for poverty in Afghanistan is poor governance. Because for the last two decades a significant amount of money (\$77 billion) was injected in the country through Official Development Aid (ODA) and around \$2 billion was inflowed between 2002-2019 through FDI to develop the economy, but still, millions of people are suffering economically. So, it shows that the Afghan government did not achieve well, particularly in terms of poverty. Thus, the current study suggests that to reduce poverty rate in the country, Afghan government should focus on how to make a good governance and reduce corruption, economic and political instability to enhance the growth.

References

- Adams, R., Lopez-Feldman, A., Mora, J., Taylor, J. E., DeWind, J., & Holdaway, J. (2008). Remittances, inequality and poverty: Evidence from rural Mexico. *Migration and development within and across borders: Research and policy perspectives on internal and international migration*, 101-130.
- Afera, N. (2015). Determinants of poverty in rural Tigray: Ethiopia evidence from rural households of Gulomekeda Wereda. *Journal of Poverty, Investment and Development*, 10(1), 95-102. https://core. ac.uk/download/pdf/234695199.pdf
- Ali, A. F. M. (2019). Difference Between Income and Expenditure Method in Measuring Poverty in Kelantan, Malaysia. International Journal of Economics, Management and Accounting, 27(1), 1-40. https://journals.iium.edu.my/enmjournal/index.php/enmj/article/view/594/314
- ANDS. (2005). Afghanistan National Development Strategy: An Interim Strategy for Security, Governance, Economic Growth & Poverty Reduction. Accessed on September 8, 2022. https://reliefweb.int/sites/ reliefweb.int/files/resources/AFA4970B33A0505E492.571.07000811C6-unama-afg-30jan2.pdf
- Atkinson, A. B. (1991). Comparing poverty rates internationally: lessons from recent studies in developed countries. *The World Bank Economic Review*, 5(1), 3-21. https://doi.org/10.1093/wber/5.1.3
- Atkinson, A. B., Rainwater, L., & Smeeding, T. M. (1994). Income distribution in European countries (No. 121). LIS Working Paper Series. https://doi.org/10.1016/S0305-750X(01)00072-9
- Bellù, L. G., & Liberati, P. (2005). Impacts of Policies on Poverty. Basic Poverty Measures. https://mpra. ub.uni-muenchen.de/id/eprint/44666
- Beverly, S. G. (1999). *Economic poverty reconsidered: Material hardship and income-poverty in the United States*. Washington University in St. Louis.
- Borko, Z. P. (2016). Determinants of Poverty in Rural Households (The Case of Damot Gale District in Wolaita Zone) A Household Level Analysis, *International Journal of African and Asian Studies*, V(29), https://core.ac.uk/download/pdf/234690246.pdf

- Boskin, M. J., Dulberger, E. L., Gordon, R. J., Griliches, Z., & Jorgenson, D. W. (1998). Consumer prices, the consumer price index, and the cost of living. *Journal of economic perspectives*, 12(1), 3-26. https:// dio.org/10.1257/jep.12.1.3
- Cancho, C. A., & Pradhan, T. (2020, August 16). Mitigating the poverty implications of COVID-19 in Afghanistan. Published on End Poverty in South Asia. https://blogs.worldbank.org/ endpovertyinsouthasia/mitigating-poverty-implications-covid-19-afghanistan
- Chamhuri, N. H., Karim, H. A., & Hamdan, H. (2012). Conceptual framework of urban poverty reduction: A review of literature. Procedia-Social and Behavioral Sciences, 68, 804–814. doi:10.1016/j. sbspro.2012.12.268
- CSO. (2018). Afghanistan Living Conditions Survey 2016-17. Central Statistic Organization: Kabul, Afghanistan. Accessed on September 7, 2022. https://washdata.org/sites/default/files/documents/ reports/2018-07/Afghanistan%20ALCS%202016-17%20Analysis%20report.pdf
- Cutler, D. M., Katz, L. F., Card, D., & Hall, R. E. (1991). Macroeconomic performance and the disadvantaged. Brookings papers on economic activity, 1991(2), 1-74. https://doi.org/10.2307/2534589
- De Silva, I. (2008). Micro-level determinants of poverty reduction in Sri Lanka: a multivariate approach. International Journal of Social Economics, 35(3), 140-158. https://doi. org/10.1108/030.682.90810847833
- Demeke, M., Guta, F., & Ferede, T. (2003). Growth, employment, poverty and policies in Ethiopia: an empirical investigation. *Issues in employment and poverty Discussion Paper*, 12, 86. http://www.ilo. int/wcmsp5/groups/public/—-ed_emp/documents/publication/wcms_120734.pdf
- Dharmadasa, R. A. P. I. S., Weerahewa, J., & Samarathunga, P. A. (2018). Labor migration and impact of remittances on poverty and income inequality: evidence from estate sector of Sri Lanka. Tropical Agricultural Research, 30(1), 69. https://doi.org/10.4038/tar.v30i1.8279
- Duclos, J. Y., & Araar, A. (2006). Poverty and equity: measurement, policy and estimation with DAD (Vol. 2). IDRC.
- Duniya, K. P., & Rekwot, G. Z. (2015). Determinants of poverty among groundnut farming households in Jigawa State, Nigeria. Asian Journal of Agricultural Extension, Economics and Sociology, 4(3), 224-230. https://doi.org/10.9734/ajaees/2015/13539
- EASO. (2020). Afghanistan Key socio-economic indicators Focus on Kabul City, Mazar-e Sharif and Herat City. https://doi.org/10.2847/497573
- Eskelinen, T. (2011). Relative poverty. *Encyclopedia of Global Justice*. Springer Netherlands, Dordrecht, 942-943.https://doi.org/10.1007/978-1-4020-9160-5
- Etuk, E., Angba, C., & Angba, A. (2015). Determinants of poverty status of fish vendor households in Lower Cross River Basin, Nigeria. *Journal of Economics and Sustainable development*, 6(14), 50-55. https:// core.ac.uk/download/pdf/234647149.pdf
- Fisher, J. D., Johnson, D. S., & Smeeding, T. M. (2013). Measuring the trends in inequality of individuals and families: Income and consumption. *American Economic Review*, 103(3), 184-88. https://doi. org/10.1257/aer.103.3.184
- Foster, J., Greer, J., & Thorbecke, E. (1984). A class of decomposable poverty measures. *Econometrica: journal of the econometric society*, 761-766. https://doi.org/10.2307/1913475
- Fox, L., Wimer, C., Garfinkel, I., Kaushal, N., & Waldfogel, J. (2015). Waging war on poverty: Poverty trends using a historical supplemental poverty measure. *Journal of Policy Analysis and Management*, 34(3), 567-592. https://doi.org/10.1002/pam.21833

- Gass, G., & Adetunmbi, O. (2000). Summary report: DFID's civil society consultation process in Nigeria (13 December 1999–2 February 2000).UK: Department for International Development.
- Grosh, M., & Glewwe, P. (2000). *Designing household survey questionnaires for developing countries*. Washington, DC: World Bank. http://hdl.handle.net/10986/25338
- Hagenaars, A., & de Vos, K. (1988). The Definition and Measurement of Poverty. *The Journal of Human Resources* vol. 23, issue 2, 211-221. https://doi.org/10.2307/145776
- Hall, S. (2011). Economic Assessment and Labour Market Survey of Mazar-i-Sharif, Pul-i-Khumri, Kandahar City and Kunduz City.
- Hall, S. (2014). Urban Poverty Report: A Study of Poverty, Food Insecurity and Resilience in Afghan Cities. Danish Refuge Council and People In Need, Kabul, Afghanistan. https://www.clovekvtisni.cz/ media/publications/729/file/143.350.2328-pin-drc-afghanistan-2014-a-study-of-poverty-foodsecurity-and-resilience.pdf
- Imran, K., Devadason, E. S., & Kee Cheok, C. (2020). Foreign remittances and regional poverty: Evidence from household data. *International Migration*, 58(4), 214-230. https://doi.org/10.1111/imig.12681
- Le, Q. H., & Nguyen, T. H. T. (2019). Impacts of migration on poverty reduction in Vietnam: a household level study. *Business and Economic Horizons (BEH)*, *15*, 261-275. https://doi.org/10.15208/beh.2019.16
- Lekobane, K., & Seleka, T. B. (2014). Determinants of household welfare and poverty in Botswana, 2002/03 and 2009/10. http://dx.doi.org/10.2139/ssrn.2819997
- Mayer, S. E. (1997). What money can't buy: Family income and children's life chances. Harvard University Press.
- Mayer, S. E., & Jencks, C. (1989). Poverty and the Distribution of Material Hardship. Journal of Human Resources, 24(1), 88-114.https://doi.org/10.2307/145934
- Mayer, S. E., & Jencks, C. (1993). Recent trends in economic inequality in the United States: Income versus expenditures versus material well-being. In *Poverty and Prosperity in the USA in the Late Twentieth Century* (pp. 121-207). Palgrave Macmillan, London. DOI: 10.1007/978-1-349-22953-6_5
- McKay, A., & Lawson, D. (2003). Assessing the extent and nature of chronic poverty in low income countries: issues and evidence. World development, 31(3), 425-439. https://doi.org/10.1016/S0305-750X(02)00221-8
- Melkamu, M., & Mesfin, M. (2016). Interconnections Between Income and Expenditure Approach to Measure Poverty in Northern Rajasthan. International Journal of Advanced Research in Management and Social Sciences, 5(1), 89-106. https://www.academia.edu/download/41556709/25_Melkamu_ and_Mesfin.pdf
- Meyer, B. D., & Sullivan, J. X. (2003). Measuring the well-being of the poor using income and consumption. V(38), 1180-1220. https://www.nber.org/system/files/working_papers/w9760/w9760.pdf
- Milanovic, B. (1999). Explaining the increase in inequality during transition. *Economics of transition*, 7(2), 299-341. https://doi.org/10.1111/1468-0351.00016
- Mohsen, A., Sadat, S. F., & Haidery, J. (2021). Effects of the macroeconomic variables on GDP growth of Afghanistan. *International Journal of Research and Analytical Reviews (IJRAR)*, 8(2).
- Mussa, R. (2014). Household expenditure components and the poverty and inequality relationship in Malawi. *African Development Review*, 26(1), 138-147. https://doi.org/10.1111/1467-8268.12070
- Nahar, F. H., & Arshad, M. N. M. (2017). Effects of remittances on poverty reduction: the case of Indonesia. *Journal of Indonesian Economy & Business*, 32(3). https://pdfs.semanticscholar.org/ b6ad/4d171f64965d8e92ab367d8190db01547e56.pdf

- NRVA. (2009). NRVA Main Report: A Profile of Afghanistan. Kabul, Afghanistan. Accessed on 5 November 2022. https://documents.wfp.org/stellent/groups/public/documents/ena/wfp213398.pdf
- NRVA. (2012). Afghanistan Living Conditions Survey. Kabul, Afghanistan. Accessed on 5 November 2022. https://www.ilo.org/surveyLib/index.php/catalog/7922/download/48991
- NSIA. (2021). Income and Expenditure & Labor Force Surveys Report 2020. National Statistic and Information Authority, Kabul, Afghanistan. Accessed on September 6, 2022. https://fscluster.org/sites/default/files/documents/income-and-lobar-force-survey-2020-15.pdf
- Ogwumike, F. O., & Akinnibosun, M. K. (2013). Determinants of poverty among farming households in Nigeria. *Mediterranean Journal of Social Sciences*, 4(2), 365-373. https://doi.org/10.5901/mjss.2013. v4n2p365
- Olowa, O. W., Awoyemi, T. T., a Shittu, M. A., & Olowa, O. A. (2013). Effects of remittances on poverty among rural households in Nigeria. *European Journal of Sustainable Development*, 2(4), 263-284. https://doi.org/10.14207/ejsd.2013.v2n4p263
- Omid, M. H. S. (2020, June 20). Ministry Confirms 90% of Afghans Live Below Poverty Line. TOLO NEWS. https://tolonews.com/business/ministry-confirms-90-afghans-live-below-poverty-line
- Oyekale, A. S., Adepoju, A. O., & Balogun, A. M. (2012). Determinants of poverty among riverine rural households in Ogun State, Nigeria. *Studies of Tribes and Tribals*, 10(2), 99-105. https://doi. org/10.1080/0972639x.2012.118.86647
- Raji, M. O., Ayoade, O. B., & Usoro, A. (2006). The prospects and problems of adopting ICT for poverty eradication in Nigeria. The Electronic Journal of Information Systems in Developing Countries, 28. doi:10.1002/j.1681-4835.2006.tb00192.x
- Ravallion, M. (1992). Public-Private Sector Wage Differentials in Peru (Issue 42). https://www.ophi.org.uk/ wp-content/uploads/Ravallion-1992.pdf
- Ravallion, M. (2001). Growth, inequality and poverty: looking beyond averages. World development, 29(11), 1803-1815. www.elsevier.com/locate/worlddev
- Rector, R. E., Johnson, K. A., & Youssef, S. E. (1999). The extent of material hardship and poverty in the United States. *Review of Social Economy*, *57*(3), 351-387. https://doi.org/10.1080/003.467.69900000007
- Ringen, S. (1988). Direct and Indirect Measures of Poverty. Journal of Social Policy, 17(3), 351–365. https:// doi.org/10.1017/s004.727.9400016858
- Ruggeri Laderchi, C., Saith, R., & Stewart, F. (2003). Does it matter that we do not agree on the definition of poverty? A comparison of four approaches. Oxford Development Studies, 31(3), 243–274. https:// doi.org/10.1080/136.008.1032000111698
- Sanchez-Martinez, M., & Davis, P. (2014). A review of the economic theories of poverty. National Institute of Economic and Social Research (NIESR) Discussion Papers, (435). https://www.niesr.ac.uk/wpcontent/uploads/2021/10/dp435_0-4.pdf
- Shroff, K. (2009). Impact of remittances on poverty in Mexico. Yale College, USA. http://www.econ.yale.edu/~granis/web/PovertyWpaper.pdf
- Slesnick, D. T. (1994). Consumption, needs and inequality. *International Economic Review*, 677-703. https:// doi.org/10.2307/2527080
- Slesnick, D. T. (2002). [Book Review] Consumption and Social Welfare, living standards and their distribution in the United States. Journal of Sociology and Social Welfare, 29(4), 172-174.
- Temory, M. D. (2017). Determinants of Household Income in Afghanistan: An Empirical Analysis (Doctoral dissertation, Ruhr-Universität Bochum). https://d-nb.info/116.750.4739/34

- Tirkaso, W. T., & Hess, S. (2015). The role of ICT expenditure for cash crop production and income generation in Southern Ethiopia. The Electronic Journal of Information Systems in Developing Countries, 71(1), 1–14. doi:10.1002/(ISSN)1681-4835
- UNDP. (2021a, September 9). Afghanistan on the brink of universal poverty: UN. Al Jazeera and News Agencies.
- UNDP. (2021b). Economic Instability and Uncertainty in Afghanistan after August 15 A Rapid Appraisal. https://www.af.undp.org/content/afghanistan/en/home/library/knowledge-products/UNDP-report.html
- United Nations. (1995). The United Nations World Summit for Social Development: Copenhagen, 6–12 March 1995. https://digitallibrary.un.org/record/198966/files/A_CONF.166_9-EN.pdf?ln=en
- United Nations. (2021). Security Council Report: Monthly Forecast (March 2021). Accessed on September 5, 2022. https://www.securitycouncilreport.org/atf/cf/%7B65BFCF9B-6D27-4E9C-8CD3-CF6E4FF96FF9%7D/2021_03_forecast.pdf
- United Nations. (2005). Poverty Statistics: Concepts, Methods and Policy Use. United Nation. https:// unstats.un.org/unsd/methods/poverty/pdf/un_book%20final%2030%20dec%2005.pdf
- United Nations. (2017). Basic consumption and income-based indicators of economic inequalities in Bosnia and Herzegovina: evidence from household budget surveys (13). https://unece.org/fileadmin/DAM/ stats/documents/ece/ces/ge.15/2017/Expert-meeting-Montenegro-2017/Bosnia_paper_formatted. pdf
- WFP. (2021). Afghanistan Food Security Update #2. https://reliefweb.int/report/afghanistan/afghanistanfood-security-update-2-22 september-2021
- World
 Bank
 (2013).
 Afghanistan
 Economic
 Update:
 Poverty
 Reduction,
 and

 Economic
 Management.
 https://openknowledge.worldbank.org/bitstream/
 handle/10986/16510/770830REVISED0box377289B00PUBLIC00.pdf?sequence=1&isAllowed=y
- World Bank. (1990). World development report: Poverty. New York, NY: Oxford University Press.
- World Bank. (2005). Introduction to Poverty Analysis. https://numerons.files.wordpress. com/2012/04/5poverty-manual-wb.pdf
- World Bank. (2015). Afghanistan Poverty Status Update. https://documents1.worldbank.org/curated/ en/594.051.468180880731/pdf/100638-v2-WP-P145128-PUBLIC-Box393238B-Poverty-Status-Update-Report.pdf

ARAŞTIRMA MAKALESİ / RESEARCH ARTICLE

A CROSS SECTIONAL ANALYSIS ON THE IMPACTS OF ICT ON TURKISH MANUFACTURING INDUSTRY *

Yasemin ÖZBAL**®

Abstract

This paper aims to examine the impacts of information and communication technologies (ICT) on firmlevel productivity in Turkish manufacturing industry. The dataset used in this paper was obtained from merging TURKSTAT's Annual Industry and Services Survey and ICT Usage in Enterprises Survey results. This study examines 2009 and 2019 data and estimates the impacts of ICT usage and ICT using labor on labor productivity to understand if the adoption of digitalization had impacts on firmlevel productivity of the manufacturing industry throughout the ten years period. The results of the empirical analysis suggest a positive impact of ICT usage and ICT using labor on all technological levels of the manufacturing industry, however according to two-digit breakdown of manufacturing sectors indicate that only nine out of twenty-two sectors have statistically significant results on ICT usage. **Keywords:** ICT, ERP, CRM, Software, Labor productivity, ICT labor, Manufacturing Industry, Turkey **JEL codes:** D24, J24, O14

I. Introduction

Industry 4.0 or the Fourth Industrial Revolution became a topic of discussion when the German government promoted the computerization of the manufacturing industry in 2011. The "new" industrial revolution is stimulated by digital technologies, especially robots, artificial intelligence, the internet of things, 3-D printing, cloud computing, different types of software which enable companies to communicate and do business with their partners and customers, and other recent technologies. Using the methods of Information and Communication Technologies (ICT) and digital technologies, especially in the production process, lies at the heart of the Fourth Industrial Revolution.

According to a report from Boston Consulting Group (2015), the impacts of "Industry 4.0" will be significant in the next 10-15 years; the report forecasts that "*in Germany alone* "*Industry 4.0*" will

^{*} This paper was adapted from a part of Yasemin Özbal's PhD dissertation will be submitted at Marmara University, Department of Economics.

^{**} PhD Candidate, Marmara University, Department of Economics, Email: yaseminozbal@gmail.com, ORCID ID: 0000-0001-9808-4465

contribute about one percent per year to GDP over ten years, create as many as 390,000 jobs, and add \notin 250 billion to manufacturing investment (or 1 to 1.5 percent of manufacturers' revenues)."¹

A business survey conducted by Pricewaterhouse Coopers (PWC) in 2016 to more than 2000 enterprises from nine major industrial sectors and 26 countries shows that the developments in digitalization will transform enterprises as well as market dynamics. According to this survey's results, the nine industries which were surveyed and that plan to invest US\$907 billion per year globally in Industry 4.0 applications over the next five years were expecting annual digital revenue increases of 2.9 % on average and a minority of the enterprises surveyed expected 50 percent increases in their digital revenues.

By implementing information and communication technologies, organizations become more flexible in the production process and adapt themselves for user-end requirements. ICTs help provide connectivity and interoperability between organizations, their partners, and their customers through facilitating their storing, sharing, and processing information. (Perakovic, et. al. 2019).

Until 2000s, the studies exploring the contribution of ICT and digitalization to productivity did not find a deep impact on productivity. However, after the beginning of 2000s, the studies that are estimating the impacts of ICT on productivity growth have found stronger results (Stiroh, 2002; Brynjollfson and Hitt, 2003; Maliranta and Rouvinen 2004, etc.). Despite a wide range of studies on ICT and productivity in the U.S., Europe and emerging countries, there are only a few studies examining the impacts of ICT on output and productivity in Turkey. The main purpose of this study is to observe the Turkish firm data and the impacts of ICT on firm-level productivity in the Turkish manufacturing industry. It is important to analyze the effects of ICT utilization on firm level since the usage of ICT at the firms operating in manufacturing industry and other sectors became widespread in the past 10 years. This study will examine, how the spread of digitalization had impacts on manufacturing sectors, by different levels of technology. Although many authors have examined the ICT and productivity relationship, there is still a need to explore it by technological breakdown of the sectors considering the fact that different sectors might be affected in diverse levels.

In consideration of the fast adoption of digitalization by Turkish firms, this study's aim is to contribute the literature on the impacts of ICT to productivity analysis in Turkey from the perspective of Industry 4.0 and the digital transformation of Turkish firms. The study is organized as follows. The second section summarizes the related literature on the relationship between productivity and information and communication technologies. Third section provides quick facts and data on the ICT usage in Turkey in recent years. The fourth section is describing the data used in this study and the fifth section presents the results of the cross-section analysis using the data from TURKSTAT. The sixth section concludes the paper.

¹ Boston Consulting Group, "Industry 4.0 The Future of Productivity and Growth in Manufacturing Industries", 2015

2. Literature Review

While there are several studies which emphasize the relationship between digitalization and productivity globally, especially in developed countries as United States, European Union countries and Australia, there are few studies conducted in this field in Turkey.

The origin of the impacts of "computer" or the "digital transformation" on production was initiated with Robert Solow's famous quote on New York Times Book Review in 1987: "You can see the computer age everywhere but in the productivity statistics." Solow questioned the reason behind the slowdown in productivity growth in the United States and developed countries in the 1970s and 1980s despite rapid development in the field of information technology. Since then, this idea was conceptualized as the "Solow paradox" to explain the slowdown in the productivity during a period when investment in information technologies is high.

Several studies have analyzed the empirical relationship between ICT and productivity in various performance measures, such as growth, productivity, and profitability. Earlier studies assessing the impacts and contributions of information and communication technologies to firm productivity encountered problems. Especially studies that observe American firm-level productivity data of the 1970s and 1980s had experienced negative correlations with economy wide productivity and information worker productivity (Brynjolffson 1993). Brynjolffson (1993) summarized a review on 18 articles that assess the impacts of IT on manufacturing industries, services sectors and both. He explains the shortfall of IT productivity or the disappointment in IT to the firm productivity levels because of deficiencies in measurement and methodologies used in these previous studies as well as because of mismanagement by developers and users of IT.

Later on, studies in the 2000s observed a significant contribution of IT to the productivity and output growth. The research of Brynjolfsson and Hitt (2003) which focuses on the impacts of computerization to firm-level productivity in the United States between 1987-1994, find out that for a sample of large-size firms, computerization contributed to the productivity and output growth in short term which is consistent with computer investments. Moreover, IT's contribution is even higher over longer-term periods.

A study by Barker, Fuss and Wavermann (2008) analyzed Australian firm data and other 17 OECD countries within a period covering 1980 to 2003. The results of their estimations indicate that the labor productivity increased throughout the years (from 1980 to 2003) with a contribution of ICT investment (IT usage, network penetration, etc.). Besides they also examine the potential spillover network and externality effects of ICT (ICT spillovers).

Maliranta and Rouvinen (2004) explore the use of ICT in Finnish business enterprises and observe the micro-level firm data in Finland between 1992 and 2001. According to the "lower bound estimate" of excess productivity of ICT-equipped labor, the additional productivity of ICT-equipped labor ranges from 8% to 18% where this effect is much higher in younger firms and in ICT-providing activities. Another result they found out from the estimations is that the excess

productivity is somewhat higher in the services sector than the manufacturing sector where the manufacturing sector benefits from the ICT-induced efficiency through internal communication while the services sector benefits through external communication.

The studies on developing countries have mixed results. Basant et. al. (2006) study Brazilian and Indian firms by implementing a survey for a three-year period (2001-2003) and find out that in both countries, econometric evidence displays a strong relationship between ICT capital and firm productivity. Crespi & Zuniga (2012) examine the relationship between technological innovation and firm productivity in six Latin American countries (Argentina, Chile, Colombia, Costa Rica, Panama, and Uruguay) since prior studies could not establish the relationship because of survey and sampling methodologies. According to their findings, in all countries they made the research, firms that invest in knowledge and use innovation increased their labor productivity compared to other firms that did not; on the other hand, they found out that firm-level determinants of innovation investment were more heterogenous than in OECD countries.

Another study on Brazilian and Indian firms by Commander, Harrison and Menezes-Filho (2011), uses a unique data set constructed by a survey that has been implemented in both countries between April and May 2005. This study pioneers an innovative way while using the Indian firm data: It investigates the policy implementations and institutional environment on ICT capital investment and productivity. Similar to previous studies in the field; using a production function estimation, they break down the capital into two: Physical capital and ICT capital stock and also use other measures to identify the adoption of ICT with dummy variables. According to their different estimation results – in line with some evidence from studies on developed countries, there have been very high returns to ICT for both countries. Moreover, they analyze the impacts of policy and institutional environment on ICT adoption in India (since they do not have sufficient data for Brazil), and the results suggest that poorer infrastructure quality and proworker labor regulation are associated with lower levels of ICT capital intensity.

More recent studies in the 2010s, using a total factor productivity approach, estimate the contribution of ICT to productivity and output growth; while some of them find a smaller contribution to the TFP growth (Hawash, Lang, 2020), some of them have much more optimistic results where they find the significant contribution of ICT (Gal, et al., 2019). Hawash and Lang (2020), using panel data of 76 developing countries from 1991 to 2014, estimate the impact of ICT on total factor productivity (TFP) by three different approaches. In contrast to prior studies that were involved in the impacts of ICT on productivity, their results show that ICT has a limited impact on TFP growth. The estimation results reveal that both ICT investments and physical ICT usage of households' variables are significant and have a positive impact on TFP, however, it is a diminishing and modest impact.

Tambe and Hitt (2012), in their study, using a dataset they created themselves by matching firmlevel IT employee data from a large sample of information technology workers (that they collect through an online job-search website) and with production inputs for approximately 1,800 firms across 20 years (from 1987 to 2006) in the United States. Since IT-using workers are subject to endogeneity bias, they found that the endogeneity does not substantively affect current IT estimates. The second finding in their study is that large and midsize firms are doing similar IT investments, although large firms have greater marginal products from these investments, while midsize firms benefit from these investments in the short-run. Their third finding is that the marginal product of IT using workers is higher (and accelerating) in the period 2000-2006 than in the prior years (1987-1999) in firms of all sizes, which contradicts the previous works suggest that the link between IT spending and productivity may have changed since 2000 (Jorgenson et al. 2008).

Harrigan, Reshef and Toubal (2018), using French firm data between 2009 and 2013, analyze the impacts of firm-level choices of ICT, R&D, exporting and importing on the evolution of productivity. To estimate firm-level productivity, they use a methodology allowing to measure both Hicks-neutral and skill-augmenting technology differences. They measure the adoption of ICT in the firms through the workers using ICT, whom they call "techies". According to their estimation results, both employment of "techies" and offshoring (exporting and importing) are orienting the firms to employ more skilled and unskilled workers. The results of the estimation also show that in between French firms which employ "techies" have skill-augmenting productivity which is 60 percent higher compared to the firms which do not employ "techies".

Aboal and Tacsir (2018) study Uruguayan firm data to understand the determinants of investments in ICTs and in other innovation activities at the firm level in both manufacturing and services sector. To assess the Uruguayan firm level data, they use a unified econometric framework based on a version of the CDM model (based on the "Crépon, Duguet and Mairesse" study in 1998). According to their results, "The *ICTs seem to be more important for innovation and productivity in the services sector than in the manufacturing sector. Second, investment in all other innovation activities is more important for the introduction of technological innovations are more important for productivity in the services sector than in the services sector. Third, non-technological innovations are more important for productivity in the services sector than in the manufacturing sector." Their findings suggest that investment in ICT increases the probability of both technological and non-technological innovations in manufacturing. In the same direction as Alvarez (2016) and Polder et al. (2009), they find that ICT investment seems to foster innovation in the services sector.*

There are few studies focusing on Turkey where the results indicate that the impacts of ICT on firms' efficiency or productivity were positive. Atasoy, Banker and Pavlou (2016) examine the longitudinal role of IT use in the firm's total number of employees. They use Information and Communication Technologies Usage in Enterprises Survey from TURKSTAT, which covers the period of 2007-2011 and establish a panel data set from it. To analyze the effects of IT use on firm-level employment, they use a "firm fixed effects model". The aim to use this model is to identify the within-firm changes in IT use and firm-level employment over time, and not by permanent unobserved differences across firms. The estimation differs by IT application types and moderated by three factors: Firm size, average wage rate, and industry technology intensity. According to their results, the use of enterprise applications affects firm-level employment over

time, whereas the effects of the use of Web applications materialize in the current year. They found a positive relationship between IT use and firm-level employment on average, and the relationship varies by the category of IT applications.

The study of Kılıçaslan, Sickles, Kayış, and Üçdoğruk Gürel (2017) examines the impacts of ICT on labor productivity growth in the Turkish manufacturing sector. Using TURKSTAT's firm data from Annual Industry and Services Statistics, they develop a measure of the stock of capital, separating it as "conventional capital" and "ICT capital". They construct capital stock series by using the perpetual inventory methodology; they use the yearly amortization allowances to measure the capital stock and derivate the ICT investment from the investment, which includes office and computing equipment, communication equipment, and software investment. Two different models are estimated in the study, the first is the growth accounting approach, while the latter is using the generalized methods of moments method to estimate the impacts of ICT on labor productivity. According to their growth accounting model results, ICT capital has no special contribution compared to conventional capital in value-added growth in the Turkish manufacturing industry with some exceptions. However, according to the static and dynamic panel data models, the ICT capital's contribution to labor productivity in the manufacturing industry is around 15-20 percent larger than the conventional capital's contribution.

Most recently, Taştan and Gönel (2020) analyze the impacts of information and communication technologies on firm-level productivity in Turkey, using firm-level data sets and constructing an unbalanced panel data set covering the period 2007-2014. This study includes parameters to estimate the impacts of ICT, such as software investments, indicators for the usage of enterprise system applications (ERP, CRM, SCM), and ICT labor. According to their estimation results, there is a positive relationship between firm productivity and ICT use; the empirical results also support the complementarity hypothesis between ICT labor and software usage variables. They had similar results as existing studies for developed and developing countries that find a positive relationship between ICT usage and productivity. In addition, they also find out that, while the ICT investments and usage have positive returns in both manufacturing and services sectors, the effect is higher for the firms in the services sector.

In most recent study of Taştan (2021), he uses a descriptive model where he investigates the impacts of ICT on firm productivity. The ICT indicators are classified under three groups: Software, infrastructure and organizational structure. This study only observes 2017 data and estimates the impacts of ICT on labor productivity. According to the estimation results, in both manufacturing and services sectors, the intensity of ICT usage and the share of ICT using labor in total have a complementary relationship. Although this study does not imply there is a causality between ICT and productivity, the results indicate ICT using firms have relatively higher levels of productivity.

The related literature which analyzes the relationship between ICT and productivity in Turkey mostly focuses on aggregate productivity and the firm productivity on the manufacturing and services sector. This study will contribute the literature by examining the manufacturing industry

on sectoral and technological level based on NACE Rev. 2 two-digit level and according to their technological intensity based on Eurostat classification.

3. The ICT Usage in Turkey in Recent Years

In Turkey, the ICT sector started to improve and increase at a faster pace in the mid-2000s. According to Informatics Industry Association in Turkey (TUBISAD), in the past five years, the market size of ICT in Turkey increased with an average pace of 23 percent (TUBISAD, 2013 & 2020). The information technologies (computer equipment, software, and other services) market size increased with a faster average pace of 29 percent while the communication technologies' market size increased with an average pace of 19 percent between 2017-2021.

According to TURKSTAT data, the usage of both Enterprise Resource Planning (ERP) and Customer Relationship Management (CRM) in all sectors increased in the recent years². While ERP usage in enterprises increased from 17.8 percent share in 2012 to 28 percent in 2021 in all sectors, CRM usage increased from 9.2 percent share in 2012 to 10.6 percent in 2021. Although the usage of Supply Chain Management (SCM) was only asked in 2012 and 2017, a decrease of share in the total and in the manufacturing, sector is observed, which implies that sectors are probably shifting from SCM usage to ERP and CRM. The usage of ERP and CRM increased higher in the manufacturing sector compared with the total. (See Table 1.) TURKSTAT provides the following data starting from 2012

_		All se	ctors		Man	ufacturin	g (Section	C)
_	Period	ERP	CRM	SCM	Period	ERP	CRM	SCM
	2012	17.8	9.2	17.5	2012	21.7	7.3	19.3
	2013	18.7	8.8	-	2013	22.7	7.8	-
	2014	14.3	7.5	-	2014	14.2	5.5	-
	2015	20.1	9.2	-	2015	24.3	7.5	-
	2016	-	-	-	2016	-	-	-
	2017	13.9	18.8	9.0	2017	16.1	18.5	8.2
	2018	-	-	-	2018	-	-	-
	2019	20.5	18.5	-	2019	25.9	19.4	-
	2020	-	-	-	2020	-	-	-
	2021	28.0	10.6	-	2021	30.9	9.8	-

Table 1: The Share of Enterprise Resource Planning (ERP), Customer Relationship Management (CRM), andSupply Chain Management (SCM) Software Usage in All Sectors and Manufacturing Sector from 2012 to 2021

Source: Turkish Statistical Institute (TURKSTAT), Survey on Information and Communication Technology (ICT) Usage in Enterprises, 2022

Note: All values reflect economic activity (NACE Rev.2. Period is the reference period.

2 TURKSTAT's "Survey on Information and Communication Technology (ICT) Usage in Enterprises" do not include agriculture, banking, and finance sectors.

According to TURKSTAT data, the usage of internet in Turkish enterprises increased from 90.9 percent in 2010 to 94.9 percent in 2019 while the rate of firms that use platforms for web sales increased from 12.3 percent in 2010 to 77.1 percent in 2019. The latter data indicate a fast adoption of digitalization in most of the enterprises (which indicates a fast spread of e-commerce) even though Turkey might be classified as a late adopter in terms of digitalization. Due to differentiating survey questions, TURKSTAT provides the proportion of enterprises employing ICT/IT specialists by size group from 2014 to 2022. While the proportion of enterprises employing an IT/ICT specialist was 10.5 percent in 2014, it increases to 13.7 percent in 2019 and to 17.8 in 2022. The increase is much more distinguishable in large size firms (employing over 250 people), the share of enterprises employing an IT/ICT specialist increases from 53.7 percent in 2014 to 72.6 percent in 2022, while it increases from 7.1 percent to 13.8 percent for small size firms and 20.5 percent to 32.3 percent for medium size firms.

4. TURKSTAT Data and Descriptive Statistics

In this study, two datasets from the Turkish Statistical Institute (TURKSTAT) obtained from annual surveys conducted to all enterprises in Turkey are merged and combined. The first dataset is the Annual Industry and Service Statistics which is based on Turkish Revenue Administration and Social Security Institution's administrative data and Annual Industry and Service Statistics Investment Expenditure Survey results. It provides data on the turnover, the number of persons employed, the number of employees, value-added at factor cost, production value, personnel cost, total purchases of goods and services, change in stocks of goods and services. The sectors of the enterprises are classified by NACE Rev.2.

The second dataset is extracted from the Information and Communication Technologies Usage in Enterprises Survey which is in line with European Statistical Office (Eurostat) methodology. This dataset covers the enterprises and businesses from the manufacturing, construction, retail and wholesale trade and services sectors. TURKSTAT states that they use Stratified Random Sampling by taking into account the economic activities (in accordance with NACE Rev.2) and enterprise size according to the number of employees. The size-classes used are small enterprises (10–49 persons employed), medium-sized enterprises (50–249 persons employed) and large enterprises (250 or more persons employed). All censuses for enterprises with 250+ persons employed are included meanwhile they used sampling for 10-49 and 50-249 size groups. TURKSTAT states they applied weighting method to obtain parameters from the dataset resulting from sampling so as to represent the universe. ³ These parameters include design weights ⁴, adjustments for nonresponse, external distribution checks and ultimate multiplying factor. For instance, there are 6,054 observations in the 2009 dataset and 12,336 observations in the 2019 dataset.

³ Details of TURKSTAT's weighting method are provided in the "Accounting Conventions" of ICT Usage Bulletin: https://data.tuik.gov.tr/Bulten/Index?p=Girisimlerde-Bilisim-Teknolojileri-Kullanim-Arastirmasi-2022-45585

⁴ Since weighted data is used in the regression, heteroskedasticity test cannot be run, therefore the t statistics results are checked.

Both surveys include same enterprises' dataset; however, since ICT Usage in Enterprises Survey do not include the enterprises' value-added, turnover, total number of employee information, both survey datasets are combined by using a key code provided by TURKSTAT. Following parameters from the ICT Usage in Enterprises Survey are used in this study's analysis: Software usage, using webpage to sell online, and the share of internet using personnel or employees.

Three software are included in the ICT Usage in Enterprises Survey. Enterprise Resource Planning (ERP) helps an enterprise for purchases, sales, marketing, finance, management, human resources and organizes these activities under an integrated system and reports. ERP software supports managers to provide information much more quickly for their decision-making. Thus, it promotes the productivity and the profitability of enterprises through which increases their competitivity in their sector. ERP is used by enterprises since the 1990s. Customer Relationship Management (CRM) is saving, evaluating, reporting, and analyzing the data deduced from all the interaction of the business and its customers. There is less costing CRM software in recent years thus it is accessible for small-sized enterprises as well. Supply Chain Management (SCM) software is the software tools or modules used in executing supply chain transactions, managing supplier relationships, and controlling associated business processes in all sectors. Supply chain management maximizes the efficiency of business activities that include planning and management of the entire supply chain which helps businesses in product development, sourcing, production, and logistics by automating operations. Therefore, it increases the physical flow of business as well as informative flow.

Using a webpage to sell the products or services online is an indicator that is used to observe the impacts of digital infrastructure on the firms' profitability and efficiency. Similarly, the share of the personnel using internet is used to observe the impacts of internet/digitalization on the firms' productivity.

Other essential indicators such as internet speed which allows businesses to facilitate their business processes are also provided in the ICT Usage in Enterprises Survey after 2012. Grimes, Ren and Stevens (2012) found out from their analysis on 6,000 firms in New Zealand (from a survey conducted in 2006) that broadband adoption boosts firm productivity by 7-10%; effects are consistent across urban versus rural locations and across high versus low knowledge-intensive sectors. Although Bertschek, Cerquera and Klein (2013) found out from their analysis on German firm data (between 2001-2003) that broadband Internet has no impact on firms' labor productivity, whereas it exhibits a positive and significant impact on their innovation activity. The employment of information technologies (IT) personnel and providing education to IT personnel are also the indicators that are provided in the ICT Survey after 2012. In this study's cross-sectional analysis, the latter indicators were not available since they are provided in the surveys after 2012. An extended model for the 2019 dataset, adding these indicators, were presented in the fourth section. To understand if the adoption of digitalization had impacts on firm-level productivity throughout the ten years period, this study examines 2009 and 2019 data and estimates the impacts of ICT usage and ICT using labor on labor productivity.

Year	ICT	All firms N	All firms (%)	Manufacturing firm N	Manufacturing firms (%)	High- tech (%)	Mid- high tech (%)	Mid-low tech (%)	Low tech (%)
	ERP	6.054	23%	2.267	31%	58%	44%	31%	25%
	CRM	6.054	14%	2.267	11%	18%	16%	9%	10%
2009	Web Page	6.054	67%	2.267	75%	88%	89%	80%	66%
	Web Order	6.054	12%	2.267	10%	10%	13%	10%	8%
	ERP	12.644	44%	4.386	56%	61%	68%	57%	49%
	CRM	12.644	31%	4.386	32%	40%	37%	33%	29%
	Web Page	12.644	73%	4.386	81%	92%	89%	85%	74%
2019	Web Order	12.644	13%	4.386	11%	11%	10%	8%	12%
	IT Specialist	12.644	39%	4.386	46%	51%	55%	46%	41%
_	Internet Speed	12.644	56%	4.386	54%	62%	60%	54%	51%

Table 2: The Percentage of ICT Using Firms (2009 – 2019)

Source: Author's calculations based on TURKSTAT Annual Industry and Services Survey and ICT Usage in Enterprises Survey 2009 and 2019 datasets.

TURKSTAT's ICT Usage in Enterprises 2009 dataset covers 6,054 observations, and 2019 dataset covers 12,644 observations. From these datasets, it can be observed that the share of ERP and CRM using enterprises in total increases in 10 years from 23 percent to 44 percent and from 14 percent to 31 percent respectively. Meanwhile for the manufacturing sector it is more distinguishable; the share of ERP using firms increases from 31 percent in 2009 to 56 percent in 2019 and the share of CRM using firms increases from 11 percent in 2009 to 32 percent in 2019. (See Table 2).

According to firm size differentiation, it is observed that the logarithm of the value added at factor cost per employees (labor productivity) of the firms using ERP software is higher in all size of firms. In 2019, the gap of labor productivity increases (See Figure 1.a). Similar results are observed for the firms that are using CRM in 2009 and 2019; the labor productivity of the firms using CRM software is higher than the ones that are not using in both 2009 and 2019 (See Figure 1.b). However, in 2019, the gap between the firms using CRM software and the firms that are not using is very low.



a) Usage of ERP by Firm Size



Source: Author's calculations based on TURKSTAT Annual Industry and Services Survey and ICT Usage in Enterprises Survey 2009 and 2019 datasets.

Notes: 0 and 1 represents the firm not using or using ERP software. The firm size is classified as micro sized (0-9 employees) as 1, small sized (10-49 employees) as 2, medium sized (50-249 employees) as 3 and large sized (250 and over employees) as 4.

b) Usage of CRM by Firm Size



Source: Author's calculations based on TURKSTAT Annual Industry and Services Survey and ICT Usage in Enterprises Survey 2009 and 2019 datasets.

Notes: 0 and 1 represents the firm not using or using CRM software. The firm size is classified as micro sized (0-9 employees) as 1, small sized (10-49 employees) as 2, medium sized (50-249 employees) as 3 and large sized (250 and over employees) as 4.

5. An Empirical Analysis on Firm Productivity in Turkey Using Digitalization Data

The cross-sectional regression helps to explain observations collected from many different individuals, or in our case enterprises, at a given time. Since the data from ICT Usage in Enterprises Survey includes observations from many different businesses each year and it also lacks some of the indicators every year, the cross-sectional analysis is the eligible method to observe the data in time.

Following the related literature and former studies, this study investigates if the firms that adopted a digitalization method (the software ERP, the software CRM, the webpage, the share of the personnel which uses the internet while executing their tasks) have higher productivity level. Therefore, two models to examine the relationship between ICT indicators and firm productivity in the manufacturing sectors separately are built, the latter including additional indicators.

5.1 Baseline Model

The model in this study is similar to one used by Taştan (2021), which observes the impacts of ICT usage on firm productivity. Since the dataset of ICT Usage in Enterprises Survey includes binary variables such as "using a software" (where the answer is Yes or No), and numerary variables such as "the number of personnel using internet", a linear regression model is the most applicable for this study. To observe the productivity level of the enterprises, the value added of each firm is divided by the total number of their employees. Then, the logarithm of the labor productivity and share of the internet using employees in total are taken since the range of values of the productivity level and the number of employees between the enterprises are large, and through logarithmic estimates the distribution of values is less skewed. The first model is as follows:

 $logproductivity_{i} = \beta_{0} + \beta_{1}ERP_{i} + \beta_{2}CRM_{i} + \beta_{3}WebPage_{i} + \beta_{4}share internet employee_{i} + u \text{ (Model 1)}$

Where "log productivity" is the natural logarithm of labor productivity (which is measured by dividing the value added of each firm by their number of personnel) of firm "i", "ERP" and "CRM" are the software use as the indicators of ICT usage of firm "i", "Web Page" indicates the firm "i" using a webpage of their own or outsource it to sell their products online, "share internet employee" is the share of the internet using employees in total of firm "i" and u is the error term.

In the following ordinary least squares (OLS) estimation, the indicator "factor" (which is the coefficient for micro, small and medium-sized firms) is used, since TURKSTAT only gathers information from a representative number of firms from SME sized firms in the ICT Usage in Enterprises Survey. The t statistics computed from heteroskedasticity robust standard errors under the OLS for each estimation are checked and when the t statistics are higher than 0.05, the indicator is considered as statistically insignificant. The results from the estimations are grouped by the technology classification from Eurostat. Table 3 below provide the results of the OLS estimation of the baseline model for the years 2009 and 2019.

Technology level	High technology	Medium-high technology	Medium-low technology	Low technology
ERP	1.020***	0.454***	0.600***	0.287**
	(0.307)	(0.109)	(0.146)	(0.113)
CRM	0.340	-0.025	0.049	0.095
	(0.353)	(0.162)	(0.232)	(0.117)
WebPage	0.643***	0.323**	0.246**	0.219***
	(0.238)	(0.150)	(0.104)	(0.060)
shareintemployee	0.011***	0.080***	0.113***	0.111***
	(0.000)	(0.022)	(0.023)	(0.026)
Constant	9.548***	9.540***	9.629***	9.510***
	(0.184)	(0.140)	(0.078)	(0.046)
Observations	49	424	669	1,058
R-squared	0.512	0.153	0.134	0.152

Table 3: OLS Estimation Results for Baseline Model by Technological Breakdown of Manufacturing

 Industry – 2009 Results

Source: Author's calculations based on TURKSTAT Annual Industry and Services Survey and ICT Usage in Enterprises Survey 2009 and 2019 datasets.

Note: Robust standard errors in parentheses

*** p<0.01, ** p<0.05, * p<0.1

Table 3: (Continued): OLS Estimation Results for Baseline Model by Technological Breakdown o	f
Manufacturing Industry – 2019 Results	

Technology level	High technology	Medium- high technology	Medium-low technology	Low technology
ERP	0.608***	0.355***	0.429***	0.388***
	(0.174)	(0.106)	(0.091)	(0.079)
CRM	0.102	-0.045	-0.041	-0.084
	(0.192)	(0.110)	(0.086)	(0.128)
WebPage	0.399**	0.557***	0.327***	0.395***
	(0.175)	(0.084)	(0.099)	(0.079)
shareintemployee	0.607***	0.062***	0.039*	-0.052
	(0.227)	(0.004)	(0.022)	(0.056)
Constant	10.708***	10.738***	10.762***	10.605***
	(0.169)	(0.066)	(0.085)	(0.044)
Observations	302	846	1,056	2,018
R-squared	0.536	0.159	0.110	0.092

Source: Author's calculations based on TURKSTAT Annual Industry and Services Survey and ICT Usage in Enterprises Survey 2009 and 2019 datasets.

Note: Robust standard errors in parentheses

*** p<0.01, ** p<0.05, * p<0.1

According to estimation results for the cumulative technological breakdowns, usage of ERP software is statistically significant and has a positive sign for all technological levels of manufacturing sectors. On the other hand, usage of CRM is statistically insignificant for all technological levels. Owning a web page is also statistically significant and has a positive sign for all technological levels, however, share of internet using employees which is statistically significant becomes insignificant for low technology and medium-low technology manufacturing sectors in 2019. Therefore, more detailed sectoral estimations are conducted.

OLS estimation results of the NACE-2 two-digit sector breakdown of the manufacturing industry are in line with estimation results for the technological breakdown. The results for two sectors – production of tobacco (12) and production of petroleum products (19) were omitted due to lack of observations. Manufacturing industry sectoral breakdown's estimation results are presented in the Appendix (See Tables A.1 and A.2). For some sectors, there are interesting results:

High-technology sectors (21, 26):

According to estimation results, ERP usage and share of internet using employees are statistically significant for both high-technology sectors (pharmaceuticals – 21 and computers – 26) in 2009 and 2019. While using CRM and owning a web page are only statistically significant for computers sector (26) in 2009, they are only statistically significant for pharmaceuticals (21) in 2019.

Medium-high technology sectors (20, 27, 28, 29, 30):

For the medium-high technology sectors, the estimation results indicate that ERP usage is statistically significant for all medium-high technology sectors in 2009 except for manufacture of electrical equipment (27) in which CRM usage is statistically significant. In 2019, the ERP usage becomes statistically insignificant for all of them except it becomes significant for electrical equipment sector (27). The results are not in line with the cumulative "medium-high technology" estimation.

Medium-low technology sectors (19, 22, 23, 24, 25, 33):

While ERP usage is statistically significant for the sectors "rubber and plastic products" (22), in both 2009 and 2019, it is significant for "fabricated metal products, except machinery and equipment" (25) sector in 2019. The webpage indicator is statistically significant for "rubber and plastic products" (22) in 2019. The share of internet using employees is significant in 2009 and insignificant in 2019, which is in line with the cumulative "medium-low technology" estimation.

Low-technology sectors (10, 11, 12, 13, 14, 15, 16, 17, 18, 31, 32):

The estimation results show that the ERP usage is statistically significant and has a positive sign for the sectors food products (10), wood and of products of wood and cork, except furniture (16), and paper and paper products (17) in 2009; and ERP usage becomes statistically significant and has a positive sign for sectors textiles (13), wearing apparels (14), in 2019.

First model's estimation results indicate that both in 2009 and 2019, nine manufacturing sectors (out of 20 sectors that were eligible to conduct an estimation) that using the software ERP have a positive returns to their labor productivity. Having a webpage also have a positive returns to the labor productivity of nine sectors in 2019.

5.2. Extended Model

Extended model includes more indicators representing firm digitalization. This model was only able to be executed for the year 2019 since 2009 data do not cover the following additional indicators. The second model is as follows:

$$\begin{split} logproductivity &= \beta_0 + \beta_1 ERP_i + \beta_2 CRM_i + \beta_3 webpage_i + \\ \beta_4 share internet employees_i + \beta_5 internet speed_i + \beta_6 IT specialist_i + u \\ (Model 2) \end{split}$$

This model includes two more indicators to the first one where "internet speed" is the indicator which shows if the internet speed of firm "i" is higher than 30 megabits per second, "IT specialist" indicates if the firm "i" hires an information technology specialist or more than one employee (the exact number of employee is not provided in the Survey), and "u" is the error term. IT employee's educational training was also used in the estimation however excluded due to the correlation with the "IT specialist" indicator. The results are included in the Appendix (See Table A.3).

OLS estimation results for all manufacturing industry sectors in 2019 indicate that employing an IT/ICT specialist is statistically significant for ten out of twenty the manufacturing industry sectors (that were eligible to conduct an estimation), while the internet speed is only statistically significant for six sectors. For both high-technology manufacturing sectors employing an IT specialist and using a high speed internet connection are statistically significant and have positive signs, and surprisingly, only for some medium-low and low-technology sectors employing an IT/ ICT specialist is statistically significant.

Although this study is the first which is observing the breakdown of the manufacturing industry based on NACE Rev. 2 sectors in two-digits, the results are in line with previous studies on the impacts of ICT to productivity. The results do not imply a direct relationship between ICT and firm-level productivity in whole manufacturing sector, but it is observed that using a software

(especially ERP) in this case has a broad-based impact on the manufacturing sector, meanwhile the share of internet using personnel and the employment of an IT personnel has an impact on the labor productivity of some sectors.

The results of OLS estimation from baseline model indicate that the labor productivity level of high technology sectors pharmaceuticals (21) and computers (26), and medium-high technology level sector electrical equipment (27) are consistently and positively impacted by the usage of ERP software. Since these mentioned sectors were stated to have high labor productivity levels and to invest more in research and development (Doğruel and Doğruel, 2008), the estimation results are coherent with former studies. The estimation results also indicate that for sectors medium-low technology level sectors rubber and plastic products (22), and fabricated metal products, except machinery and equipment (25) and low-technology level sectors food products (10), textiles (13); and paper and paper products (17), the usage of ERP has positive impacts on the labor productivity.

Although the estimation results do not imply a direct relationship between the labor productivity and ICT indicators such as ICT usage and share of internet using employees in total, it is observed by the technological breakdown and sectoral breakdown that ERP using firms and the share internet using employees in total have positive impacts on the labor productivity of the firms with higher technology. Therefore, the estimation results of this study indicate that firms that are investing on ICT software (ERP) and ICT infrastructure (internet speed) would have higher level of labor productivity, in line with former studies on Turkey. The relationship might also be the reverse, the firms with higher level of labor productivity would invest to ICT software and ICT infrastructure to increase their output and profit. One of the issues about calculations is that the ICT capital is not observed from the datasets, and this prevents to measure its impacts on productivity.

6. Conclusion

In this study, the impacts of ICT on firm-level productivity in Turkish manufacturing industry based on NACE Rev. 2 two-digit level are analyzed with an aim to observe if ICT has differing impacts on different technological intensity levels. Our approach is similar to previous studies, especially Taştan's (2021) recent study on the firm-level evidence from Turkey, and Maliranta and Rouvinen's (2004) article about the use of ICT in Finnish business enterprises.

Unfortunately, the lack of data and observations prevents doing a more comprehensive research using other ICT indicators. An extended model using additional indicators for 2019 dataset was also carried out. The analysis shows that internet speed and IT employment are statistically significant for some of the sectors and have a positive impact on productivity. This should be assessed carefully because the lack of data (about the investments of the enterprise, the age of the enterprise, the organizational structure, etc.) prevents to do a more detailed research.

Based on the aforementioned estimation results, some policies for the enterprises and policymakers might be suggested. Given that the software usage, especially the Enterprise Resource Planning has a positive impact on labor productivity on more than half of the manufacturing industry in Turkey, training to improve the personnel that are using the software would be strongly recommended. Since the adoption of ICT-usage has a positive impact on firm productivity, central government and policymakers might implement policies that will encourage the firms to adopt more digitalization; these policies will include technology-based incentives, loans and increasing the technological infrastructure of the organized industrial zones.

References

- Aboal, D., & Tacsir, E. (2018). Innovation and productivity in services and manufacturing: the role of ICT. *Industrial and Corporate Change*, 27(2), 221-241.
- Álvarez, R. (2016). The impact of R&D and ICT investment on innovation and productivity in Chilean firms. *Inter-American Development Bank Technical Note Series*: Washington DC, IDB-TN-1056, June.
- Atasoy, H., Banker, R. D., & Pavlou, P. A. (2016). On the longitudinal effects of IT use on firm-level employment. *Information Systems Research*, 27(1), 6-26.
- Barker, G. R., Fuss, M., & Waverman, L. (2008). The Contribution of ICT to Productivity in 18 OECD Countries: Focus on Australia. ANU Centre for Law and Economics Working Paper, (3).
- Basant, R., Commander, S. J., Harrison, R., & Menezes-Filho, N. (2006). ICT adoption and productivity in developing countries: new firm level evidence from Brazil and India.
- Bertschek, I., Cerquera, D., & Klein, G. J. (2013). More bits-more bucks? Measuring the impact of broadband internet on firm performance. *Information Economics and Policy*, 25(3), 190-203.
- Boston Consulting Group, (2015) "Industry 4.0 The Future of Productivity and Growth in Manufacturing Industries"
- Brynjolfsson, E. (1993). The productivity paradox of information technology. *Communications of the ACM*, 36(12), 66-77.
- Brynjolfsson, E., & Hitt, L. M. (2003). Computing productivity: Firm-level evidence. *Review of economics and statistics*, 85(4), 793-808.
- Commander, S., Harrison, R., & Menezes-Filho, N. (2011). ICT and productivity in developing countries: New firm-level evidence from Brazil and India. *Review of Economics and Statistics*, 93(2), 528-541.
- Crépon, B., Duguet, E., & Mairessec, J. (1998). Research, innovation and productivi [ty: an econometric analysis at the firm level. *Economics of Innovation and new Technology*, 7(2), 115-158.
- Crespi, G., & Zuniga, P. (2012). Innovation and productivity: evidence from six Latin American countries. *World development*, 40(2), 273-290.
- Doğruel, A. S., & Doğruel, F. (2008), Türkiye Sanayiine Sektörel Bakış, İstanbul, TUSİAD, Yayın No: TÜSİAD-T/2008–05/466.
- Gal, P., Nicoletti, G., Renault, T., Sorbe, S., & Timiliotis, C. (2019). Digitalisation and productivity: In search of the holy grail–Firm-level empirical evidence from EU countries. *OECD Economics Department Working Papers*
- Grimes, A., Ren, C., & Stevens, P. (2012). The need for speed: impacts of internet connectivity on firm productivity. *Journal of Productivity Analysis*, 37(2), 187-201.

- Harrigan, J., Reshef, A., & Toubal, F. (2018). Techies, trade, and skill-biased productivity, *National Bureau of Economic Research*, (No. w25295).
- Hawash, R., & Lang, G. (2020). Does the digital gap matter? Estimating the impact of ICT on productivity in developing countries. *Eurasian Economic Review*, 10(2), 189-209.
- Jorgenson, D. W., Ho, M. S., & Stiroh, K. J. (2008). A retrospective look at the US productivity growth resurgence. *Journal of Economic perspectives*, 22(1), 3-24.
- Kılıçaslan, Y., Sickles, R. C., Atay Kayış, A., & Üçdoğruk Gürel, Y. (2017). Impact of ICT on the productivity of the firm: evidence from Turkish manufacturing. *Journal of Productivity Analysis*, 47(3), 277-289.
- Maliranta, M., & Rouvinen, P. (2004). ICT and business productivity: Finnish micro-level evidence. *The Economic Impact of ICT; Measurement, Evidence and Implications*, 12(6), 213-240.
- Peraković, D., Periša, M., & Zorić, P. (2019). Challenges and Issues of ICT in Industry 4.0. *Design, simulation, manufacturing: The innovation exchange*, 259-269.
- Polder, M., Leeuwen, G. V., Mohnen, P., & Raymond, W. (2009). Productivity effects of innovation modes.
- Price Waterhouse Coopers (PWC), (2016). Industry 4.0: Building the digital enterprise https://www.pwc. com/gx/en/industries/industries-4.0/landing-page/industry-4.0-building-your-digital-enterpriseapril-2016.pdf (Accessed on 10 October 10 2022)
- Stiroh, K. J. (2002). Information technology and the US productivity revival: what do the industry data say?. *American Economic Review*, 92(5), 1559-1576.
- Solow, R. (1987). You can see the Computer Age Everywhere but in the Productivity Statistics. *New York Review of Books.*
- Tambe, P., & Hitt, L. M. (2012). The productivity of information technology investments: New evidence from IT labor data. *Information systems research*, 23(3-part-1), 599-617.
- Taştan, H. (2021). Firmaların Bilişim Teknolojileri Kullanımı ve Verimlilik İlişkisi, *Türkiye Ekonomisinde* Büyüme, Kalkınma ve Eşitsizlik, Efil Yayınevi, 223-251
- Taştan, H., & Gönel, F. (2020). ICT labor, software usage, and productivity: Firm-level evidence from Turkey. *Journal of Productivity Analysis*, 53(2), 265-285.
- TÜBİSAD (2013), Türkiye Bilişim Sanayicileri Derneği, Bilgi ve İletişim Teknolojileri 2012 Pazar Verileri Raporu https://www.tubisad.org.tr/tr/images/pdf/tubisad-bilgi-merkezi-sunumu-23-mayis-2013. pdf (Accessed on 1 November 2022)
- TÜBİSAD (2020), Türkiye Bilişim Sanayicileri Derneği, Bilgi ve İletişim Teknolojileri 2019 Pazar Verileri Raporu https://www.tubisad.org.tr/tr/images/pdf/tubisad-bit-2019.pdf (Accessed on 1 November 2022)
- TÜBİSAD (2020), Turkey's Digital Transformation Index 2020 https://www.tubisad.org.tr/tr/images/pdf/ tubisad_dde_endeks_report_eng.pdf (Accessed on 29 December 2022)
- TURKSTAT (2022) Annual Industry and Service Statistics Survey. http://www.tuik.gov.tr
- TURKSTAT (2022) Information and Communication Technologies Usage in Enterprises Survey. http://www.tuik.gov.tr

Appendix

Technology level		-				Low techn	ology				
NACE Code	10	11	12	13	14	15	16	17	18	31	32
						Lathar	Wood and of products	Dapar	Drinting and		
						and	and cork.	and	reproduction		
	Food		Tobacco		Wearing	related	except	paper	of recorded		Other
Sector	products	Beverages	products	Textiles	apparel	products	furniture	products	media	Furniture	manufacturing
ERP	0.939**	0.124		0.242*	0.077	-0.083	3.036***	1.955***	0.175	0.377	-0.014
	(0.399)	(0.799)		(0.143)	(0.156)	(0.158)	(0.155)	(0.547)	(0.220)	(0.244)	(0.329)
CRM	-0.335	2.690**		0.520**	0.047	0.014	-0.585***	-0.844	-0.153	-0.139	0.344
	(0.216)	(0.797)		(0.203)	(0.131)	(0.136)	(0.023)	(0.522)	(0.209)	(0.307)	(0.300)
WebPage	0.408***			0.364**	-0.025	0.148	0.257	0.446	0.359	0.029	0.725*
	(0.124)			(0.143)	(0.098)	(0.197)	(0.272)	(0.377)	(0.234)	(0.309)	(0.404)
shareintemployee	0.096***	0.781***		0.141***	0.039	0.106	-0.463	-0.680***	0.220	0.265***	-0.477*
	(0.030)	(0.082)		(0.013)	(0.035)	(0.179)	(0.988)	(0.233)	(0.355)	(0.072)	(0.241)
Constant	9.584***	10.080***		9.495***	9.554***	9.531***	9.325***	9.648***	9.448***	9.326***	9.266***
	(0.093)	(0.422)		(0.112)	(0.069)	(0.162)	(0.289)	(0.311)	(0.249)	(0.300)	(0.377)
Observations	204	7		239	256	42	37	40	101	94	36
R-squared	0.276	0.951		0.323	0.022	0.029	0.055	0.368	0.062	0.244	0.231

Table A.I: OLS Estimation Results for Baseline Model by Sectoral Breakdown of Manufacturing Industry - 2009

Robust standard errors in parentheses

*** p<0.01, ** p<0.05, * p<0.1

Technology level			Medium-lo	w technolog	3y			Mediu	ım-high techno	ology		High-technology	
NACE Code	19	22	23	24	25	33	20	27	28	29	30	21	26
Sector	Coke and refined petroleum products	Rubber and plastic products	Other non- metallic mineral products	Basic metals	Fabricated metal products, except machinery and equipment	Repair and installation of machinery and equipment	Chemicals and chemical products	Electrical equipment	Machinery and equipment n.e.c.	Motor vehicles, trailers and semi- trailers	Other transport equipment	Basic pharmaceutical products and pharmaceutical preparations	Computer, electronic and optical products
ERP		0.694***	0.612*	0.573	0.426*	0.594	1.271***	-0.220	0.311**	0.823***	1.593***	1.298***	0.761**
		(0.182)	(0.321)	(0.429)	(0.222)	(0.725)	(0.329)	(0.245)	(0.155)	(0.236)	(0.315)	(0.321)	(0.272)
CRM		0.310	0.039	0.830*	-0.034	-1.419*	-0.439	0.604**	-0.161	0.167	-0.289	-0.171	0.761**
		(0.335)	(0.460)	(0.465)	(0.259)	(0.836)	(0.320)	(0.237)	(0.198)	(0.390)	(0.340)	(0.316)	(0.276)
WebPage		0.215	-0.005	0.874**	0.409**	0.007	-0.435	1.008***	0.333*	0.592	-0.833***	0.525	0.770***
		(0.201)	(0.230)	(0.338)	(0.181)	(0.312)	(0.400)	(0.177)	(0.200)	(0.448)	(0.267)	(0.329)	(0.216)
shareintemployee		0.084***	0.080***	0.110**	0.394***	0.508***	0.092***	0.321***	0.066***	-0.173	0.248	0.427***	0.012***
		(0.020)	(0.022)	(0.052)	(0.098)	(0.041)	(0.011)	(0.038)	(0.022)	(0.639)	(0.338)	(0.087)	(0.000)
Constant		9.584***	9.754***	9.265***	9.393***	9.731***	10.353***	9.101***	9.542***	9.065***	10.186***	9.705***	9.294***
		(0.178)	(0.139)	(0.243)	(0.152)	(0.201)	(0.323)	(0.094)	(0.178)	(0.453)	(0.191)	(0.269)	(0.034)
Observations		170	174	99	189	36	74	78	151	101	20	23	26
R-squared		0.257	0.116	0.277	0.153	0.456		0.402	0.136	0.278	0.615	0.784	0.685

Table A.I: (Continued): OLS Estimation Results for Baseline Model by Sectoral Breakdown of Manufacturing Industry – 2009

Robust standard errors in parentheses

*** p<0.01, ** p<0.05, * p<0.1

A Cross Sectional Analysis on the Impacts of Ict on Turkish Manufacturing Industry

Technology level						Low tee	chnology				
NACE Code	10	11	12	13	14	15	16	17	18	31	32
Sector	Food products	Beverages	Tobacco products	Textiles	Wearing apparel	Leather and related products	Wood and of products of wood and cork, except furniture	Paper and paper products	Printing and reproduction of recorded media	Furniture	Other manufacturing
ERP	0.615***	0.025		0.311***	0.416**	0.155	0.240	0.848***	0.258	0.070	0.018
	(0.226)	(0.710)		(0.092)	(0.200)	(0.222)	(0.199)	(0.187)	(0.182)	(0.097)	(0.331)
CRM	-0.030	-1.056**		0.107	-0.738	0.486*	0.158	-0.366*	0.061	-0.059	0.248
	(0.159)	(0.385)		(0.112)	(0.600)	(0.259)	(0.137)	(0.202)	(0.197)	(0.096)	(0.310)
WebPage	0.397	0.837**		0.199*	0.582***	-0.088	0.427**	0.319	0.257	0.232**	0.551**
	(0.249)	(0.389)		(0.105)	(0.155)	(0.191)	(0.187)	(0.294)	(0.159)	(0.105)	(0.219)
shareintemployee	-0.111	0.281		-0.004	0.014	0.799*	0.467*	-0.282***	-0.049	-0.058	0.324
	(0.102)	(0.577)		(0.148)	(0.196)	(0.475)	(0.253)	(0.097)	(0.229)	(0.080)	(0.413)
Constant	10.511***	11.282***		11.006***	10.414***	10.740***	10.365***	10.907***	10.703***	10.682***	10.398***
	(0.068)	(0.395)		(0.093)	(0.096)	(0.088)	(0.147)	(0.274)	(0.140)	(0.091)	(0.155)
Observations	452	21		479	457	49	104	132	71	182	67
R-squared	0.078	0.639		0.112	0.146	0.257	0.146	0.261	0.081	0.058	0.286

Table A.2: OLS Estimation Results for Baseline Model by Sectoral Breakdown of Manufacturing Industry - 2019

Robust standard errors in parentheses

*** p<0.01, ** p<0.05, * p<0.1

Technology level			Medium	-low technolo	gy			Medi		High-technology			
NACE Code	19	22	23	24	25	33	20	27	28	29	30	21	26
Sector	Coke and refined petroleum products	Rubber and plastic products	Other non- metallic mineral products	Basic metals	Fabricated metal products, except machinery and equipment	Repair and installation of machinery and equipment	Chemicals and chemical products	Electrical equipment	Machinery and equipment n.e.c.	Motor vehicles, trailers and semi- trailers	Other transport equipment	Basic pharmaceutical products and pharmaceutical preparations	Computer, electronic and optical products
ERP		0.403***	0.503*	0.374	0.344***	0.740*	0.651	0.621***	0.194	0.257	0.451	0.669***	0.451***
		(0.155)	(0.280)	(0.276)	(0.103)	(0.404)	(0.477)	(0.179)	(0.144)	(0.178)	(0.455)	(0.190)	(0.132)
CRM		-0.253	0.092	0.079	-0.022	-0.310	0.269	0.050	-0.231	0.009	0.262	0.401**	0.104
		(0.195)	(0.215)	(0.333)	(0.104)	(0.386)	(0.496)	(0.147)	(0.157)	(0.171)	(0.389)	(0.169)	(0.188)
WebPage		0.475***	0.426	0.233	0.231*	-0.161	0.866*	0.346*	0.637***	0.317***	0.424	0.556***	0.284
		(0.164)	(0.275)	(0.191)	(0.132)	(0.318)	(0.485)	(0.202)	(0.110)	(0.112)	(0.334)	(0.137)	(0.290)
shareintemployee		0.191	0.026***	0.344**	0.074	0.695*	-0.496	-0.165	0.117*	0.062***	0.787	1.109***	0.606**
		(0.544)	(0.007)	(0.136)	(0.262)	(0.387)	(0.949)	(0.256)	(0.068)	(0.003)	(0.578)	(0.126)	(0.260)
Constant		10.675***	10.543***	10.810***	10.836***	10.964***	10.783***	10.669***	10.733***	10.902***	10.461***	10.435***	10.785***
		(0.194)	(0.223)	(0.051)	(0.119)	(0.230)	(0.358)	(0.170)	(0.082)	(0.107)	(0.230)	(0.123)	(0.288)
Observations		219	238	154	356	85	89	150	255	293	59	43	259
R-squared		0.188	0.142	0.108	0.080	0.143	0.221	0.303	0.176	0.096	0.280	0.922	0.480

Table A.2 (Continued): OLS Estimation Results for Baseline Model by Sectoral Breakdown of Manufacturing Industry - 2019

Robust standard errors in parentheses

*** p<0.01, ** p<0.05, * p<0.1

A Cross Sectional Analysis on the Impacts of Ict on Turkish Manufacturing Industry

Technology level						Low techno	ology				
NACE Code	10	11	12	13	14	15	16	17	18	31	32
Sector	Food products	Beverages	Tobacco products	Textiles	Wearing apparel	Leather and related products	Wood and of products of wood and cork, except furniture	Paper and paper products	Printing and reproduction of recorded media	Furniture	Other manufacturing
ERP	0.433**			0.319***	0.300	0.198	0.184	0.792***	0.248	0.058	-0.099
	(0.186)			(0.095)	(0.204)	(0.149)	(0.200)	(0.204)	(0.238)	(0.099)	(0.298)
CRM	-0.014	-0.952**		0.073	-0.802	0.464	0.107	-0.386*	0.050	-0.142	0.064
	(0.152)	(0.436)		(0.105)	(0.625)	(0.277)	(0.147)	(0.207)	(0.200)	(0.098)	(0.251)
WebPage	0.349	0.853**		0.214**	0.522***	-0.144	0.424**	0.308	0.326*	0.204*	0.664***
	(0.264)	(0.381)		(0.102)	(0.143)	(0.126)	(0.190)	(0.290)	(0.189)	(0.104)	(0.210)
IT Specialist	0.576***	-0.017		-0.056	0.278	0.538**	0.867***	0.176	-0.098	0.196	0.341
	(0.187)	(0.743)		(0.117)	(0.170)	(0.239)	(0.179)	(0.196)	(0.173)	(0.138)	(0.264)
internet speed	0.092	-0.141		0.219**	0.256	0.197*	-0.124	-0.046	0.203	0.166**	0.420**
	(0.164)	(0.332)		(0.111)	(0.171)	(0.116)	(0.178)	(0.154)	(0.217)	(0.083)	(0.171)
share int employee	-0.117	0.229		0.011	0.026	0.623	0.508*	-0.307***	-0.072	-0.060	0.117
	(0.097)	(0.646)		(0.142)	(0.193)	(0.394)	(0.291)	(0.103)	(0.244)	(0.080)	(0.293)
Constant	10.480***	11.321***		10.924***	10.339***	10.647***	10.391***	10.938***	10.601***	10.641***	10.214***
	(0.090)	(0.408)		(0.088)	(0.129)	(0.101)	(0.151)	(0.289)	(0.157)	(0.097)	(0.184)
Observations	452	21		479	457	49	104	132	71	182	67
R-squared	0.093	0.643		0.140	0.164	0.441	0.188	0.268	0.113	0.090	0.385

Table A.3: OLS Estimation Results for Extended Model by Sectoral Breakdown of Manufacturing Industry - 2019

Robust standard errors in parentheses

*** p<0.01, ** p<0.05, * p<0.1

Technology level			Medium-	low technolo	gy			Mediu	m-high technol	ogy		High-tech	nology
NACE Code	19	22	23	24	25	33	20	27	28	29	30	21	26
Sector	Coke and refined petroleum products	Rubber and plastic products	Other non- metallic mineral products	Basic metals	Fabricated metal products, except machinery and equipment	Repair and installation of machinery and equipment	Chemicals and chemical products	Electrical equipment	Machinery and equipment n.e.c.	Motor vehicles, trailers and semi- trailers	Other transport equipment	Basic pharmaceutical products and pharmaceutical preparations	Computer, electronic and optical products
ERP		0.321**	0.436	0.115	0.239**	0.659	0.989**	0.572***	0.193	0.020	0.053	0.506***	0.312**
		(0.159)	(0.297)	(0.233)	(0.100)	(0.403)	(0.411)	(0.178)	(0.152)	(0.237)	(0.625)	(0.161)	(0.128)
CRM		-0.282	0.100	-0.058	-0.034	-0.224	0.197	0.033	-0.295**	0.030	0.161	0.018	-0.016
		(0.204)	(0.215)	(0.310)	(0.105)	(0.407)	(0.393)	(0.157)	(0.146)	(0.183)	(0.386)	(0.215)	(0.157)
WebPage		0.412**	0.431	0.163	0.226*	0.007	0.471	0.343*	0.617***	0.359***	0.342	0.406***	0.188
		(0.162)	(0.274)	(0.214)	(0.129)	(0.293)	(0.393)	(0.200)	(0.115)	(0.118)	(0.340)	(0.087)	(0.298)
IT Specialist		0.398**	0.129	0.603**	0.328**	0.631	-0.733**	0.225	0.046	0.554***	0.609	0.485***	0.360***
		(0.192)	(0.234)	(0.285)	(0.130)	(0.495)	(0.297)	(0.184)	(0.140)	(0.208)	(0.518)	(0.163)	(0.127)
internet speed		-0.067	0.233	0.136	-0.112	-0.245	0.655*	-0.266*	0.256**	-0.052	0.235	0.323***	0.385**
		(0.142)	(0.194)	(0.249)	(0.110)	(0.273)	(0.366)	(0.141)	(0.113)	(0.134)	(0.342)	(0.078)	(0.168)
share int employee		0.198	0.025***	0.371***	0.072	0.562	-0.548	-0.090	0.142**	0.057***	0.670	0.825***	0.574**
		(0.548)	(0.007)	(0.138)	(0.257)	(0.379)	(0.687)	(0.266)	(0.070)	(0.002)	(0.583)	(0.176)	(0.240)
Constant		10.727***	10.440***	10.782***	10.862***	10.947***	10.843***	10.749***	10.627***	10.862***	10.376***	10.472***	10.614***
		(0.189)	(0.260)	(0.035)	(0.112)	(0.208)	(0.260)	(0.187)	(0.094)	(0.122)	(0.207)	(0.087)	(0.307)
Observations		219	238	154	356	85	89	150	255	293	59	43	259
R-squared		0.222	0.158	0.240	0.107	0.177	0.381	0.353	0.212	0.134	0.327	0.943	0.514

Table A.3 (Continued): OLS Estimation Results for Extended Model by Sectoral Breakdown of Manufacturing Industry - 2019

Robust standard errors in parentheses

*** p<0.01, ** p<0.05, * p<0.1

ARAŞTIRMA MAKALESİ / RESEARCH ARTICLE

THE EVOLUTION OF THE DIGITAL DIVIDE IN TURKEY

Deniz Ece DALGIÇ-TETIKOL^{*} Emin KÖKSAL^{**} Bülent GÜLOĞLU^{***}

Abstract

Disparities in information and communication technology (ICT) access and use usually correlate with demographics and socioeconomic factors. The term "digital divide" refers to demographic, economic, and social inequalities regarding access to and use of ICTs and has critical policy implications. A longitudinal analysis of the digital divide is particularly imperative to understand a country's progress toward digitalization at a high level. Bearing in mind that the full potential of digital advancements can be achieved with the widespread adoption of digital technologies, such analysis is of particular importance for emerging economies like Turkey. In this study, we aim to examine the evolution of digital gaps in Turkey to analyze the dynamics of the digital divide. By this objective, we examine the change in broadband penetration in Turkey and the evolution of digital gaps between different social groups over device access, Internet access, and Internet use between 2008-2020. The results of this study reveal significant digital inequalities between different social groups in Turkey. Although Internet access rates point to progress to some extent, the digital divide in terms of actual Internet use persists between different social groups and regions in Turkey.

Keywords: Digital Divide, Information and Communication Technology, Device Access, Internet Access, Internet Use, Digitalization, Inequality, Turkey

JEL Classification: D63, I3, J1

I. Introduction

A country's digitalization potential and capacity depend on the growth and penetration of ICTs in the region which is coupled with the concomitant rise in the even distribution of possession of the technologies within society. According to the latest data provided by the International Telecommunications Union (ITU, 2020), there are 4.6 billion Internet users worldwide, meaning that 59 percent of the population are Internet users today which was only 16 percent in 2005. The ITU estimates that 4.9 billion people, or 63 percent of the world's population, will be online in 2021. This represents a 17 percent increase over 2019, with 782 million people estimated to have

^{*} Istanbul Technical University, Email: dalgicd15@itu.edu.tr, ORCID: 0000-0003-2267-2986

^{**} Bahcesehir University, Email: emin.koksal@eas.bau.edu.tr, ORCID: 0000-0003-4232-3193

^{***} Istanbul Technical University, Email: guloglub@itu.edu.tr, ORCID: 0000-0002-5889-9616

used the Internet during that time. However, 2.9 billion people remain unconnected. Moreover, very few of those who are counted as Internet users can fully engage with all Internet services. Also, there are strong differences either by region or by different social groups.

Turkey is undergoing digital transformation and morphing into more innovative and technologically advanced organizations and systems. However, given the relationship between social connectivity and digital connectivity (Chen, 2013), the disadvantaged groups of the population would become more vulnerable to remaining socially and digitally excluded, which in turn would cause digital gaps between different social groups and regions. The digital gap, also known as the digital divide, is a new form of social inequality derived from the unequal access to and use of the modern ICT by individuals, households, businesses, and geographic areas at different socioeconomic levels. Scholars proved that ICT access and use are usually associated with socioeconomic and demographic characteristics, such as economic class, gender, race and ethnicity, age, disability, education, rural residency, occupational status, networks, and geographies, and the digital divide were found to be highly associated with these factors. Although technology becomes more integrated into everyday life and digital-intensive activities become an increasingly important component of the economies, foundational access inequalities continue to cause a gap between people who use the ICT and those who do not.

Although internet access in Turkey has been on an upward trend and smartphones are nearly ubiquitous among society, the digital divide is a critical issue in Turkey such that demographic and socioeconomic factors significantly affect Internet access and use in the country (Dalgic-Tetikol et al., 2022). The main objective of this study is to examine the evolution of digital gaps in Turkey by using longitudinal data from TurkStat's "household ICT use and access" survey for the period 2008-2020. Our hypothesis is that there are digital gaps for certain social groups in Turkey. Therefore, in this study, we aim to investigate whether and for which groups these gaps have widened, narrowed, or disappeared over the years. We look at different aspects concerning the digital divide namely, device access, Internet access, and Internet use, and present them by mostly using graphs in order to show the dynamics of the digital gaps and analyze how large the gaps are in each aspect along with different variables. The results also reveal that the digital gaps do not appear to be closing in the near future. This study thus provides the most comprehensive and detailed analysis to date examining the evolution of the digital divide in Turkey. For this reason, we believe that it provides a useful body of knowledge in the design of policies to address the digital divide in Turkey.

The particular emphasis of this article is therefore on how the digital divide in Turkey has changed over time. In accordance with the objective to study the evolution of the digital divide in Turkey, we particularly examine the change in Internet adoption in Turkey, and digital gaps between different social groups over device access, Internet access, and Internet use with respect to years. We investigate how each demographic and socioeconomic variable has progressed between 2008-2020 in terms of ICT adoption.

2. Literature Review

The widespread growth of information and communication technologies (ICT) in recent decades has created incentives for individuals to widen their participation in social, political, and economic areas of life. Notwithstanding such incentives for individuals, using ICT entails having access to technology and infrastructure, as well as learning how to deal with new ICT concepts. ICT access inequality which is referred to as the digital divide, exists among certain social groups and countries. As the Internet reaches critical importance, some social scientists are starting to investigate the demographic and socioeconomic patterns of ICT access and use. Scholars already showed that disparities in ICT access and use usually correlates with demographics and socioeconomic factors such as gender (Antonio and Tuffley, 2014; Gray et al., 2017; Korupp and Szydlik, 2005; Mumporeze and Prieler, 2017), age (Grishchenko, 2020; Peacock and Künemund, 2007; Selwyn et al., 2003), education (Bonfadelli, 2002; Chaudhuri et al., 2005; Talukdar and Gauri, 2011), income (Chaudhuri et al., 2005; Fuchs, 2009; Grishchenko, 2020; Korupp and Szydlik, 2005; Talukdar and Gauri, 2011), employment status (Robles and Torres-Albero, 2012; van Dijk and Hacker, 2003), and region of residence (Lucendo-Monedero et al., 2019; Ruiz-Rodriguez et al., 2018). On the other hand, Internet access opportunities as well as device access opportunities such that having access to various Internet-enabled devices such as computers and mobile devices, play a major role in explaining the diversity of Internet use (Lopez-Sintas et al., 2020; Reisdorf et al. 2022). In a developing country context, device access opportunities are also associated with demographics and socioeconomic factors (Lopez-Sintas et al., 2020).

There are a few studies investigating the digital divide within the Turkish context as well: Acilar (2011) discussed gender, age, education and geographic location aspects of the digital divide in Turkey based on a summary of 2010 data. Polat (2012) emphasized that digital inequalities are interwoven with other social inequalities, but existing policy initiatives fail to address the most disadvantaged groups, indicating the lack of a national strategy for digital exclusion. Köksal and Anil (2015) examined the determinants of broadband access and broadband usage in Turkey in 2012 and found that the digital divide is significantly associated with demographics and region. Dalgic-Tetikol et al. (2022) validated this result by empirically examining the digital divide from device access, Internet access, and Internet use perspectives with recent data (2020), and showed that although the majority of people have an Internet-enabled device (smartphones), age, household income and education are significant predictors of Internet access; in fact, even when access is available, large disparities exist among gender, age, income, education, and different employment groups in terms of Internet use. Also, region is another significant factor affecting both Internet access and use in Turkey. Moreover, Dalgic-Tetikol et al. (2022) and Köksal (2021) underlined the lack of coherent vision on the demand side policies in Turkey to increase Internet penetration in the country - that is, the demographic and socioeconomic factors have been disregarded while developing regulations and related policies.

It is widely acknowledged by scholars that the demographic and socioeconomic patterns of ICT access and use have critical policy implications. Thereof, several studies give particular emphasis

on how differences in ICT access and use in society change over time. The literature on the digital divide is vast across individuals, regions, countries, disciplines, and services (Pérez-Amaral et al., 2021). Grishchenko (2020) underlined that the study of the digital divide requires an integrated approach to analyze its dynamics and changes. Mack et al. (2021) highlighted the importance of a longitudinal analysis of the digital divide. The summary of the literature in this section focuses on the studies which investigate the changes in ICT adoption over time.

Hoffman et al. (2000) is one of the early studies that give particular emphasis on how differences in ICT adoption are changing over time. Polykalas (2014) analyzed the historic evolution of the digital divide across the Member States of the European Union between 2004-2013. The results show that a clear improvement has been achieved in terms of rural broadband coverage, however, the EU policies have not achieved quantitative targets to mitigate the digital divide across the Member States. Ragnedda and Kreitem (2018) shed light on the digital inequalities in the particular setting of East EU by comparing and contrasting the differences and similarities between East EU countries in terms of Internet access and online engagement by analyzing the period 2008-2017. Their analysis shows that Internet penetration is rapidly rising across Europe so the number of people with no connectivity opportunities is significantly diminishing. However, despite the narrowing digital gaps, the first level digital divide has not been completely bridged. With a similar aim that of this study, Nishijima et al. (2017) sought to fill the gap in the literature on the digital divide in Latin American countries and analyze four nationally representative survey data (of years 2005, 2008, 2011, and 2013) on evolution and determinants of the digital divide between 2005 and 2013 in Brazil. The results demonstrate a diminishing trend in the digital gaps; however, digital illiteracy still possesses challenges in ICT access, especially among the elderly. Jin et al. (2018) explored the trends of the digital divide in China between 2004 and 2016, and investigate regional and stratificational digital divides in particular, including the access divide and the usage divide, and found that regional access and usage divides in China have decreased over time. Grishchenko (2020) assessed the digital inequality trends in Russia between 2008-2018, and found that sociodemographic and economic characteristics are associated with uneven distribution of ICT access and use. Specifically, the most disadvantageous groups in Russia in terms of ICT access and Internet use are the low-income, the elderly, individuals with disabilities, and those living in rural areas. The results highlight the fact that despite overall positive trend in access and use of the Internet, those social groups still remain on the unpreferred side of the divide, which exacerbates social inequality. Garín-Muñoz et al. (2019, 2022) and Pérez-Amaral et al. (2020, 2021) examined the evolution of the use of ICT in Spain and analyze the Internet adoption patterns of selected Internet services. Garín-Muñoz et al. (2019) measured the effect of individuals' socioeconomic characteristics on the adoption of Internet services such as e-commerce, e-banking, and e-government by using logistic regression techniques. Pérez-Amaral et al. (2021) measured specific digital gaps which are mainly classified according to demographic and socioeconomic variables by using survey data for the period 2007-2019. The results show that most of the gaps are narrowing. In fact, in the case of gender, the gaps end up much smaller or even reversing in signs in some cases such that in the case of VoIP and social networks women

become more likely users than men. However, some gaps concerning older groups persist. Also, for low-education levels, the digital divide remains high.

Some of the contributions listed above use aggregate data, while others use cross-section, pool, or panel data. Some of them are limited to one dimension of the digital divide such as connectivity or usage or consider only one or a limited set of variables such as gender, income, race, etc. whereas, in this article, we cover a large data set such that a variety of demographic and socioeconomic variables and different aspects of the digital divide in the analysis.

3. The Data

The study draws upon annual data collected in Turkey as part of the Turkish Statistical Institution's (TurkStat) "Information and Communication Technology (ICT) Usage Survey on Households and Individuals", which is prepared and carried out in accordance with EuroStat's survey on "ICT Usage in Households and by Individuals". ¹ The survey aims at collecting information about the information and communication technologies owned by households and individuals and their use every year since 2004 (except 2006). Due to missing and malformed data before 2008, this study covers the years from 2008 to 2020 and includes around 13,000-33,000 individuals of age 16-74 each year. Table 1 shows the number of individual and household participants each year with their gender distribution. The survey's questionnaire alters each year based on the evolving situation of ICT such that new variables emerge, and some variables are either renamed or omitted in some years. Therefore, the raw panel data underwent rigorous analysis, filtering, and harmonization to standardize the information collected throughout the observation period.

	2008	2009	2010	2011	2012	2013	2014	2015	2016	2017	2018	2019	2020
Number of	13314	12524	13236	26355	27394	23428	20150	19623	25058	29359	28888	28675	32955
participants													
Number of	5161	4773	5094	10235	10605	11537	9814	9827	11268	12780	12822	12947	14498
households													
Male	6380	6025	6392	12873	13255	10778	9259	8966	11699	13924	13719	13729	16420
Female	6934	6499	6844	13482	14139	12650	10891	10657	13359	15435	15169	14946	16535

Table 1: Number of Observations throughout the Observation Period 2008-2020

The demographic and socioeconomic variables of interest are as follows: The sampling universe is respondents of ages 16-74, who are any type of Internet user from novice to experienced. Age, income, and level of education are included as categorical variables. We categorized age into four: 16-35 (young/early working age), 36-50 (prime working age), 51-65 (mature working age), and 66-74 (elderly); monthly household income into three: less than minimum wage (low income), more than minimum wage but less than twice the minimum wage (mid-income), and more than

¹ TurkStat Household Information Technologies Usage Statistics Metadata, https://data.tuik.gov.tr/Kategori/ GetKategori?p=Bilim,-Teknoloji-ve-Bilgi-Toplumu-102, accessed on 20.04.2022
twice the minimum wage (high income) 2 . To identify participants' level of education, they are asked about the highest level of education attained. We categorized the education levels into three: below high school, high school, and tertiary degree. ³ We assess individuals who are employed, unemployed, student, homemaker, and retired. ⁴ Finally, we include 12 geographical regions of Turkey to assess if ICT access and use vary among different regions of Turkey and categorized them into three namely, west, central and east according to geographical location. ⁵ ⁶

In the survey, participants were asked if they had Internet-enabled device(s) and an Internet connection and whether they used the Internet in the past 12 months for various types of personal use. Also, they were asked to report how often they use the Internet. The devices considered consist of desktop, laptop, tablet, smartphone, smart TV, and game consoles that enable connectivity. Internet access measure includes access to the Internet from any "Internet-enabled" device via fixed or mobile broadband or both. Internet use indicates whether the individual used the Internet in the last 12 months. We only take individual and household access and use opportunities.

4. Internet Penetration in Turkey

The Internet penetration rate corresponds to the percentage of the total population of a given country or region that uses the Internet. An Internet user is defined as anyone with the capacity to use the Internet, which requires the person to have available access to an Internet connection and the basic knowledge that is necessary to use Internet technology. Turkey has a fixed broadband penetration rate of 20.07 percent and mobile broadband penetration rate of 76.40 in 2020 (Q4) which are below OECD averages of 33.19 percent and 118.40 percent, respectively⁷ Despite the upward trend in broadband adoption in Turkey, the lower rates of fixed and mobile broadband penetration rates indicate lower level of ICT utilization in Turkey compared to many other OECD countries. The evidence therefore leads us to investigate whether social categories play a role in different Internet access and use patterns with lower adoption rates. The TurkStat survey data demonstrate the upward trend in Internet penetration in Turkey between the years 2008-2020.

² The net monthly minimum wage in Turkey in 2020 is 2,324.71 Turkish Lira.

³ The classification is based on International Standard Classification of Education (ISCED) 2011.

⁴ The classification is based on International Classification of Status in Employment (ICSE) 1993.

⁵ The geographical categorization of TurkStat for Turkey's regions is based on the European Nomenclature of Territorial Units for Statistics (NUTS).

⁶ *West:* Istanbul, West Marmara, Aegean, and East Marmara; *Central:* West Anatolia, Mediterranean, Central Anatolia, and West Blacksea; *East:* East Blacksea, Northeast Anatolia, Middle east Anatolia, and South East Anatolia

⁷ OECD, Broadband Portal, http://www.oecd.org/digital/broadband/broadband-statistics/



Figure 1: Share of Survey Participants with Internet Access between the Years 2008-2020

Source: Authors' elaboration based on TurkStat ICT Usage Survey

Figure 1 shows the trend of the share of TurkStat survey participants who reported having Internet access. There is a clear upward trend in the share of people with Internet access – that is, Internet penetration has increased in Turkey over years. As seen in the figure, the share was as low as below 30 percent back in 2008 but as of 2020, the share is over 90 percent. Except for the drop in 2011, the Internet penetration rate in Turkey has steadily increased. Bearing in mind that accessibility does not necessarily transform into usage, we look into the usage trends separately to see whether access and usage perform differently in certain periods.



Figure 2: Internet Access and Internet Use Trends between the Years 2008-2020

Source: Authors' elaboration based on TurkStat ICT Usage Survey

Figure 2 shows changes in the share of survey participants with Internet access and who actually use the Internet between the years 2008-2020 in separate trend curves. When we look at the Internet use curve in particular, we again see an upward trend starting from around 30 percent in 2008 up to almost 80 percent in 2020 with a constant increase. However, from the figure, we can deduce that access does not always transform into usage. Despite the continuous investments in infrastructure deployment and increasing infrastructure availability, there are other factors that limit actual usage.

5. Methodology

To analyze the evolution of the digital divide in Turkey, we measure the digital 'gaps' for each demographic and socioeconomic group. We adopt a similar method that is used in Pérez-Amaral (2021) which we believe enables us to understand whether the gaps are narrowing, widening, or remaining unchanged over time. Therefore, we used the following equation to measure the digital gaps:

Digital gap =
$$(P_H - P_L)/P_H$$

where *P* is the percentage penetration of a given social group (gender, age, education, etc.).

Each group may involve two, three, four, or more categories in the sample. H and L refer to the best and poor-performing categories within the particular social group, respectively. Therefore, P_H refers to the rate of access or use of the most advantageous category in terms of Internet access such that for each social group the most advantageous categories as the following: men, 16-35 age, high education, high household income, student, west, whereas P_L refers to the less advantageous categories compared to the abovementioned categories. The equation thus refers to the difference in penetrations relative to the penetration of the highest category. As also claimed by Pérez-Amaral et al. (2021), the formula admits a straightforward interpretation since the calculations yield a value bounded between zero and one, denoting the percentage difference in the penetration for the category L of a given digital divide indicator (device access, Internet access, Internet use) relative to the penetration in the highest category H. Based on these calculations, in the following sections, we present a set of graphs illustrating the evolution of the digital gaps for each aspect of the digital divide in Turkey.

6. Results

In this section, we present the results of our examination of the historic evolution of the digital divide in Turkey. The results are presented in a set of graphs that illustrate the change in various aspects of the digital divide over time and how each demographic and socioeconomic predictor of the digital divide has evolved. ⁸

⁸ All figures are authors' elaboration based on TurkStat's survey on ICT usage in households for years 2008-2020.

a. The Digital Gap over Device Access

As a first step, we analyze the evolution of access to Internet-enabled devices in Turkey over the years. The TurkStat dataset contains sufficiently diversified information about Internet-enabled devices that allows for a comprehensive analysis of device ownership and device diversity among various social groups in Turkey over the years. Although access to an Internet-enabled device is necessary for connectivity, it is not sufficient by itself to connect to the Internet and maintain connectivity. Nevertheless, it is important to analyze the device ownership patterns in society to examine whether it has been a significant factor affecting connectivity. The analysis on device diversity, in particular, can give an indication of the change in Internet utilization by different social groups as one can diversify their Internet experience by using different types of devices (van Deursen and van Dijk, 2019).





Source: Authors' elaboration based on TurkStat ICT Usage Survey

Figure 3 shows the share of the population using various Internet-enabled devices, throughout the years. This figure demonstrates that smartphones have been the most common type of device used to go online. As of 2020, almost everyone has a mobile phone that enables connectivity. Device availability at home does not necessarily enable connectivity for every household member. Many households do not have enough devices for everyone or not everyone can use all devices to access the Internet. Therefore, although device availability gives an indication of how the country performs in terms of device opportunities itself, it is particularly critical to detect which household members can actually access those devices and which cannot. Therefore, we try to uncover the digital gaps between social groups of different demographic or socioeconomic characteristics on device access.

Figure 4 includes six graphs that illustrate the gap between different social groups in terms of the share of people in each group that have access to an Internet-enabled device. Each curve

represents the social group and demonstrates how each category of the representative social group has progressed over time relative to the most advantageous category in that social group. Therefore, the y-axis represents the percentage gap between the categories with the best-performing category. For example, the first graph on the top left-hand side illustrates the change in the device access gap between different low – and middle-income households (represented by blue and orange lines, respectively), and high-income households.



Figure 4: The Digital Gap over Device Access

As all graphs in Figure 4 show, the digital gap over device access has been nearly closed for all social groups, which appears to be a very promising outcome of the widespread use of Internetenabled devices. However, it is worth mentioning that the measurement of device access here includes all Internet-enabled devices, namely smartphone (or mobile phone that enables Internet connection), desktop, laptop/tablet, smarttv, and game console. From Figure 2, we deduce that today smartphone is a prevalent type of Internet-enabled device used to go online. So, we predict that the narrowed device divide between different social groups appears to be due to the widespread use of smartphones. Therefore, we specifically examine smartphone access gaps in Figure 5



Figure 5: Smartphone Access Gap

In Figure 5, we see that smartphones are available to almost everyone, hence smartphone ownership can no longer be considered an indicator for the digital divide in Turkey in terms of device access. However, although smartphones are getting more powerful and functional each year, with larger screens, faster processors, and more memory and storage coupled with unlimited data plans offered more commonly by service providers and with relatively faster networks than before, they are still limited in convenience to perform some important and relatively more sophisticated tasks such as job applications and interviews, taking online courses, writing papers for school, etc. Furthermore, previous research shows smartphones are minority groups', such as "younger, poorer and less educated users", only mean of Internet access and that their online activity over smartphones remains limited to social activities (Tsetsi and Rains, 2017). Given the evidence from the literature, we, therefore, find it important to analyze the device divide in more detail. For such an analysis, we choose to particularly consider computers (such as desktops, laptops, and tablets) because they offer a different Internet experience than that provided by smartphones, while at the same time, they can be considered a more imperative tool of today's

social and professional life compared to other Internet-enabled devices like smarttvs and game consoles. Figure 5 shows the evolution of computer access gaps between different social groups.



Figure 6: Computer Access Gap

The evolution of the gaps and where the country stands with the computer access divide is apparently not the same as with the smartphone access divide which has been nearly bridged. As Figure 6 clearly illustrates that the gaps over computer access persist, indicating that the majority of the population is indeed deprived of Internet activities that can be performed effectively solely on a reliable computer as they are the primary current prerequisite for performing certain activities. On the other hand, it is worth noting that although device availability has improved, devices are of limited use without a proper connection. We, therefore, extend the analysis in the following sections to assess the evolution of Internet access and Internet use in Turkey over the years.

b. The Digital Gap over Internet Access

Based on TurkStat data, we showed that Internet penetration in Turkey has been on an upward trend (see Section 4) such that today the share of the population who reported having Internet access has exceeded 90 percent. While this points to progress, the digital divide persists among certain social groups.





Figure 7 illustrates the gaps observed in Internet access over the years concerning demographic and socioeconomic differences. The gaps are generally decreasing over time, insomuch that the gaps between some groups are nearly closed. For example, the gap between regions is considerably small towards the end of the period (2020) compared to 10 years ago. For different income and age groups, although the gaps have been substantial among the different categories earlier, today, only the second most advantageous categories, namely middle-income and middle-education categories, could catch up with the most advantageous categories whereas low-income and low-education, are still behind of others. On the other hand, older age groups have made much slower,

even stagnant, progress over the years. Despite the relative improvement in recent years, there is an almost 40 percent gap between the older and younger age groups in terms of Internet access. As for gender, there have often been small gaps between women and men, but they have always been unfavorable for women since 2008. Finally, for different employment statuses, homemakers and retired have usually been at the margins among their group, yet homemakers have made relatively better progress than the retired, who are now the least advantageous in terms of Internet access among other occupations.

c. The Digital Gap over Internet Use

Access to an Internet connection is a necessary but not a sufficient condition for Internet use. It is therefore critical to analyze how the gaps over Internet use have progressed over time, given increased and improved Internet access and Internet use in general as well as narrowing gaps over Internet access.



Figure 8: The Digital Gap over Internet Use

Figure 8 illustrates the evolution of digital gaps over Internet use and shows a more precise picture concerning the differences in Internet adoption in Turkey over the years. Although access divides have been mostly narrowed, high differences in usage remain for all social groups considered. Figure 8 presents a picture drastically different from Figure 7. Despite the overall increase in Internet access (Figure 1) and the decrease in access gaps for most social groups (Figure 7), in Figure 4, we see how terrible the usage gaps between certain groups still are. The increase in the number of individuals with Internet access has not been sufficient to close the digital divide. Despite a decline, especially in recent years, there is still a notable gap between the older and younger age groups. Younger adults are usually more likely than their elders to be earlier adopters of innovation and digital technologies⁹, which explains the large gap of almost 100 percent in the initial year. However, the Internet adoption rate by those in the oldest age group, although it has been increasing for the last five years, is not at the desired pace to catch the younger adults up in a short time. A similar situation is also valid for education and income, particularly for those with the lowest education and income. In terms of regions, the gap between them appears to continue steadily. This might be due to region-specific factors like more rural areas where access to (high)connection is low or cultural dynamics that constrain certain social groups' linkage with the digital realm. In terms of occupational status, the gap between the students and employed and unemployed groups has been significantly narrowed. However, retired and homemakers still use the Internet much less than other employment groups. The gap indeed appears to need time to narrow if the adoption by these groups continues at similar rates. Our analysis regarding the gender gap in usage points to an interesting result. Unlike in the case of Internet access, the gender gap in terms of Internet usage, albeit has dropped compared to earlier persists - that is, female users are 20 percent less than men users despite having access opportunities alike. Moreover, the gender gap has barely improved since 2017, which is an indication of the presence of various barriers preventing a group of women from going online.

7. Conclusion

This study focuses on the digital gaps in Turkey for the period 2008-2020. For that, we investigate how the digital divide in Turkey has changed over time; particularly examine the change in digital gaps between different social groups over device access, Internet access, and Internet use. Unlike many other studies in the literature that focus solely on one variable or one aspect of the digital divide, in this article, we try to convey a detailed analysis by taking different aspects of the digital divide. This comprehensive approach fills the gap in the literature by providing an accurate body of knowledge in the design of policies to address the digital divide in Turkey.

Closing the device divide is the first integral step to closing the digital divide. There is an upward trend in device access, that is, the share of people with device access has increased over time so

⁹ Faverio, M. (n.d.). Share of those 65 and older who are tech users has grown in the past decade. *Pew Research Center*. Retrieved July 29, 2022, from https://www.pewresearch.org/fact-tank/2022/01/13/share-of-those-65-and-olderwho-are-tech-users-has-grown-in-the-past-decade/

the vast majority of the population now has a device that can connect to the Internet. However, smartphone ownership has increased at a faster rate compared to other Internet-enabled devices. It appears that people in Turkey use smartphones as their primary means of online access. Although the increase in smartphone ownership is valuable progress towards the digital divide, reliance on smartphones for connectivity and online engagement cannot perfectly aid in mitigating the problem. On top of that despite the increased computer penetration over years, unlike smartphone access gaps, computer access gaps are still substantial. The socioeconomically disadvantaged groups have very low access rates compared to the advantaged groups. The results indicate that their Internet experience is limited to their capability to utilize what smartphones offer them.

The results reveal that Internet access gaps have been narrowing over time, and in several cases they become small. There has never been a significantly large gap between women and men. In the case of regions, income, and education, their mid categories in particular, although large at the beginning, the gaps have ended up considerably small. There is also a diminishing trend in the gaps between other social groups. Nevertheless, the elderly, retired, and low-income and low-education people are still far behind in terms of Internet access. The results point to the need for demand-side programs intended to stimulate broadband adoption widely by those groups. The policymakers' attention should not remain exclusively on the supply side. Instead, they should pay increasing attention to the demand side policies alongside supply side policies if greatly expanded adoption of broadband is the policy goal (Hauge and Prieger, 2010). Although Internet access points to progress to some extent, the digital divide in terms of actual usage persists between different social groups and regions. The lower levels of online engagement overall along with larger gaps in actual usage indicate lower and uneven digital participation in society. As shown in the Figures, not all groups are homogenous such that their digital engagement has been associated with their demographic and socioeconomic characteristics.

The Internet has become an imperative in the lives of individuals. The results in the present paper point to the need for accelerated policies targeted at inferior social groups. Given the gaps among society in terms of ICT access and use, the gaps between individuals who can access and use the technology and those who cannot, will continue widening unless the necessary actions are taken today. There are a number of ways to improve ICT access and use and provide individuals with the opportunities of a stable and open Internet. This includes strategic objectives such as expanded digital infrastructure which enhances availability and accessibility, reduced telecommunications costs, improved network efficiency, more competitive and diverse broadband markets, strengthened digital literacy through a restructured education system, empowering human capital that can use the technologies effectively and implementing policies at the local level. Each of these measures contributes to narrowing the digital gaps. The right policy should particularly target disadvantaged groups such as women, senior age, low-income, low-educated groups, homemakers, and retired people. In conclusion, reducing the digital divide in Turkey requires a combination of policies which concentrate on the underlying demographic and socioeconomic factors that contribute to the digital divide.

References

- Acilar, A. (2011). Exploring the aspects of digital divide in a developing country. *Issues in Informing Science and Information Technology*, 8, 231-244.
- Antonio, A., & Tuffley, D. (2014). The Gender Digital Divide in Developing Countries. *Future Internet*, 6(4), 673–687. https://doi.org/10.3390/fi6040673
- Bonfadelli, H. (2002). The Internet and Knowledge Gaps: A Theoretical and Empirical Investigation. *European Journal of Communication*, 17(1), 65–84. https://doi.org/10.1177/026.732.3102017001607
- Chaudhuri, A., Flamm, K. S., & Horrigan, J. (2005). An analysis of the determinants of Internet access. *Telecommunications Policy*, 29(9), 731–755. https://doi.org/10.1016/j.telpol.2005.07.001
- Chen, W. (2013). The Implications of Social Capital for the Digital Divides in America. *The Information Society*, 29(1), 13–25. https://doi.org/10.1080/01972.243.2012.739265
- Dalgic-Tetikol, D. E., Guloglu, B., & Köksal, E. (2022). Determinants of internet adoption in Turkey and the need for a more coherent vision on information and communication technologies policy. *Competition* and Regulation in Network Industries, 23(4), 311-336. https://doi.org/10.1177/178.359.17221143060
- Fuchs, C. (2009). The Role of Income Inequality in a Multivariate Cross-National Analysis of the Digital Divide. Social Science Computer Review, 27(1), 41–58. https://doi.org/10.1177/089.443.9308321628
- Garín-Muñoz, T., Pérez-Amaral, T., & Valarezo, Á. (2022). Evolution of the Internet gender gaps in Spain and effects of the Covid-19 pandemic. *Telecommunications Policy*, 102371.
- Garín-Muñoz, T., López, R., Pérez-Amaral, T., Herguera, Í., & Valarezo, A. (2019). Models for individual adoption of eCommerce, eBanking and eGovernment in Spain. *Telecommunications Policy*, 43(1), 100-111. https://doi.org/10.1016/j.telpol.2018.01.002
- Gray, T. J., Gainous, J., & Wagner, K. M. (2017). Gender and the Digital Divide in Latin America*. Social Science Quarterly, 98(1), 326–340. https://doi.org/10.1111/ssqu.12270
- Grishchenko, N. (2020). The gap not only closes: Resistance and reverse shifts in the digital divide in Russia. *Telecommunications Policy*, 44(8), 102004. https://doi.org/10.1016/j.telpol.2020.102004
- Hauge, J. A., & Prieger, J. E. (2010). Demand-Side Programs to Stimulate Adoption of Broadband: What Works? *Review of Network Economics*, 9(3). https://doi.org/10.2202/1446-9022.1234
- Hoffman, D. L., Novak, T. P., & Schlosser, A. (2000). The evolution of the digital divide: How gaps in Internet access may impact electronic commerce. *Journal of computer-mediated communication*, 5(3), JCMC534. https://doi.org/10.1111/j.1083-6101.2000.tb00341.x
- ITU (2020). Measuring Digital Development ICT Price Trends 2020. Geneva
- Jin, J., Meng, F., Chen, A., Shi, L., & Wang, T. (2018). Mapping the evolutive trajectories of China's digital divide: a longitudinal observation. *Ragnedda M. and Mutsvario B., Digital Inclusion. An international Comparative Analysis, Rowman & Littlefield.*
- Korupp, S. E., & Szydlik, M. (2005). Causes and Trends of the Digital Divide. European Sociological Review, 21(4), 409–422. https://doi.org/10.1093/esr/jci030
- Köksal, E. (2021). Regulation of Fiber and the Internet. In M. Eroğlu & M. Finger (Eds.), *The Regulation of Turkish Network Industries* (pp. 383–401). Springer International Publishing. https://doi.org/10.1007/978-3-030-81720-6_19
- Köksal, E., & Anil, B. (2015). The Determinants of Broadband Access and Usage in Turkey: Do Regions Matter? Topics in Middle Eastern and African Economies, 17(1), 113-132. https://doi.org/10. 2139/ ssrn.2362376

- Lopez-Sintas, J., Lamberti, G., & Sukphan, J. (2020). The social structuring of the digital gap in a developing country. The impact of computer and internet access opportunities on internet use in Thailand. *Technology in Society*, 63, 101433. https://doi.org/10.1016/j.techsoc.2020.101433
- Lucendo-Monedero, A. L., Ruiz-Rodríguez, F., & González-Relaño, R. (2019). Measuring the digital divide at regional level. A spatial analysis of the inequalities in digital development of households and individuals in Europe. *Telematics and Informatics*, 41, 197–217. https://doi.org/10.1016/j. tele.2019.05.002
- Mack, E. A., Helderop, E., Ma, K., Grubesic, T. H., Mann, J., Loveridge, S., & Maciejewski, R. (2021). A broadband integrated time series (BITS) for longitudinal analyses of the digital divide. *PloS* one, 16(5), e0250732.
- Mumporeze, N., & Prieler, M. (2017). Gender digital divide in Rwanda: A qualitative analysis of socioeconomic factors. *Telematics and Informatics*, 34(7), 1285–1293. https://doi.org/10.1016/j. tele.2017.05.014
- Nishijima, M., Ivanauskas, T. M., & Sarti, F. M. (2017). Evolution and determinants of digital divide in Brazil (2005–2013). *Telecommunications policy*, *41*(1), 12-24.
- Peacock, S. E., & Künemund, H. (2007). Senior citizens and Internet technology: Reasons and correlates of access versus non-access in a European comparative perspective. *European Journal of Ageing*, 4(4), 191–200. https://doi.org/10.1007/s10433.007.0067-z
- Pérez-Amaral, T., Valarezo, A., López, R., & Garín-Muñoz, T. (2021). Digital divides across consumers of Internet services in Spain using panel data 2007–2019. Narrowing or not?. *Telecommunications Policy*, 45(2), 102093.
- Pérez-Amaral, T., Valarezo, Á., López, R., Garín-Muñoz, T., & Herguera, Í. (2020). E-commerce by individuals in Spain using panel data 2008–2016. *Telecommunications Policy*, 44(4), 101888. https:// doi.org/10.1016/j.telpol.2019.101888
- Polat, R. K. (2012). Digital exclusion in Turkey: A policy perspective. *Government Information Quarterly*, 29(4), 589–596. https://doi.org/10.1016/j.giq.2012.03.002
- Polykalas, S. E. (2014, November). Assessing the evolution of the digital divide across European Union. In Web and Open Access to Learning (ICWOAL), 2014 International Conference on.
- Ragnedda, M., & Kreitem, H. (2018). The three levels of digital divide in East EU countries. World of Media. Journal of Russian Media and Journalism Studies, 1(4), 5-26. DOI: https://doi.org/10.30547/ worldofmedia.4.2018.1
- Reisdorf, B. C., Fernandez, L., Hampton, K. N., Shin, I., & Dutton, W. H. (2022). Mobile Phones Will Not Eliminate Digital and Social Divides: How Variation in Internet Activities Mediates the Relationship Between Type of Internet Access and Local Social Capital in Detroit. Social Science Computer Review, 40(2), 288–308. https://doi.org/10.1177/089.443.9320909446
- Robles, J. M., & Torres-Albero, C. (2012). Digital divide and the information and communication society in Spain. Sociologija i prostor: časopis za istraživanje prostornoga i sociokulturnog razvoja, 50(3(194)), 291-307. https://doi.org/10.5673/sip.50.3.1
- Ruiz-Rodríguez, F., Lucendo-Monedero, A. L., & González-Relaño, R. (2018). Measurement and characterisation of the Digital Divide of Spanish regions at enterprise level. A comparative analysis with the European context. *Telecommunications Policy*, 42(3), 187–211. https://doi.org/10.1016/j. telpol.2017.11.007

- Selwyn, N., Gorard, S., Furlong, J., & Madden, L. (2003). Older adults' use of information and communications technology in everyday life. Ageing & Society, 23(5), 561–582. https://doi. org/10.1017/S0144686X03001302
- Talukdar, D., & Gauri, D. K. (2011). Home Internet Access and Usage in the USA: Trends in the Socio-Economic Digital Divide. Communications of the Association for Information Systems, 28. https:// doi.org/10.17705/1CAIS.02807
- van Deursen, A. J., & van Dijk, J. A. (2019). The first-level digital divide shifts from inequalities in physical access to inequalities in material access. *New Media & Society*, 21(2), 354–375. https://doi. org/10.1177/146.144.4818797082
- van Dijk, J., & Hacker, K. (2003). The Digital Divide as a Complex and Dynamic Phenomenon. *The Information Society*, 19(4), 315–326. https://doi.org/10.1080/019.722.40309487

ARAŞTIRMA MAKALESİ / RESEARCH ARTICLE

WHAT A HARD DECISION: CAPITAL STRUCTURE OF REITS IN ISTANBUL STOCK EXCHANGE

Vijdan UĞURLUAY^{*} ⁽¹⁾ Aslı AYBARS^{**} ⁽¹⁾

Abstract

By examining the factors that determine the capital structure for Turkish Real Estate Investment Trusts (REITs) in Borsa Istanbul (BIST) over the period 2014-2020, we attempt to present a contribution to the capital structure literature. In our study, we use panel data analysis that provide us proofs about the impact of financial performance, stock performance, and corporate structure on capital structure decisions of REITS. Some of the findings are remarkably similar to those of prior studies in this array of literature while our independent variables and capital structure may seem to be connected differently from the leading capital structure theories ("pecking order theory and trade-off theory"). The prominent capital structure theories of pecking order and trade-off theory likewise receive mixed support, although the link between capital structure and our independent variables appears to be skewed. As a result, the theory's assumptions indicate that capital structure changes are driven by survival.

Keywords: REIT, Debt ratio, ISE, Trade-off Theory, Pecking Order Theory, Capital Structure, Turkey **JEL Classification:** C23, G17, G32

I. Introduction

One of the most important areas of the executives' attention is capital structure decisions because these decisions not only affect the firm's market value but also its survival. Capital structure can also be expressed as the financing structure that allows the companies to continue existing investments and/or make new ones.

Specifically, after the study of Modigliani & Miller published in 1958, there is a great increase in the number of empirical studies on this subject. In this prominent study, under the supposition that there is no market failure, it is contended that capital structure choices have no impact on business value. The data of 43 oil companies operating in the USA between 1947 and 1948 are

^{*} PhD Candidate Marmara University, Faculty of Business Administration, E-mail: vijdanugurluay@marun.edu.tr, ORCID ID: 0000-0001-7713-2317

^{**} Marmara University, Faculty of Business Administration, E-mail: asli.aybars@marmara.edu.tr, ORCID ID: 0000-0002-7899-2367

analyzed by using cross-sectional regression analysis and it is found that changes in the capital structure have no effect on the company's profitability or value.

The Trade-off Theory argues that a company's capital structure strikes a balance between the expense of financial hardship from debt and the tax benefits of debt. Up to the optimal borrowing point, businesses gain from the tax benefits of debt, which raises the enterprise value. The risk of financial trouble and bankruptcy rises if the company continues to borrow more than is necessary, diminishing the tax benefits of the loan. (Damodaran, 2001). This explains why businesses cannot borrow continuously.

The Pecking Order Theory states that when businesses need money, they primarily rely on internal resources, which are free of any asymmetric information issues. Then, they use external sources and finally equity financing. Therefore, this theory suggests that profitable firms borrow less and this situation cannot be explained by the Trade-off Theory. As long as the firms' profitability is high, they do not need to use external resources.

The current article aims to determine which factors influence capital structure decisions by analyzing the effects of financial performance, stock performance, and corporate structure on book debt ratio. Determining the attitudes of real estate investment trusts regarding capital structure decisions will enable investors to make more effective decisions while investing in these businesses. In addition, it will shed light on the ability of real estate investment trusts, which have important effects on the economy in which they operate, to make strategic decisions in terms of the continuity of their activities. In the study, the methodology utilized rests upon panel data analysis with random effects.

This paper proceeds as follows: Firstly, literature review of theoretical and empirical studies about capital structure will be revealed. Then, data and methodology are discussed in order to present the details of the sample and the hypothesis of the study is revealed based on the corporate structure background given in previous studies. Finally, the results are presented and discussed.

2. Literature Review

The theoretical background of capital structure has been debated since the seminal paper of Modigliani & Miller (1958). Two main theoretical models, designed for the determinants of capital structure, are Pecking Order Theory and Trade-off Theory.

The Pecking Order Theory suggests that there is a hierarchy in firms' sources of financing. Initially, internal funds are used but when firms need external funding, they prefer to borrow primarily and then issue corporate bonds. As a final preference, equity is issued because investors assume that managers issue stocks as long as the stock is highly priced and prefer to borrow money as long as the stock is low-priced. Therefore, investors will not buy shares before debt

capacity of a firm is exhausted and investors will force firms to follow a financial hierarchy. As a result, when there is asymmetric knowledge, issuing equity has a higher cost than borrowing.

The Trade-off Theory claims that the benefits and drawbacks of borrowing are balanced since the best debt-to-equity ratio can increase a firm's worth. The study of Kraus and Litzenberger (1973) states that financing with debt has tax saving benefits and provides a financial discipline for managers but over-indebtedness can cause bankruptcy and huge agency costs. At that point, a firm should measure the costs and benefits of financing with debt and equity to find the optimal debt ratio. If the leverage ratio of a firm is determined by balance between tax advantages of debt, weighted costs of bankruptcy and agency costs, it is said that the firm is following the static tradeoff model.

The finance literature also provides several empirical studies that explain that the capital structure patterns differ from one country to another or one sector to another. The empirical studies in developed markets cannot present fully supportive results for emerging markets and also the results for different sectors are not parallel. Therefore, there are several empirical studies for different countries and sectors in the literature.

The study, conducted with the data belonging to 123 manufacturing firms in ISE, focuses on the variables that identify the capital structure of firms (Sayilgan et al., 2006). The data between 1993 and 2002 is used to perform panel data analysis in which firm size, profitability, non-debt tax shield, growth rate and fixed asset ratio are employed to analyze capital structure. The analysis reveals a positive correlation between business size and leverage ratio, whereas profitability and borrowing rate show a negative correlation.

Li et al. (2009) investigate whether the capital structure is affected by the factors of governmental ownership, foreign ownership and institutional investor ownership by using dummy variables for data of Chinese firms. The study demonstrates that governmental ownership is positively associated with access to long-term financing sources and leverage while firms that are not publicly owned have significantly less short-term liabilities and total debt than publicly owned firms. Similarly, a study that analyzes the sector balance sheets of Central Bank of the Republic of Turkey between 1996 and 2008, reports a positive relationship between growth opportunities, size, profitability, asset structure and borrowing (Sayilgan & Uysal, 2011).

In the study of Bessler, Drobetz and Kazemieh (2011), it is determined that factors such as market value-book value ratio and profitability have negative effects on corporate leverage. On the other hand, tangibility, size, expected inflation and average industry leverage ratio have positive effects on leverage ratio.

Titman and Wessels (1998) use tangibles, non-debt tax shields, growth, uniqueness of business, industry, size, volatility of income and profitability to explain leverage. As a result of this study, it is presented that long-term debt to equity and short-term debt to equity are negatively related to firm uniqueness. In addition, short-term debt has a bad impact on size and profitability.

However, long term debt and short-term debt are not significantly impacted by either volatility or asset structure.

In the study by Matias and Serrasqueiro (2017), the small and medium-sized companies operating in different regions of Portugal are examined for the period between 2007 and 2011 to identify the factors affecting capital structure. Financial leverage ratio represents the capital structure and the factors included in the study are firm size, profitability, growth and firm age. As a result of the analysis, there is a significantly positive relationship between leverage and not only firm size but also asset structure. It is found that the relationship between company age and leverage is strong and unfavorable. Contrarily, the relationship between growth and financial leverage is not statistically different. It is determined that there is a significant and negative relationship between firm age and leverage. On the other hand, there is no statistically significant relationship difference between growth and financial leverage.

Antoniou et al. (2008) focus on two different type of economies: capital market oriented (England and USA) and bank oriented (France, Germany and Japan). By using panel data analysis, the factors affecting the capital structures of companies are tried to be determined. In this study, financial leverage ratio represents the capital structure. Asset structure, firm size, profitability, growth opportunity and stock performance are used as independent variables within the scope of the analysis. As a result of the analysis, whereas firm size and asset structure are found to have significantly positive effects on leverage ratio, growth opportunity and stock performance are found to have significantly negative effects on leverage ratio.

3. Methodology

This study employs panel data analysis since this methodology combines time series and cross – sectional observations; thus, enabling data variability, enhanced informativeness, and higher degrees of freedom. Accordingly, the model applied is regarded to be superior to the models that only utilize one of those dimensions. Furthermore, panel data controls for heterogeneity, whereas time-series and cross-sectional analysis can come up with biased results in the case of heterogeneity (Baltagi, 2001). Additionally, problem of multi-collinearity is also reduced (Wooldridge, 2002).

3.1. Dataset

In this study, the data set of the analysis rests upon 27 REITs traded in BIST between the years 2014 and 2020. For the 7 years' data period, the firms have to be listed in BIST uninterruptedly; thus, the dataset is strongly balanced. The empirical study is based on data attained from Finnet 2000 Plus: Financial Markets Data Terminal and the raw sample from 2014 to 2020 includes 35 REITs traded in BIST. The REITs, which do not have data as to the financial ratios utilized in the

study and are not traded uninterruptedly in BIST, are excluded from dataset in order to obtain a strongly balanced panel dataset.

3.2. Variable Selections

In the literature, leverage ratios are widely used as variables defining capital structure. As in the literature, the dependent variable is selected to be the leverage ratio defined as the ratio of financial debt, short-term debt and long-term debt, to financial debt plus equity. Moreover, there are several leverage ratios used as a measure of leverage in numerous studies such as Total debt/Total Assets, Short Term Debt/Total Assets and Long-Term Debt/Total Assets (Wald, 1999; MacKay and Phillips, 2005; Mocnik, 2001; Prasad, Dheeria and Woodruff, 2002). This study covers specifically the capital structure of REITs which is one of the most levered sectors and prefers to grow with interest bearing debt instead of equity. Since, the debt-to-capital ratio includes interest-bearing debt while it excludes all other liabilities, unlike debt ratio defined as total debt to total assets", it measures the amount of asset financed with debt (Rajan and Zingales, 1995). This measure is the most appropriate one to the objective of the analysis.

Since previous studies have identified a number of factors that influence capital structure, the following eight variables are selected as independent variables of the study.

The first variable regarded to influence the capital structure of the firm is Market-to-Book value (MTB), a measure of growth opportunities, it is considered to be negatively related to DCR (debt to capital ratio). It is known that firms with high market to book value are more profitable companies and these companies are expected to issue less debt due to their greater growth potential and therefore have lower target leverage ratios (Rajan and Zingales, 1995; Chen and Zhau, 2004).

The size (SIZE) of the firm, which is calculated by taking the natural logarithm of the firm's total assets value, is an important factor in terms of firm's performance, operating capacity, and management structure. It is considered that large-scale companies may have more opportunity to access capital markets easily and to obtain funding sources so SIZE is positively related to DCR (Rajan and Zingales, 1995). It should be mentioned that the capital structure literature has a number of empirical studies that highlight the existence of a favorable link between size and capital structure. (Marsh, 1982; Titman and Wessels, 1988; Rajan and Zingales, 1995; Bevan and Danbolt, 2002; Mocnik, 2001). In these studies, the firm size is represented with either net sales or total assets.

Another variable to affect the capital structure is Tangibility (TAN), which is calculated as the ratio of PPE (Property Plant Equipment) to total assets. PPEs can be used as collateral so it can be effective to use this resource easily when there is a need for funds. The study of Bessler, Drobetz and Kazemieh (2011) indicates that high tangibility ratio triggers high leverage ratio.

Profitability (PRO) is measured by the ratio of net profit to net sales. It is stated that companies with high profitability ratios will no longer need to use high debt in their capital structures according to Pecking Order Theory. A large part of empirical studies also demonstrates the presence of a negative relationship between leverage ratio and profitability due to the fact that companies generally prefer internal sources in funding (Rajan and Zingales, 1995; Bevan and Danbolt, 2002; Huang and Song, 2006).

Some of the previous empirical studies have shown the presence of a negative relationship between market performance (RETURN) and DCR. The study of Antoniou et al. (2008), which rests upon panel data analysis performed with the data of firms in US, UK, France, Germany, and Japan, indicates that debt ratios move in significantly negative direction while stock returns are increasing.

Another variable is selected to be firm age (AGE) and it is considered to be negatively related to DCR because increased experience makes firms more risk averse to use external sources instead of internal sources. The study conducted by Matias ve Serrasqueiro (2017) with the data of small and medium size firms between 2007 and 2011 indicates that there is a significantly negative relationship between firm age and leverage ratio.

Free Float Rate (FFR) is found to be negatively related to DCR. The study of Guner (2016), conducted with 131 publicly traded Turkish firms, implies that the firms with %50-%75 free float rate tend to have lower leverage ratios because as firms become more transparent their investors can easily follow the financing decisions of the firms. This makes firms more conservative about leverage ratios.

Lastly, firms with government share (GOV) can easily reach the funding resources so DCR of these companies is higher than those firms that do not have any government share. Li et al. (2009) identify that firms with government share have higher leverage ratios.

As stated above, panel data analysis with strongly balanced panel data is utilized in the empirical part of the study. The regression analyses aim to compare the explanatory power of 8 independent variables on the dependent variable DCR to test the effect of three main focus groups: financial performance, market performance, and corporate structure. A summary of the variables utilized together with their abbreviations can be seen in Table 1 below.

Abbreviation	Definition	Explanation
Dependent Variables		
DCR	Debt to capital ratio	The financial debt "(short-term and long-term debt)" to equity plus financial debt
Independent Variables		
MTB	Market-to-Book	Market value of equity to book value of equity"

Table 1: Variables

SIZE	Firm size	Natural logarithm of total assets
TAN	Tangibility	PPE (Property Plant Equipment) to total assets
PRO	Profitability	Net income to total assets"
RETURN	Stock Performance	Change in the year-end stock close price
AGE	Firm age	The number of years that has passed since incorporation date
FFR	Free Float Rate	Rate of public shares
GOV	Government Effect	Equal to 1 if the firm has government share, otherwise 0

3.3. The Models Utilized

The model that tests the impact of financial performance on book debt ratio is demonstrated as in the Model (1) below.

Model (1): Financial performance

$$DCR_{it} = \beta_0 + \beta_1 MTB_{it} + \beta_2 SIZE_{it} + \beta_3 TAN_{it} + \beta_4 PRO_{it} + \varepsilon_{it}$$

Additionally, the analysis in Model (2) tests whether the market performance is significant while taking capital structure decisions:

Model (2): Market performance

$$DCR_{it} = \beta_0 + \beta_1 MTB_{it} + \beta_2 SIZE_{it} + \beta_3 TAN_{it} + \beta_4 PRO_{it} + RETURN_{it} + \varepsilon_{it}$$

Finally, we want to measure the contribution of REITs' corporate structure to capital structure as stated in Model (3).

Model (3): Corporate structure

$$DCR_{it} = \beta_0 + \beta_1 MTB_{it} + \beta_2 SIZE_{it} + \beta_3 TAN_{it} + \beta_4 PRO_{it} + \beta_5 RETURN_{it} + \beta_6 AGE_{it} + \beta_7 FFR_{it} + \beta_8 GOV_{it} + \varepsilon_{it}$$

where, subscript i represents the firms (i=1,2, ..., 27); t represents time (t=1,2, ..., 7) and ε_{it} is the error term.

In order to identify the most adequate estimator to our panel data analysis, we follow a series of statistical tests for three models stated in Equation (1), Equation (2), and Equation (3). At first, it has to be detected whether there is unit/time effect or the model is a classical model or not. We use Breusch Pagan LM Test, F-test, and Likelihood Ratio test and select the most adequate model according to the results consistent in all of these three tests. Then, the Hausman Specification Test helps us to determine the type of effect, either fixed or random. Finally, we

check the basic assumptions of panel regression model; multi-collinearity, cross-sectional dependence, autocorrelation, and heteroscedasticity. In order to reach the robust estimators, we have to eliminate these issues if there is any of them.

The issue of multi-collinearity is checked with the help of Variance Inflation Factor (VIF) and cross-sectional dependence is tested by Pesaran's, Friedman's, and Frees' tests. Bhargava, Franzini, Narendranathan Durbin Watson (DW) and Baltagi-Wu LBI tests are utilized to detect autocorrelation. Since the results suggest that all three models have random effects, we use the tests of Levene, Brown, and Forsythe to detect the issue of heteroscedasticity.

As a result of these statistical tests, all three models are determined to be one-way models with random time effect. After determining the adequate model, basic assumptions are checked. The concerns about multicollinearity are checked tested on the level of the variance inflation factors (VIFs), which are all below 10.0 in the regression models discussed. The issue of heteroscedasticity, autocorrelation, and cross-sectional dependence are fixed with Driscoll-Kray standard errors to obtain the robust estimators (Tatoglu, 2012).

4. Results and Discussion

As can be seen in Equation (1), Equation (2), and Equation (3), three different models are created with three groups of variables to see the impact of financial performance, market performance, and corporate structure on capital structure decisions. Within this context and as stated above, 27 REIT companies traded in BIST from 2014 to 2020, whose data can be accessed, are included in this study. In addition, strongly balanced panel data is used and the above-mentioned statistical tests are run to obtain the best regression models.

According to the results of the study, the variable MTB used to evaluate the effect of growth opportunities on capital structure is found to have a significantly positive relation with DCR at %5 significant level. However, the literature indicates that the increase in growth opportunities makes firms more risk averse about debt so our results are not supportive of the previous studies (Chen & Zhao, 2006)

The size of the company is found to be positively related to DCR at 5% significance level in all three models as expected. Emerging market studies (Booth, Demirguc-Kunt and Maksimoviç (2001); Huang and Song (2006) and the study of Rajan and Zingales (1995) performed with G7 countries found a positive relationship between the leverage ratio and firm size.

The tangible assets make firms more willing to use financing resources because of their strong collateralized asset. However, our results indicate that the increase in tangibles do not trigger firms about using financing solutions. The hypothesis cannot be supported by the data of Turkish public REITs as this variable is not found to have any significant link with the selected dependent variable in all three models.

The Pecking Order Theory suggests that firms primarily choose to finance their investments from internal sources so profitable firms borrow less. Thus, firms go to the way of using debt in case that the amount of investment exceeds their profits (Myers 1984). In theory, it is stated that leverage and profitability should be in an inverse relationship. The significantly negative coefficient at 5% level in all three models supports what the theory says.

Better market performance triggers firms to use less debt because the investors observe that the firm's financial and operational situation is well enough to invest in and this causes the stock price to increase. Thus, the Pecking Order Theory supports this result.

It is expected that the older firms are less aggressive about growth opportunities so the use of debt becomes lower as the age increases. However, we obtain insignificant results about the age effect on capital structure.

As the free float rate increases, the firms tend to use more debt because the higher free float rate is a result of low operational performance and use of equity financing in previous years. This is supported by the findings of Model (3) as can be seen from the positive and significant coefficient.

Our empirical results suggest that there is a negative relationship between DCR and governmental dummy at %5 significance level so REITs with government shares tend to use less debt. There are several empirical studies that shed light to capital structure of firms from different countries so the literature presents different results for different countries. For example, the study of Deesomsak et al. (2004) indicates that the government involvement helps Canadian firms to borrow with lower rates because of the government guarantee. However, the results for Turkish public REITs are found to be the opposite during the selected observation period, like the study of Khaki and Akin (2020). The empirical results of this study state that there is negative relationship between government share and leverage for Gulf Cooperation Council (GCC) countries.

	Financial Performance	Stock Performance	Corporate Structure
V	Model 1	Model 2	Model 3
variable	DCR	DCR	DCR
MTB	8.969	9.867	9.994
	(2.962)**	(2.625)**	(2.35)**
SIZE	2.388	2.727	2.992
	(1.018)**	(0.94)**	(0.999)**
TAN	-6.580	-3.922	-4.447
	(5.258)	(5.103)	(5.011)
PRO	-49.726	-51.137	-48.4
	(18.585)**	(18.727)**	(15.6)**

Table 2: Panel Data Analysis Results for 3 Models Designed to See the Impacts of Financial Performance,

 Stock Performance and Corporate Structure of ISE REITs on Book Debt Ratio

RETURN		-5.412	-5.343
		(2.683)*	(2.35)**
AGE			-0.099
			(0.07)
FFR			0.17
			(0.059)**
GOV			-4.899
			(1.718)**
Constant	-47.546	-53.99	-60.531
	(20.839)**	(17.903)**	(20.566)**
R ²	0.65	0.746	0.783
Firm-years	189	189	189
legend	* p<0.10;	** p<0.05;	***p<0.01;

6. Conclusion

In this study, it is aimed to determine the debt utilization dynamics of REITs registered in BIST. Overall regression results of three models provide evidence that higher market-to-book ratio and size of the firm are associated with more book debt ratio. However, a statistically significant negative coefficient that is observed in profitability suggests that more profitability triggers less book debt ratio. These results are consistent with the expectations the study. On the other hand, tangibility is insignificant, which is contrary to what is expected because firms with more tangible assets can easily reach long-term debts. Nevertheless, the model designed to see the DCR behavior of Turkish REITs sample tells us that tangibility has a statistically insignificant effect on book debt ratio. In the third model, which is designed to measure the impact of corporate structure on book debt ratio, we have found significant results for the free float rate and government share but the result for government share is contrary to what is expected. Since the firms with government share are expected to reach external sources easily. In the regression results, the sign of the coefficient on FFR is positive and statistically significant which indicates that higher free float rate is linked with higher book debt ratio. However, the coefficient on AGE is found to be small and statistically insignificant.

In this study, the focus point is the REITs because of the importance in the Turkish economy and the different capital structure of REITs when compared to other sectors. Further studies can be performed for different periods to see the effects of different interest policies of Turkish government and inflation shocks on capital structure because REITs are sensitive to interest rate policies and inflation.

References

- Antoniou, A., Guney, Y. & Paudyal, K. (2008). The Determinants of Capital Structure: Capital Market-Oriented versus Bank-Oriented Institutions, Journal of Financial and Quantitative Analysis, 43 (1), 59-92.
- Baltagi, B. H. (2001). Econometric analysis of panel data. (2nd ed.). United Kingdom: John Wiley&Sons, Ltd.
- Bessler, W., Drobetz, W. & Kazemich, R. (2011). Factors affecting capital structure in H. Baker and G. Martin (eds), Capital Structure and Corporate Financing Decisions, Hoboken, NJ: John Wiley & Sons.
- Bevan, A. & Danbolt, J. (2002). Capital Structures and its Determinants in the UK-a Decompositional Analysis. Applied Financial Economics, 12(3), 159-170.
- Booth, L. V. A., Demirguc-Kunt, A. & Maksimovic, V. (2001). Capital structures in developing countries. Journal of Finance, 56(1), 87-130.
- Chen, L. & Zhau, X. (2004). Profitability, Mean Reversion of Leverage Ratios and Capital Structure Choices. Michigan State University and Kent State University, Working Paper. Available at: http://ssrn.com/ abstract=614082.
- Damodaran, A. (2001). Corporate finance: Theory and Practice. New York: John Wiley and Sons.
- Deesomsak, R., Paudyal, K., & Pescetto, G. (2004). The determinants of capital structure: Evidence from the Asia Pacific Region. Journal of Multinational Financial Development, 14(4-5), 387-405.
- Guner, A. (2016). The determinants of capital structure decision: new evidence from Turkish companies. Procedia Economics and Finance, 38, 84–89.
- Huang, G., & Song, F. M. (2006). The determinants of capital structure: Evidence from China. China Economic Review, 17(1), 14-36.
- Khaki, A. R., & Akin, A. (2020). Factors affecting the capital structure: New evidence from GCC countries. Journal of International Studies, 13(1), 9-27.
- Kraus, A., & Litzenberger, R. H. (1973). A State-Preference Model Of Optimal Financial Leverage. The Journal of Finance, 28(4), 911-922.
- Li, K., Yue, H., & Zhao, L. (2009). Ownership, institutions, and capital structure: Evidence from China. Journal of Comparative Economics, 37(3), 471-490.
- MacKay, P., & Phillips, G. M. (2005). How Does Industry Affect Firm Financial Structure? The Review of Financial Studies, 18(4), 1433-1466.
- Marsh, P. (1982). The Choice Between Equity and Debt: An Empirical Study. The Journal of Finance, 37(1), 121-144.
- Matias, F. & Serrasqueiro, Z. (2017). Are There Reliable Determinant Factors of Capital Structure Decisions? Empirical Study of SMEs in Different Regions of Portugal. Research in International Business and Finance, 40, 19–33.
- Mocnik, D. (2001). Asset Specificity and a Firm's Borrowing Ability: An Empirical Analysis Of Manufacturing Firms. Journal of Economic Behavior & Organization, 55, 69-81.
- Modigliani, F. & Miller, M. H. (1958). The Cost of Capital, Corporation Finance and the Theory of Investment. The American Economic Review, 48(3), 261-281.
- Myers, S. C. (1984). The capital structure puzzle. The journal of finance, 39(3), 574-592.
- Prasad, D., Dheeriya, P. L. & Woodruff, C. (2002). A Comparative Study of the Capital Structure of Indian Firms with the Firms of Developing European and Asian Countries. www.sbaer.uca.edu/ research/1999/WDS/99wds196.htm.

- Rajan, R. G. & Zingales, L. (1995). What do we know about capital structure? Some evidence from international data. The Journal of Finance, 50(5), 1421-1460.
- Sayilgan, G., Karabacak, H., & Kuçukkocaoglu, G. (2006). The Firm-specific Determinants of Corporate Capital Structure: Evidence from Turkish Panel Data. Investment Management and Financial Innovations, 3(3), 125–139.
- Sayilgan, G. & Uysal, B. (2011). Türkiye Cumhuriyet Merkez Bankası Sektörel Bilançoları Kullanılarak Sermaye Yapısını Belirleyen Faktörler Üzerine Bir Analiz: 1996 – 2008. Ankara Üniversitesi SBF Dergisi, 66(4), 101-124.
- Titman, S. & Wessels, R. (1988). The Determinants of Capital Structure Choice. Journal of Finance, 43(1), 1-19.
- Wald, J. K. (1999). How firm characteristics affect capital structure: an international comparison. Journal of Financial research, 22(2), 161-187.

Wooldridge, J. M. (2002). Econometric analysis of cross section and panel data. London: The MIT Press.

Tatoglu, F. Y. (2012). Panel veri ekonometrisi. İstanbul: Beta Yayınları.